

LUND UNIVERSITY

Efficient mm-Wave Transmitter Design in CMOS Technology

Forsberg, Therese

2018

Document Version: Publisher's PDF, also known as Version of record

Link to publication

Citation for published version (APA): Forsberg, T. (2018). Efficient mm-Wave Transmitter Design in CMOS Technology. [Doctoral Thesis (compilation), Faculty of Engineering, LTH]. The Department of Electrical and Information Technology.

Total number of authors:

General rights

Unless other specific re-use rights are stated the following general rights apply:

Copyright and moral rights for the publications made accessible in the public portal are retained by the authors and/or other copyright owners and it is a condition of accessing publications that users recognise and abide by the legal requirements associated with these rights. • Users may download and print one copy of any publication from the public portal for the purpose of private study

- or research.
- You may not further distribute the material or use it for any profit-making activity or commercial gain
 You may freely distribute the URL identifying the publication in the public portal

Read more about Creative commons licenses: https://creativecommons.org/licenses/

Take down policy

If you believe that this document breaches copyright please contact us providing details, and we will remove access to the work immediately and investigate your claim.

LUND UNIVERSITY

PO Box 117 221 00 Lund +46 46-222 00 00

Efficient mm-Wave Transmitter Design in CMOS Technology

Therese Forsberg



LUND INSTITUTE OF TECHNOLOGY

Doctoral Dissertation Lund, Sweden 2018

Department of Electrical and Information Technology Lund University P.O. Box 118 SE-221 00 LUND SWEDEN

ISSN 1654-790X, no.117 ISBN 978-91-7753-802-8 (print) ISBN 978-91-7753-803-5 (pdf) Series of licentiate and doctoral dissertations.

© Therese Forsberg 2018. Produced using $\[AT_EX\]$ Documentation System. Printed in Sweden by *Tryckeriet i E-huset*, Lund. November 2018. Till min morfar Arnold Meyer

Abstract

An increasing demand of higher data rates in wireless communication forces the industry to look to higher frequencies to find the required bandwidths. This thesis is about analog transmitters in CMOS for millimeter-wave communication, and it focuses on improving power amplifiers and frequency generation circuits, and increase their efficiency. This thesis starts with an introduction to millimeter-wave transmitters in CMOS, standards and beamforming. It then continues with a brief introduction to millimeter-wave power amplifier design and design of local oscillators at millimeter-wave frequencies. The last part of the thesis consist of six papers, which present eleven manufactured and measured millimeter-wave circuit designs. Paper I presents a two-stage, 65-nm CMOS, Class-A PA for the 60-GHz band. It employs capacitive cross-coupling neutralization for higher differential isolation and gain, without the need to increase the power consumption. It achieves 18.5 % peak-added-efficiency. Paper II presents a varactorless VCO in 65 nm CMOS, operating in the 60-GHz band. In paper III, the efficiency of the popular source-node filtering technique for improved phase-noise performance is investigated through measurements of two same-chip 60-GHz VCOs in FD-SOI CMOS. The filtered VCO achieves a state-of-the-art figure-of-merit of -187.3 dBc/Hz. Paper IV presents two FD-SOI CMOS VCOs for the 30-GHz and the 60-GHz band, that achieve ultra-low power consumption, also at full supply voltage. In paper V, a phase-locked loop in 28-nm FD-SOI CMOS for 5G transceiver systems is proposed. Its VCO operates at around 55 GHz. The paper describes the disadvantages of using a too high input reference frequency, but also proposes a new architecture that handles the increased settling time by mode-switching. It also includes a novel charge-pump current-mismatch mitigation technique based on feedback, and a novel wideband and low-power injection-locked divide-by-three circuit. The phase-locked loop consumes only 10 mW of power, has an integrated jitter of 176 fs, and demonstrates a state-of-the-art figure-of-merit of -245 dB. Paper VI describes a wideband injection-locked divide-by-two circuit in 28-nm FD-SOI CMOS. It achieves a locking range of 30 % at the low power consumption of 4.3 mW.

Populärvetenskaplig sammanfattning

Vi lever i en uppkopplad värld, där vi tar för givet att det alltid finns mobileller WiFi-täckning så att vi kan kontakta dem som vi bryr oss om, hitta rätt i en ny stad, finna nog med information att fatta ett beslut, eller helt enkelt bara roa oss. Detta är inte på något vis endast ett svenskt fenomen. Det finns idag 7,9 miljarder mobilabonnemang på jorden, vilket är 200 miljoner fler än vad det finns människor. En framtidsvision som redan håller på att hända är att inte bara människor, utan även föremål som t ex hushållsapparater, bilar och medicinska implantat skall kunna vara uppkopplade. I varje uppkopplad pryl finns det en sändare och mottagare, som kan omvandla de elektromagnetiska signalerna i luften till digitala data som vi har i våra datorer och telefoner. Eftersom signalerna i luften är analoga, så är även den del av sändaren och mottagaren som sitter närmast antennen analog. Just denna del, specifikt sändaren och dess ingående delar, är vad denna avhandling handlar om. För att en sändare skall passa i konsumentelektronik, måste den vara billig och dessutom strömsnål, så att batteriet håller länge. Billigare kan den bli om man gör den i en vanlig sorts kiselteknologi som kallas CMOS, och strömsnålare kan den bli om man noggrant går igenom varje liten del i sändaren och försöker att hitta nya sätt att spara ström. Sändare utsätts också hela tiden för hårdare krav. Ju bättre upplösning som filmen som vi vill streama har, desto mer data måste vi skicka samtidigt. När alla vill ha högre datahastigheter, tar utrymmet i luften som vi delar helt enkelt slut. Tänk dig ett rum, där alla skriker samtidigt. Ingen kommer att bli hörd och förstådd. Det finns redan många system för att utnyttja vårt gemensamma utrymme bättre, och det finns hårda krav på hur sändare får lov att sända för att störa andra så lite som möjligt. Ett system för att använda rummet bättre, är att alla pratar endast när det är deras tur. Ett annat snillrikt system efterliknar en situation där alla i rummet talar olika språk, vilket gör det lättare att urskilja en enskild person. Ibland använder avancerade system även trattar (antenner) för att se till att ljudet går mest i den riktning som man vill. Att alla pratar på olika frekvenser är ytterligare ett sätt, ungefär som om någon bara får tala med basröst, och någon annan bara i falsett. Men oavsett hur många finurliga sätt som man har kommit på, börjar platsen ta slut. Vad som ligger härnäst i utvecklingen är att tillåta och möjliggöra att kommunikationen kan ske på högre och högre frekvenser. Det är svårt, eftersom naturen har ordnat det så, att ju högre frekvenser som används, desto mer dämpas signalen i luften. Dessutom blir det svårare att använda den billiga och vanliga CMOS-kiselteknologin, eftersom de individuella transistorerna får svårare att hänga med och fungera korrekt. Vad sändarna som den här avhandlingen handlar om måste klara är att skicka rätt data med en bärvågsfrekvens på upptill ungefär 65 GHz. Eftersom en elektromagnetisk signal med en frekvens mellan 30 och 300 GHz har en våglängd som är mellan en och tio millimeter, så kallas det här frekvensområdet för millimetervågor. Som en jämförelse använder inga mobiltelefoner i dagsläget frekvenser över 6 GHz. Det finns redan produkter som kan kommunicera kring 60 GHz, men jämfört med flera miljarder mobiltelefoner är de är extremt få, och för att kunna bli en del av vår vardag behöver de bli billigare, och förbättras genom att varje individuell del görs mer effektiv. Vi kallar kretsar för mer effektiva om de kan utföra sina uppgifter korrekt men med mindre förbrukad effekt, eller om de kan göra sin uppgift bättre med samma effektförbrukning som innan. Denna avhandling fokuserar på två delar av analoga millimetervågssändare: effektförstärkare, som ofta förbrukar mer än hälften av den totala effekten i en sändare och där varje uns av förbättring ger stora effekter, samt den grupp av kretsar som är ansvariga för frekvensgenereringen. En förstärkare som är ovanligt effektiv i sin klass har konstruerats, likaså en faslåst loop som utför sin uppgift lika bra som andra publicerade faslåsta loopar, men med mindre än halva effektförbrukningen. Olika förslag på ytterligare förbättringar av viktiga delar i faslåsta loopar har utvecklats, och lett till konstruerade och mätta spänningsstyrda oscillatorer och injektionslåsta delare som visar att dessa kretsar kan klara sin uppgift och samtidigt bara förbruka en femtedel till en tredjedel av vad som är standard. Sammantaget presenteras elva kretsar som alla innehåller exempel på hur millimetervågssändare kan bli mer effektiva.

Preface

This dissertation summarizes my academic work for a PhD degree in Electrical Engineering in the Analog RF Design group, at the Department of Electrical and Information Technology, Lund University, Sweden. This work was funded by the Swedish Research Council (Vetenskapsrådet) and the European Commission in the framework of the H2020-ICT-2014-2 project Flex5Gware (Grant agreement no. 671563). The chip fabrication was donated by STMicroelectronics. The dissertation is divided into two parts. The first part gives an introduction to the research field, and the second part consists of the included research papers, as listed below.

Included Research Papers

The main scientific contribution is derived from the following publications:

- Paper I T. Forsberg, H. Sjöland and M. Törmänen, "A two-stage mmwave PA with 18.5% PAE in 65 nm CMOS," in Proc. of 2015 Asia-Pacific Microwave Conference (APMC), Nanjing, China, Dec. 6–9 2015, pp. 1–3. 1
- Paper II T. Forsberg, H. Sjöland and M. Törmänen, "A 65 nm CMOS varactorless mm-wave VCO," in Proc. of 2014 International Symposium on Integrated Circuits (ISIC), Singapore, Dec. 10–12 2014, pp. 54–57.
- Paper III T. Forsberg, J. Wernehag, A. Nejdel, H. Sjöland and M. Törmänen, "Two mm-Wave VCOs in 28-nm UTBB FD-SOI CMOS," *IEEE Microwave and Wireless Components Letters*, vol. 27, no. 5, pp. 509–511, May 2017.
- Paper IV T. Forsberg, J. Wernehag, H. Sjöland and M. Törmänen, "Two Ultra-Low Power mm-Wave Push-Pull VCOs in FD-SOI CMOS," in Proc. of 2018 Asia-Pacific Microwave Conference (APMC), Kyoto, Japan, Nov. 6–9 2018, pp. 1–3.
- Paper V M. Abdulaziz, T. Forsberg, M. Törmänen and H. Sjöland, "A 10 mW mm-Wave Phase-Locked Loop with Improved Lock Time in 28-nm FD-SOI CMOS," *IEEE Transactions on Microwave The*ory and Techniques, pp 1–12, 2018. Submitted.

¹Won Best Student Paper Award at the conference

Paper VI T. Forsberg, J. Wernehag, H. Sjöland and M. Törmänen, "A 4.3mW mm-Wave Divide-by-Two Circuit with 30% Locking Range in 28-nm FD-SOI CM," in Proc. 2018 IEEE Nordic Circuits and Systems Conference (NORCAS), Tallinn, Estonia, Oct. 30–31 2018, pp. 1–4.

Related Publications

During my Ph.D. studies, I have also co-authored the following papers, which contain overlapping material, but are not considered a part of this dissertation.

- I. Din, S. Anderson, T. Forsberg and H. Sjöland, "A 24 GHz, 18 dBm, Broadband, Three Stacked Power Amplifier in 28nm FDSOI," in *Proc.* of *IEEE NORCAS 2018*, Tallinn, Estonia, Oct. 30–31 2018, pp. 1–4.
- A. Bondarik, T. Forsberg, D. Sjöberg, H. Sjöland and M. Törmänen, "Microstrip Antenna Array Integrated with a Two-stage mm-wave CMOS Power Amplifier," *IEEE Microwave and Wireless Components Letters*, 2018. *Submitted*.

Acknowledgments

The work presented in this thesis would not have been possible without the help and support of many.

First of all, I would like to express my sincere thanks to my supervisor Markus Törmänen. One of the most impressive ways that his deep understanding of all subjects at hand manifests is his ability to swiftly find the core of the most complex technical problems and know where to focus to reach a solution. Without his guidance, none of this would have been possible. I will be forever thankful for the chance that I was given to do research in such an interesting field. I am also grateful to my co-supervisor, who is also my initial main supervisor, Henrik Sjöland. Not only is his knowledge of the field brilliant, he has also taught me about how to properly present my research, and he is a true role model when it comes ethical questions surrounding research in general. A huge thank you also goes to my co-supervisor Johan Wernehag, for his support in trying times and his ability to put thing into perspective.

Thank you also to all the people in the Integrated Electronics Systems group that made life in our corridor feel friendly and supportive. I will always remember fondly my fellow PhD students in the RF Analog group that I had the fortune to work with, and that brightened up my days. Thanks to Carl Bryant and Jonas Lindstrand for the circuits that we created once when time was short, thank you to Anders Nejdel and Waqas Ahmad for the collaboration to create pad designs that has since been used by me and many others. I would also like to extend a very special thank you to Mohammed Abdulaziz, who endured long sessions in the lab with me, striving to get our design to work. His untiringly positive outlook on things is truly inspiring and it has set a great example for me.

To all the people at EIT that provide invaluable technical and administrative support: I am grateful for your efforts. Thank you to Göran Jönsson, Andreas Johansson, and Martin Nilsson for maintaining the lab, and to Lars Hedenstjerna for making beautifully perfect parts that made my measurements possible. Thank you to Stefan Molund and Erik Johnsson for not giving up trying to fix all of my software issues. Thank you to Pia Bruhn, Anne Andersson and Elisabeth Nordström for your magical ability to make all my administrative task and questions seem trivial.

During my PhD studies I have been fortunate enough to be able to go to international conferences, through the financial support of ÅForsk, the IEEE Microwave Theory and Techniques Society, and the Royal Physiographic Society of Lund. For this, I am very grateful.

I would like to send a thank you to all the people involved in, and around, the Microwave Electronics Laboratory at Chalmers University of Technology, for allowing me to make some of my measurements with them, and for providing such a welcoming and scientifically interesting environment.

I owe so much to my previous colleagues at Ericsson in Lund. Sometimes it is Lars Sundström or Peter Caputa that provides me with some invaluable piece of information to further my knowledge, sometimes it is Per Sandrup that helps me to get hold of some crucial software licenses in the middle of Christmas vacations, or Peter Herrder that lends me essential pieces of lab equipment, and always, it is everyone, just being wonderfully supportive.

I would also like to thank my family. To my mother Bodil, my father Bo-Göran, and my mother-in-law Birgitta, thank you for always being there for me. To my amazing sister Ulrika and to Daniel, thank you for being my role models for my academic undertakings, and for being generally awesome people. Thank you to Sven, my kind husband, who has patiently endured my long days before deadlines and never stopped believing in me.

Finally, to all the people that I was unable to mention here, thank you!

There Jourserg

List of Acronyms

3GPP	3rd Generation Partnership Project
AM	Amplitude modulation
CML	Current-mode logic
CMOS	Complementary metal-oxide-semiconductor
CP	Charge pump
DAC	Digital-to-analog converter
dc	Direct current
DPD	Digital pre-distortion
FD-SOI	Fully depleted silicon-on-insulator
FOM	Figure of merit
GaAs	Gallium-Arsenide
HBT	Heterojunction bipolar transistor
IF	intermediate frequency
ILFD	Injection-locked frequency divider
ISM	Industrial, scientific and medical
LO	Local oscillator
mm-wave	millimeter-wave
NMOS	${\it n-channel\ metal-oxide-semiconductor\ field-effect\ transistor}$
NR	New radio
OFDM	Orthogonal frequency-division multiplexing
PA	Power amplifier
PAE	Power-added efficiency
PAPR	Peak-to-average power ratio
PCB	Printed circuit board
PFD	Phase-frequency detector
PLL	Phase-locked loop
PM	Phase modulation
PMOS	$\label{eq:p-channel} p-channel metal-oxide-semiconductor field-effect transistor$

PN	Phase noise
PPA	Pre-power amplifier
PSRR	Power supply rejection ratio
QAM	Quadrature amplitude modulation
\mathbf{RF}	Radio frequency
SOI	Silicon-on-insulator
UTBB	Ultra-thin body and buried oxide
VCO	Voltage-controlled oscillator
WPAN	Wireless personal area network

List of Symbols

&	Logical AND operation
β	Feedback factor
Δ	Difference
$\Delta \Phi$	Phase difference
Δf	Frequency difference [Hz]
λ	Wavelength [m]
ω	Angular frequency [rad/s]
$\omega_{carrier}$	Angular frequency of the carrier [rad/s]
ω_n	Natural frequency of a PLL [rad/s]
ω_{osc}	Angular frequency of oscillation [rad/s]
ω_{range}	Locking range of an ILFD [rad/s]
ω_{res}	Resonance frequency [rad/s]
ϕ_{in}	Input signal phase
ϕ_{out}	Output signal phase
ω_{res}	Resonance frequency [rad/s]
ζ	Damping factor of a PLL
η	Drain efficiency of a PA $[\%]$
θ	Angle [rad]
A	Amplification, gain
В	Noise integration bandwidth [Hz]
C	Capacitance [F]
C_i	Capacitance i [F]
C_{filter}	Filter capacitance [F]
C_{par}	Parasitic capacitance [F]
C_t	Tank capacitance [F]
CLK	Clock signal
CP_{1dB}	1-dB compression point in a PA [dBm]
DN	The "down"-signal to a CP

DN	The "down"-signal to a CP
F	Amplifier noise factor
f	Signal frequency [Hz]
f_0	Output frequency of an ILFD [Hz]
f_{bb}	Baseband signal frequency [Hz]
f_{corner}	Flicker noise corner in a VCO [Hz]
f_{fb}	Feedback signal frequency to the PFD in a PLL [Hz]
f_{in}	Input signal frequency [Hz]
f_{LO}	Local oscillator signal frequency [Hz]
f_{max}	Maximum signal frequency [Hz]
f_{min}	Minimum signal frequency [Hz]
f_{osc}	Signal oscillation frequency [Hz]
f_{out}	Output signal frequency [Hz]
f_{ref}	Reference signal frequency [Hz]
f_{res}	Resonance frequency [Hz]
f_T/f_{max}	Frequency of unity current gain/unity power in a transistor $\left[\mathrm{Hz}\right]$
f_T/f_{max} G	Frequency of unity current gain/unity power in a transistor [Hz] Gain [dB]
f_T/f_{max} G g_m	Frequency of unity current gain/unity power in a transistor [Hz] Gain [dB] Transistor transconductance [S]
f_T/f_{max} G g_m H(s)	Frequency of unity current gain/unity power in a transistor [Hz] Gain [dB] Transistor transconductance [S] Transfer function in the Laplace plane
f_T/f_{max} G g_m H(s) I	Frequency of unity current gain/unity power in a transistor [Hz] Gain [dB] Transistor transconductance [S] Transfer function in the Laplace plane Current [A]
f_T/f_{max} G g_m H(s) I I_{cp}	Frequency of unity current gain/unity power in a transistor [Hz] Gain [dB] Transistor transconductance [S] Transfer function in the Laplace plane Current [A] Charge-pump current [A]
f_T/f_{max} G g_m H(s) I I_{cp} I_{inj}	Frequency of unity current gain/unity power in a transistor [Hz] Gain [dB] Transistor transconductance [S] Transfer function in the Laplace plane Current [A] Charge-pump current [A] Injected current [A]
f_T/f_{max} G g_m H(s) I I_{cp} I_{inj} I_{osc}	Frequency of unity current gain/unity power in a transistor [Hz] Gain [dB] Transistor transconductance [S] Transfer function in the Laplace plane Current [A] Charge-pump current [A] Injected current [A] VCO oscillation current [A]
f_T/f_{max} G g_m H(s) I I_{cp} I_{inj} I_{osc} k	Frequency of unity current gain/unity power in a transistor [Hz] Gain [dB] Transistor transconductance [S] Transfer function in the Laplace plane Current [A] Charge-pump current [A] Injected current [A] VCO oscillation current [A] Boltzmann constant $[m^2 \cdot kg \cdot s^{-2} \cdot K^{-1}]$
f_T/f_{max} G g_m $H(s)$ I I_{cp} I_{inj} I_{osc} k K_{VCO}	Frequency of unity current gain/unity power in a transistor [Hz] Gain [dB] Transistor transconductance [S] Transfer function in the Laplace plane Current [A] Charge-pump current [A] Injected current [A] VCO oscillation current [A] Boltzmann constant $[m^2 \cdot kg \cdot s^{-2} \cdot K^{-1}]$ VCO gain [Hz/V]
f_T/f_{max} G g_m $H(s)$ I I_{cp} I_{inj} I_{osc} k K_{VCO} $\mathscr{L}(\Delta f)$	Frequency of unity current gain/unity power in a transistor [Hz] Gain [dB] Transistor transconductance [S] Transfer function in the Laplace plane Current [A] Charge-pump current [A] Injected current [A] VCO oscillation current [A] Boltzmann constant $[m^2 \cdot kg \cdot s^{-2} \cdot K^{-1}]$ VCO gain [Hz/V] Single-sided PN [dBc/Hz], at distance Δf Hz from the carrier
f_T/f_{max} G g_m H(s) I I_{cp} I_{inj} I_{osc} k K_{VCO} $\mathscr{L}(\Delta f)$ L_{filter}	Frequency of unity current gain/unity power in a transistor [Hz] Gain [dB] Transistor transconductance [S] Transfer function in the Laplace plane Current [A] Charge-pump current [A] Injected current [A] VCO oscillation current [A] Boltzmann constant $[m^2 \cdot kg \cdot s^{-2} \cdot K^{-1}]$ VCO gain [Hz/V] Single-sided PN [dBc/Hz], at distance Δf Hz from the carrier Filter inductance [H]
f_T/f_{max} G g_m H(s) I I_{cp} I_{inj} I_{osc} k K_{VCO} $\mathscr{L}(\Delta f)$ L_{filter} L_t	Frequency of unity current gain/unity power in a transistor [Hz] Gain [dB] Transistor transconductance [S] Transfer function in the Laplace plane Current [A] Charge-pump current [A] Injected current [A] VCO oscillation current [A] Boltzmann constant $[m^2 \cdot kg \cdot s^{-2} \cdot K^{-1}]$ VCO gain [Hz/V] Single-sided PN [dBc/Hz], at distance Δf Hz from the carrier Filter inductance [H]
f_T/f_{max} G g_m $H(s)$ I I_{cp} I_{inj} I_{osc} k K_{VCO} $\mathscr{L}(\Delta f)$ L_{filter} L_t M_i	Frequency of unity current gain/unity power in a transistor [Hz] Gain [dB] Transistor transconductance [S] Transfer function in the Laplace plane Current [A] Charge-pump current [A] Injected current [A] VCO oscillation current [A] Boltzmann constant $[m^2 \cdot kg \cdot s^{-2} \cdot K^{-1}]$ VCO gain [Hz/V] Single-sided PN [dBc/Hz], at distance Δf Hz from the carrier Filter inductance [H] Tank inductance [H]

P	Power [W]
P_{dc}	dc power consumption [W]
P_{in}	Input signal power [W]
P_{out}	Output signal power [W]
$P_{out,max}$	Maximum saturated output signal power [W]
Q	Quality factor
R	Resistance $[\Omega]$
R_i	Resistance $i [\Omega]$
R_L	Load resistance $[\Omega]$
r_o	Transistor output resistance $[\Omega]$
R_P	Parallel loss resistance $[\Omega]$
RST	Reset signal
Т	Signal cycle time [s]
T_{ref}	Reference signal cycle time [s]
UP	The "up"-signal to a CP
V	Voltage [V]
$V_{b,high}$	dc bias voltage for high-side injection in an ILFD $\left[\mathrm{V}\right]$
$V_{b,low}$	dc bias voltage for low-side injection in an ILFD $\left[\mathrm{V}\right]$
$V_{b,inj}$	dc bias voltage for high-side injection in an ILFD $\left[\mathrm{V}\right]$
V_{bias}	dc bias voltage [V]
v_{ctrl}	Control voltage in to the VCO in a PLL [V]
V_{osc}	Oscillation voltage [V]
V_{in}	Input signal voltage [V]
V_{out}	Output signal voltage [V]
V_{var}	Control voltage to the varactor [V]
V_b	dc bias voltage [V]
VDD	dc voltage supply [V]
X_L	Load impedance $[\Omega]$

Introduction

Chapter 1

Introduction

1.1 Motivation

In today's fast-paced world, we have come to take wireless data access for granted. Wherever we go, we expect cellphone or WiFi coverage to aid us in getting in touch with those important to us, finding our way, acquire enough information to make a decision, or simply to entertain ourselves. The number of cellphone subscribers is steadily rising, and already today there are more than 7.9 billion subscriptions in the world – that is 200 million more subscriptions than there are people [1]. It is expected that in the future, subscribers can also be objects, such as home appliances, cars, or medical implants. Hence, the global mobile data traffic is growing fast, and it is forecasted to keep growing at an even faster pace, as in shown in Fig. 1 [2]. The expected required data speeds are rising along with it. The higher the quality of the photos or movies that we send, the more data needs to be transferred rapidly. This data is modulated onto a signal of a specific carrier frequency when sent though the air, and the wider the frequency span that it spreads over around the carrier frequency, the more data it can carry. But we all share the same air, and if everyone yells at the same time, no one will be heard. Several standards for wireless communication try to solve this problem, restricting when and what each access node can send and employing complex techniques that minimize cross-talk. Soon, however, that will not be enough either. So far, almost all carrier frequencies for wireless communication have been below a few GHz, because wireless communication at higher frequencies is hindered by physical phenomena such as higher in-air attenuation, higher penetration loss, and more shadowing effects from obstructing objects. However, going to higher carrier frequencies for wireless communication is a necessary step in the continuous data revolution. Some future communication frequencies are in the millimeterwave (mm-wave) range. The name "mm-wave" refers to frequencies between 30 GHz and 300 GHz, where wave-lengths are in the range of 1-10 mm. During the past years, a few systems have demonstrated the feasibility of mm-wave



Figure 1: Global mobile data traffic, with forecast. Offloaded traffic is wireless traffic though for instance WiFi or Bluetooth, i.e. not through cellphone networks. Based on [2].

communication, and are today commercially available. However, this does not mean that all problems are solved. To reach a bigger market, the electronics need to be even cheaper. They also need to use less power and still meet the same requirements, which will make batteries last longer and benefits both personal economy and the environment. Going for cheaper circuits often mean using CMOS technology, because it is readily available and adapted for massmarket production. It also facilitates integrating more of the system on the same chip. However, it has some drawbacks compared to other technologies, that need to be addressed and mitigated when designing the transceiver circuits. The cost of manufacturing circuits in CMOS rises with the physical size of the circuit, so they also need to be kept as small as possible. All of these desired traits are interconnected and have complex trade-offs. Some are not entirely understood yet, and all of them are subject to attempts of circumvention.

The aim of this thesis is to investigate how some aspects of mm-wave transmitter circuits in CMOS can be made more efficient, meaning that they are able to perform their task while using both a smaller area and/or a lower power consumption than previous art.

1.2 Outline of the Thesis

The thesis is organized as follows:

- Chapter 1 presents the motivation for the dissertation, and its organization.
- Chapter 2 introduces mm-wave radio transmitter systems, their building blocks and some commonly used performance metrics.
- Chapter 3 explores mm-wave power amplifier design.
- Chapter 4 focuses on efficient frequency generation circuits.
- Chapter 5 gives summaries and conclusions of the included papers along with the author's contribution.
- Chapter 6 provides a discussion with suggestions for future work.
- **Paper I** presents a mm-wave class-A power amplifier with high efficiency.
- **Paper II** presents a varactorless mm-wave voltage controlled oscillator.
- **Paper III** presents an empirical investigation of the impact of using a noisereducing architecture in mm-wave voltage controlled oscillators.
- **Paper IV** presents two ultra low-power mm-wave voltage controlled oscillators.
- **Paper V** presents a mm-wave analog phase-locked loop for 5G applications, with very low power consumption and an improved lock time.
- **Paper VI** presents a wideband injection-locked divide-by-two circuit.

Chapter 2

mm-Wave Radio Transmitters

This chapter introduces CMOS analog mm-wave transmitter systems. It discusses the environment in which they will work, and presents challenges and solutions related to their design on architecture level.

2.1 Communication at mm-Wave Frequencies

During the past decade, commercial interest in communication on mm-wave frequencies has steadily grown. Expanding to higher frequencies will alleviate the existing congestion in the spectrum below 6 GHz, and at the same time it also promises wider continuous bandwidths and higher data rates. Some frequency ranges in the mm-wave spectrum have attracted special interest, and they will be discussed next. It important to note that since mm-wave communication still has shortcomings, most new standards at mm-wave frequencies are extensions of already existing standards, and their primary use is as a complement to the existing infrastructure.

2.1.1 60 GHz

Around 60 GHz, a worldwide free Industrial, Scientific and Medical (ISM) band offers bandwidths of several GHz. Its allocation depends somewhat on geographical location, as is shown in Fig. 2. One of the reasons to place an ISM band around 60 GHz is because of the increased atmospheric attenuation due to oxygen molecules, see Fig. 3. This means that networks operating on this frequency will by necessity be very local, in practice only within a single room. For a specific sort of networks, this is not a disadvantage, since small networks also mean increased privacy, less interference from other nearby users operating at the same frequency, and a larger potential for frequency reuse.

The first standardized wireless personal-area network (WPAN) in the ISM band to offer Gbit/s data rates is IEEE 802.15.3c, which targets applications such as kiosk downloading, conference ad-hoc systems and video streaming. It divides the band into 4 channels, and each is 2.16 GHz wide [3], as is shown in Fig. 2.



Figure 2: Frequency allocation of the ISM band around 60 GHz, comprised of four subchannels. Based on [3].

The WiFi standard 802.11ad, also known by its previous trade name WiGig, is also implemented in the ISM band. It keeps the proposed division into 4 channels. Although 802.11ad also supports more simple modulation schemes, the primary modulation scheme is orthogonal frequency division multiplexing (OFDM). Using this scheme together with 64-point quadrature amplitude modulation (64QAM), the highest data rates mentioned in the standard reaches 6.75675 Gbit/s, making it one of today's fastest wireless technologies. This puts very stringent requirements on the hardware, and usually also leads to a high power consumption. For this reason, other modes are also available in the standard, that better suits handheld, battery-powered devices. However, all 802.11ad-enabled devices are capable of at least 1 Gbit/s [5].

In the ISM band around 2.5 GHz, new standards have continuously developed over time. Some become very well used, like WiFi and Bluetooth, others become important in niche applications or simply fail to get traction. The same can be expected also for the 60 GHz band. Except for the WPAN and WiFi standards mentioned above, several other standards are already in different stages of development.

2.1.2 70, 80, and 90 GHz

Around 80 GHz, the mm-wave E-band also offers wide bandwidths, without the limitations of high atmospheric attenuation that affects the 60 GHz ISM band. It is, however, more dependent on weather conditions, and the signal will be more attenuated in rainy conditions. The E-band is a licensed band for



Figure 3: Total attenuation due to atmospheric gases at sea level. Based on [4].

communication, and it consists of three parts of spectrum: 71 to 76 GHz, 81 to 86 GHz, and 92 to 95 GHz. One of its primary planned uses is backhaul for cellular networks. Backhaul is, by definition, a wired or wireless point-to-point communication link. For cellular communication, it denotes the link between a base station and the main switch board. This kind of wireless backhaul in the E-band is especially well suited for the small cells that are anticipated in population-dense areas, where the distance between nodes is no more than about a kilometer [6].

2.1.3 30 GHz

Another of the high-interest frequency ranges is located around 30 GHz. Here, the cellphone industry see a way forward for cellular communication. From the first tentative steps [7], the spectrum around 30 GHz is now becoming integrated into the 3GPP standards, under the name new radio (NR). The planned allocation of spectrum in different parts of the world is shown in Fig. 4, along with the first two named bands. The standard explicitly states that many more bands will be defined within the frequency range in the future. In each band, OFDM will be the primary modulation, and bandwidths up to 400 MHz will be supported [8].



Figure 4: Planned spectrum allocation and band definitions for 5G NR. Based on [8].

2.2 Transmitter Architectures

The basic function of an analog transmitter is summarized in Fig. 5. First, the digital-to-analog converter (DAC) converts the baseband digital signal into an analog one. The analog signal will in practice have some levels of harmonics, noise, and unwanted tones. In each moment, the signal is characterized by its amplitude and phase. Next, mixer(s) upconvert the signal to the target frequency. An ideal mixer multiplies the two input signals in the time domain, and in the frequency domain is creates the sum and difference of the signals. The mixing operation is dependent on the presence of a locally generated frequency signal of good quality. How to create a local oscillator (LO) signal is further discussed in chapter 4. Since the transmitter almost always share a chip with one or more receivers, and they also need an LO signal for their frequency translation, the LO is often also providing signals for the receiver. Finally, in the last step of the transmitter, some amplification of the up-converted signal is needed before it is sent to the antenna. The amplification can be variable or fixed, and it consists of a power amplifier (PA), which is sometimes preceded by a pre-power amplifier (PPA). Power amplifiers are presented in chapter 3. At lower frequencies, filters are usually needed between each stage of the transmitter. However, at mm-wave frequencies, the required filtering can be alleviated, as the individual circuit blocks are tuned to the intended frequency range, providing some filtering.

Generally, transmitter front-ends use either a homodyne or superheterodyne architecture. Both architectures were first described almost a hundred years ago, and both are still commonly used today.



Figure 5: Basic functions of an analog transmitter.



Figure 6: Homodyne transmitter.

2.2.1 The Homodyne Transmitter

The basic principle of a homodyne transmitter is shown in Fig. 6, where I/Q modulation is used. I/Q is short for in-phase and quadrature, and it refers to the practice of using two sinusoidal LO signals that have the same frequency, but are 90° out of phase. Conventionally, the *I* signal is a cosine waveform, and the *Q* signal is a sine waveform. This provides a powerful way to create any signal, because any signal can be generalized as

$$x(t) = A(t)cos[\omega_{carrier}t + \phi(t)]$$
(1)

$$= A(t)[\cos(\omega_{carrier}t)\cos(\phi(t)) - \sin(\omega_{carrier}t)\sin(\phi(t))], \qquad (2)$$

where the information is carried by the phase and amplitude. Hence, with the baseband signals defined as

$$x_{baseband,I}(t) = A(t)cos(\phi(t)) \tag{3}$$

$$x_{baseband,Q}(t) = A(t)sin(\phi(t)), \tag{4}$$

it is clear that this architecture can be used to create any signal. For the overall performance of the transmitter, it is very important that the phase difference does not deviate from 90° . An advantage of the homodyne architecture is that it is straightforward, and that it does not require a multitude of subcircuits. However, it demands that all analog parts work at the output frequency, which makes the design more difficult at mm-wave frequencies. The matching between the subcircuits will require more attention, and the local oscillator design will become challenging, especially in terms of noise and balance in the quadrature generation. Also, the PA may disturb and cause pulling in the local oscillator, as it will be working on almost the same frequency.

Some examples of demonstrated homodyne transmitters in CMOS for mmwave applications can be found in [9–11].

2.2.2 The Superheterodyne Transmitter

The basic principle of a superheterodyne transmitter is depicted in Fig. 7(a), illustrated as a two-step, single-sideband transmitter. The digital signal is first up-converted to an intermediate frequency (IF), and unwanted signals are removed by a filter. The signal is then upconverted again, using a second mixer and a second LO signal, to the intended carrier frequency.

In practice, the first LO and mixer are often responsible for the quadrature. Since they will operate at lower frequencies than the LO and mixer in the homodyne transmitter, the design will be less challenging. A disadvantage of this architecture is that it introduces additional circuits. However, at mm-wave frequencies, the filters will be less bulky, and the filters can fit on-chip if the IF is chosen carefully. Some recent examples of demonstrated superheterodyne transmitters in CMOS for mm-wave applications can be found in [12–16].

A special case of the superheterodyne transceiver is the sliding-IF architecture. Instead of keeping the IF constant, both LO frequencies move together. One advantage of this is that transmitter may not need multiple LOs. Each LO can be, for example, half of the transmitter carrier frequency, or they can be multiples of each other. Fig. 7(b) shows an architecture that has demonstrated promising results in mm-wave receivers [17].

2.2.3 Transmitter Architectures for Beamforming

Transmission on mm-wave frequencies is, as previously mentioned, difficult due to the high propagation loss. Every obstacle, such as walls or even raindrops,



(b)

Figure 7: (a) Superheterodyne transmitter. (b) Sliding IF transmitter.

will seem large to the small-wavelength signals. At around 60 GHz the high atmospheric attenuation due to oxygen molecules will add to the attenuation. However, the signal will not only face difficulties going from antenna to antenna. Using CMOS technology for cost and integration benefits will also mean choosing an inferior technology for high-frequency power applications. As technologies are down-scaled, the supply voltage is also scaled down, and the maximum voltage swing the transistors can handle without breaking is also lower. Additionally, the transistor gain decreases with increasing frequency. Hence, it can be difficult to even get a high power output signal to the antenna.

One way to address this problem is to maximize the signal from the antenna by making it increasingly directional, which ensures that all the power arrives at the correct target. Increasing the size of the antenna aperture will increase the directionality of the antenna [18], but also make it a lot more bulky and impractical. To steer the beam in one direction would also require moving the whole antenna. However, an advantage of high frequency communication is that the wavelengths are short and the antennas can thus be small. Instead of scaling up one antenna, it is possible to use many small, but carefully placed antennas, and make them work together. This configuration is called a phased array of antennas. Even if the signal from each antenna may be weak, many added signals will combine into a stronger signal. By controlling the phase of the signal that is sent from each sub-antenna in the array, they can be forced to create one, directional beam. This technique is called beamforming, and the basic principle is presented in Fig. 8(a). Changing the phase of each signal will quickly shift the beam to point somewhere else, without the need to physically move any antenna. This technique is called beam steering. Figure 8(b) and (c) show examples of this.

To summarize, beamforming and beam steering inherently means using many antennas in an array, and to carefully and continuously control the phase of the signal at each antenna. In practice, also the amplitude to each antenna is controlled, to suppress sidelobes of the signal. The phase shifting circuitry has to be able to produce a wide range of phases, with high resolution and linearity. The phase shift can, in theory, be implemented anywhere in the transmit chain, and each implementation has its advantages and disadvantages.

In one of the extreme cases, the phase shift is performed in the digital domain, and is thus simply called digital beamforming. This requires one DAC and a whole analog transmitter for each antenna. Depending on how challenging the LO signal distribution is, it may be possible to use one LO for all transmitters. In any case, the size of the whole circuit will increase, and the power consumption will drastically increase. However, this solution gives excellent precision, reconfigurability, and spectrum usage due to almost unlimited degrees of freedom.

In the other extreme case, the phase shifters are placed as close to the antenna as possible. The phase-shifting circuits are most often implemented as passive structures, such as delay lines. In this case, only one transmitter is needed for the whole phased array. However, the phase shifting will interfere with the matching between the PA and the antenna, and introduce losses. For these reasons, phase shifting circuitry can be placed before the PA instead. Examples of this can be found in for instance [19], which employs a switched-delay phase shifter, and in [20], which instead has an active vector modulator. The approach to place the phase shift before the power amplifier will create a need for as many PAs as there are antennas in the phased array. However, this suits CMOS PAs very well, since they are less expensive than specialized technologies that have better power-handling capabilities. In any case, when implementing the phase shift at mm-wave frequencies, obtaining enough precision and resolution is difficult. Placing the passive phase shifting circuits in the beginning of a superheterodyne transmitter chain, or simply right after the DAC, will make the control easier, but it will also mean that a larger portion of the transmitter have to be duplicated for each antenna. Another way of creating a phase shift, that does not require passive structures and that has demonstrated the ability

to create a wide range of phase shifts with high resolution, is to perform the phase shift inside the frequency generation circuitry [17].



Figure 8: (a) The principle of beamforming, illustrated with a phased array of eight antennas. Each phase shift equals a small time delay of the signal. With the correct timing, the signals from the antennas will create common wavefronts, that move in a defined direction. (b) An example of beam steering. Simulated 60-GHz radiation pattern of a beam and its sidelobes, from a 4x4 phased array of patch antennas on a printed circuit board (PCB). Figure courtesy of Dr. A. Bondarik. (c) An example of how beam steering can be used as a high-speed data wireless interface.

Chapter 3

mm-Wave Power Amplifiers

The PA is the last block in the transmitter before the antenna, and it must be able to deliver the required power to transmit the information, and at the same time meet the requirements on linearity, bandwidth and dc power consumption. This chapter will give an introduction to design considerations for mm-wave power amplifiers (PAs).

So far, the market for hand-held communication devices have favored PAs in technologies like GaAs HBT, that have good power-handling capabilities and work well at high frequencies. CMOS technologies struggle with delivering gain and power at high frequencies, and the limited output voltage swing before the devices break lowers the attainable maximum output power. However, as mentioned in chapter 2.2.3, an individual CMOS transmitter will not necessarily have to deliver as much power if it is used together with other PAs and a phased array of antennas. In that setting, the lower price of CMOS PAs actually makes it preferable. Also, this has huge integration advantages, if the rest of the transmitter is also designed in CMOS technology.

3.1 PA Metrics

The following metrics are the most commonly used metrics to describe a PA.

The **gain** of the PA is the ratio between the average input power P_{in} , in W, and the average output power P_{out} , in W, delivered to the load, i.e. $A = P_{out}/P_{in}$. If all quantities are expressed in decibels, then $G[dB] = P_{out}[dBm] - P_{in}[dBm]$.

When P_{in} is increased, P_{out} will also increase, but only up until the **maximum output power** $P_{out,max}$ is reached. Already before that point, the gain will have decreased. The **1 dB gain compression point** (CP_{1dB}) expresses where the gain has dropped 1 dB compared to the expected value. It is usually related to the output power, in dBm.

Compared to the other parts of a transmitter, the PA consumes a relatively large amount of the total power. Hence, an important metric is the **power consumption** (P_{dc}) , and also how efficiently the power is used. The **drain**


Figure 9: Principle of an ideal tuned PA.

efficiency is defined as $\eta = P_{out}/P_{dc}$. However, since it does not include the input signal power, another metric called the **power-added efficiency** (PAE) is often used instead. It is defined as $PAE = (P_{out} - P_{in})/P_{dc}$.

Since mm-wave PAs are tuned circuits their wideband operation, i.e. their **tuning range**, is also considered.

3.2 PA Fundamentals

An idealized representation of a mm-wave PA is shown in Fig. 9. In this example, the amplification is provided by a transistor in a common-source configuration. The transistor is biased through an ideal bias-T, with an infinite inductance towards the voltage supply VDD that only allows dc to pass through, and an infinite capacitance towards the load. Emphasizing the tuned nature of the PA, the load consists of the explicit load resistance R_L , but also a capacitance and an inductance that in this case forms a parallel resonant tank. The parasitic capacitances of the transistor, C_{par} , will be included in C_t , and likewise, R_L will include the resistive loss in the tank. The resonance frequency of the output circuitry is $f_{res} = 1/(2\pi L_t C_t)$.

3.2.1 Output Matching

As an example, if the aim is to deliver 20 dBm of output power to a 50 Ω load, which is a typical antenna impedance, it will lead to a peak output voltage that is over 3 V. Since this will risk breaking the transistor, an impedance matching network will be placed between the PA and the antenna, to make R_L look smaller when seen from the PA. An example of this is shown in Fig. 10(a) and (b). When transforming R_L in this way, the high output voltage becomes a high output current instead, but the delivered power stays the same. Consequently, the PA transistor must be able to handle large currents to deliver the necessary



Figure 10: (a) PA with output matching network. (b) PA with a transformer. (c) Differential PA with a balun.

output power. To be capable of that, the transistor size has to be large, and that in turn leads to a high input capacitance. To prevent the high input capacitance from excessively loading the previous stage, i.e. the mixer, PAs are often built with cascaded stages with increasingly large transistors in each stage.

Another reason for the importance of the output matching network is that it plays a role in terminating unwanted harmonics, but at the same time it must have a wide enough bandwidth for the signal. The matching network will be a part of the tank and set the resonance frequency of the PA. When designing a mm-wave amplifier, a design goal for the matching network is to provide the load impedance that maximizes the output power, called power matching or large signal matching. Off-chip matching networks and baluns for singleended to differential conversion are usually feasible at lower frequencies where packaging capacitances are small. However, at mm-wave frequencies, baluns and inter-stage matching between cascaded PA stages are naturally designed on-chip. Figure 10(c) shows a differential PA with a balun on the output.



Figure 11: Effects of PA nonlinearities. (a) Relation between input power and output power, and the power in the unwanted third-order intermodulation product. (b) A modulated signal with adjacent channel leakage.

3.2.2 Differential PAs

One of the necessary initial design choices for a PA is whether to use a differential or a single-ended PA topology. Both have advantages and disadvantages. A single-ended PA can be more straight-forward to design. However, the mixer in a transmitter usually has a differential output signal, which benefits a differential PA, but a balun is still needed at the PA output for differential-tosingle-ended signal conversion to interface the antenna. Physically, the balun may not consume so much more space, because it can double as a part of the output matching network, as shown in Fig. 10(c). A drawback is that passive components in the transmit chain result in increased signal losses. Other advantages of the differential approach are reduced even-order harmonics, increased dynamic range and less coupling to the LO path.

3.3 PA Classes

Two parameters in PAs are especially dependent on each other, namely the efficiency and the linearity.

The efficiency of a PA is very important, because the PA itself can easily consume more power than all other transmitter parts combined. As an example, if a PA with 50 % drain efficiency has to deliver 30 dBm of power (1 W), it will consume 2 W. The efficiency of a PA will also decrease with input power for almost all PA topologies, and it is rare for any PA to work at its maximum output power for longer periods. Usually, it is working with signals at least a few dBs below their maximum capacity, because modern modulation schemes have high peak-to-average power ratios (PAPR).

Linearity is of concern because it can lead to amplitude compression and

spectral regrowth, which in turn lead to high adjacent channel leakage, see Fig. 11(a) and (b), respectively. Any PA is inevitably nonlinear, since they operate with large signals, but the nonlinearity can be more or less severe, depending on design choices. More complex signal modulation schemes will lead to higher linearity requirements. To estimate the linearity, CP_{1dB} can be measured with a single-tone input, and the intermodulation products can be measured using two input tones. Other nonlinear effects are for instance AM/AM and AM/PM conversion, which is when input amplitude changes create gain and phase variations.

The trade-off between efficiency and linearity is made visible by categorizing PAs into different classes. The most established classes are A, AB, B, C, D, E, F, and inverse F. Generally, the more linear the class is, the less efficient it is. Roughly, the classes fall into two categories: PAs in which the transistor works as a voltage-controlled current source (A to C), and PAs in which the transistor works as a switch (D to inverse F). In the first category, an input sinusoidal signal will, more or less, look like an amplified version of itself at the output. In the second category, the output signal will have lost basically all information that was contained in the amplitude. A very powerful way to correct nonlinearities in PAs is to use digital predistortion (DPD) of the signal. This method uses knowledge obtained though PA calibration to preshape the signal in the digital domain, in a way that counteracts the nonlinearities of the PA. However, DPD may consume a significant amount of power and has its limits. Hence, it cannot always be relied upon to fix the shortcomings of the analog designs.

3.3.1 Class-A PAs

Class-A PAs are among the most straight-forward to design. A PA in class A has a very linear operation, but also the lowest maximum efficiency. However, sometimes the transmitter requires high linearity, which makes class A a good choice despite its low efficiency. Fig. 12 shows class A operation. To ensure that the output signal is always amplified as linearly as possible, the transistors in a class-A amplifier are always on. Since they conduct during the whole period of the sinusoidal input signal, their conduction angle is 360°. The highest efficiency that can possibly be attained is when the output signal is at its maximum voltage swing. However, the maximum drain efficiency will not be reached in real amplifiers, for a number of reasons. The large signals will change the transconductance of the transistors over the period of the signal, and forcing the amplifier to work with too high output voltage while also pushing high currents may cause long-term damage to the circuit. Additionally, the output matching network will add loss, making the maximum theoretical drain efficiency a less realistic goal.



Figure 12: Drain current versus drain-source voltage in class-A, class-B, and class-AB operation. Crossover distortion in class B operation.

3.3.2 Class-AB PAs

The term class-AB is slightly ambiguous. While a class-A PA is linear and has a conduction angle of 360°, a class B PA has a conduction angle of 180°. Traditionally, it is consists of two transistors that work in parallel, and each one is only conducting half of the time. This is achieved by biasing the transistor so that it is not on during the full period. The output will still be quite linear, but suffer from distortion in the transition period, where the voltage is close enough to zero to prevent conduction in any of the transistors. Class B operation is shown in Fig. 12, along with class AB operation. Class AB is just a way of expressing that the PA has a conduction angle between 180° and 360°. It has been demonstrated that advanced CMOS processes are now fast enough to allow mm-wave class AB CMOS PAs that can provide enough linearity for the complex modulations used in the 60 GHz band [16,21].

3.4 Techniques for mm-Wave PA Efficiency Enhancement

Since the design of the class-A PA presented in this thesis, techniques for enhancing the efficiency in mm-wave PAs have demonstrated increasing maturity. Some examples are given below.

Pushing into deep class AB operation is one way of increasing the efficiency, which have demonstrated good overall performance also at mm-wave frequencies. Except for the negative impact on linearity, it may also cause the power gain to drop. However, an attractive way of increasing the gain and stability of a PA is to use capacitive cross-coupling neutralization, which counteracts the gate-drain capacitance [22]. Figure 13 shows capacitive cross-coupling neutralization in a differential PA. Examples of mm-wave PAs in class AB can be found in [16, 21, 23–25].

Another example is outphasing architectures, where the signal is split into two parts before the PA: one contains the phase information of the signal, and the other one contains the amplitude information. The signals are then amplified separately using nonlinear, but efficient switching PAs. Finally, the signals are combined before the antenna. This architecture has attracted a lot of interest [26–28], and challenging design aspect is the implementation of the final combiners, which may introduce loss and distortion.

Progress in efficiency enhancement has also been made by using dual-mode PAs, which can switch between high-power and low-power mode depending on the momentary transmitter need [23, 29–31]. Some also include frequency reconfigurability [32]. They suffer from the same kind of difficulties as the outphasing designs, with challenging output power combiner design and layout floor planning, but demonstrate state-of-the-art performance.

Even Doherty PAs, a topology that has proven invaluable at lower frequencies to battle low efficiencies at power back-off, have been successfully demonstrated in CMOS at mm-wave frequencies, for instance in [33–36]. However, Doherty PAs are difficult to adapt to wideband operation.



Figure 13: Example of cross-coupling neutralization in a differential PA. The optimal neutralization capacitance matches the parasitic gate-drain capacitance of the transistors.

3.5 Techniques for High Output Power

To obtain high output power from CMOS PAs, one of two paths is typically chosen [37]. The first is to use a high supply voltage and stacked transistors, which is possible without exceeding breakdown voltages if a silicon-on-insulator (SOI) CMOS technology is used [38, 39]. The second is to use passive power combination networks to add the power from several individual PAs. Although the power combination networks are inevitably lossy and degrade the efficiency, they are capable of achieving high maximum output powers [40–42]. An example of a parallel-series combiner is shown in Fig. 14.



Figure 14: Example of a series power combiner for CMOS PAs.

Chapter 4

mm-Wave Frequency Generation

An integral part of any transceiver is the local oscillator. In a homodyne transceiver design, it provides the carrier frequency signal, onto which the data to be transmitted is modulated. In a superheterodyne transceiver, one local oscillator signal is required for each mixer. In each case, the LO needs to be able to produce a wide range of frequencies with enough output power, and add as little noise as possible to the signal to be transmitted.

This chapter will first discuss voltage-controlled oscillators (VCOs), and then phase-locked loops (PLLs) will be introduced. For both, some common metrics, design challenges, and solutions at mm-wave frequencies will be presented.

4.1 Voltage Controlled Oscillators

An oscillator can generally be described as a highly non-linear circuit that converts dc power into an output ac waveform. The frequency of its output signal is set using a control signal, which is most commonly a voltage. This chapter briefly covers the VCO fundamentals and a closer look at the crosscoupled VCO topology, along with some design considerations and examples.

4.1.1 VCO Metrics

When designing a VCO for a transceiver system, the most common design considerations concern the following properties of the VCO.

The **Frequency range** (Hz) of the VCO declares what frequencies the VCO can synthesize. Unless stated otherwise, it is inferred that all frequencies within the range are possible to create. In the middle of the frequency range is the **center frequency** (Hz), which should coincide with the center frequency of the targeted frequency band. The VCO **tuning range** (Hz), is expressed in GHz as $f_{max}-f_{min}$, or in percent as $100 \cdot 2 \cdot (f_{max} - f_{min})/(f_{max} + f_{min})$. Designing the tuning range so that the VCO covers a wider frequency span than the intended frequency band is needed to account for variations due to process and operation temperature. In practice, some connections between the

two exist. For instance, achieving a higher tuning range is more difficult at a higher center frequency because parasitic capacitances will limit the range of the variable capacitance. Connected to the frequency range is also the VCO gain, defined as output frequency change per control voltage change (Hz/V). A high gain makes it possible to cover a wide tuning range, but may also be problematic if the VCO gets too sensitive to small variations, i.e. noise, on the control voltage. The ideal frequency vs control voltage function within the frequency range is a straight line with a constant gain, but in practice the gain often varies. For instance, if an NMOS varactor is used for the frequency tuning, the result will be an S-shaped curve with less gain close to the minimum and maximum input control voltage.

Depending on the intended use of the VCO, some **output requirements** on waveform, power, voltage swing and drive capability must be met. The VCO will drive mainly capacitive loads, such as mixers, dividers or buffer amplifiers. Generally, the loss in a passive mixer will be less, the higher the input voltage swing from the VCO is. In an ideal case, the VCO output would be a square wave that switches as abruptly as possible. In practice, the output will be more of sinusoidal shape because the higher harmonics will be filtered out by the relatively narrow-band VCO.

The **Phase noise** (PN) of the VCO profoundly affects the performance of the whole transceiver circuit. The phase noise metric describes how much the frequency of the VCO signal statistically deviates from the intended frequency. In practice, all VCOs will exhibit this kind of noise, and it has trade-offs with both the tuning range and the power consumption. The phase noise is measured in dBc/Hz at some offset from the VCO oscillation frequency, often expressed as $\mathscr{L}(\Delta f)$.

The **power consumption** (P_{dc}) , in W, of a VCO is always important, especially if the VCO is to be a part of hand-held battery-powered devices. It is also an important parameter to consider, as both the phase noise and the tuning range can be directly improved by increasing the power consumption.

The VCO may also be more or less sensitive to unintended inputs from its surrounding environment. **Pushing** and **power supply rejection ratio** (PSRR) concerns the sensitivity to changes in the supply voltage, like a voltage drop or supply voltage noise. **Pulling** is when the VCO frequency is changing with the load impedance.

The VCO figure-of-merit (FOM) combines some of the most important metrics above into one number, to facilitate comparison between different VCO designs. To assign fair weights to the different parameters, the most used FOM is built upon the heuristic Leeson model of phase noise in oscillators [43], where known trade-offs in VCO design are established. This FOM has unit dBc/Hz and it is calculated as

$$FOM = \mathscr{L}(\Delta f) - 20log(f_{osc}/\Delta f) + 10log(P_{dc}/1mW),$$
(5)

where $\mathscr{L}(\Delta f)$ is the phase noise in dBc/Hz, measured at distance Δf Hz from the carrier oscillation frequency f_{osc} in Hz, and P_{dc} is the power consumption in W. Obviously, a FOM will always be general as it does not take into account the specific requirements of individual applications.

4.1.2 VCO Fundamentals

The simplest model of a basic VCO is a linear feedback model, as shown in Fig. 15. Considering only the dependence of frequency in the amplification $A(\omega)$ and the feedback path $\beta(\omega)$, the relation between input and output voltage can be expressed as

$$V_{out} = \frac{A(\omega)}{\left(1 - \left(A(\omega) \cdot \beta(\omega)\right)\right)} V_{in} \tag{6}$$

When some frequency ω_{res} makes the denominator of Eq. (6) equal to 0, the gain will also be infinite, and an oscillation will spontaneously start. An oscillation can also start at all frequencies within the 3-dB frequency bandwidth of the circuit, provided that it is initiated by the ever-present device noise. However, as the oscillation amplitude increases and eventually stabilizes at some amplitude, both the amplification and the feedback in the system also become dependent on the oscillation amplitude V_{osc} . The criteria for oscillation will then instead become $|\beta(\omega_{osc})A(0)| > 1$ and $|\beta(\omega_{osc})A(V_{osc})| = 1$ and $\angle(\beta(\omega_{osc})A(V_{osc})) = 360^{\circ}$. In other words, the gain around the loop exceeds 1 at startup, and the oscillation is sustained when the phase around the loop is 360° and the gain around the loop is 1.

4.1.3 The Cross-Coupled Oscillator

A very commonly used architecture for CMOS VCOs, appreciated for its versatility and ease-of-use, is the cross-coupled design. One of its most basic forms is shown in Fig. 16, but countless variations of its basic structure exist. This



Figure 15: A linear feedback model of a VCO.



Figure 16: Example of a basic cross-coupled VCO.

VCO generates a differential output signal, taken from the drain terminals of the NMOS devices M_1 and M_2 . The signal in the example figure will be sinusoidal, with some harmonic content. The signal will ideally swing around VDD, with a swing of VDD.

The self-oscillation frequency is set by the tank, which consists of an inductance L_t and a capacitance C_t . Both of these circuit elements will also contribute to a parallel loss resistance R_P . The loss resistance R_P will also include the output resistance of the transistor, r_o . C_t also includes the parasitic capacitances C_{par} . The self-oscillation frequency will be $f_{osc} = 1/(2\pi\sqrt{L_tC_t})$. To be able to tune the self oscillation frequency, a part of C_t is often made variable, and in the example in Fig. 16, the variable part is a varactor.

At the self-oscillation frequency, the value of the tank impedance will be just R_P . The voltage gain of each common-source stage is then $-g_m R_P$. Hence, the oscillation criterion becomes $g_m R_P > 1$. In practice, g_m is always designed with some margin to account for process and temperature variations. A variable current source at the source nodes of M1 and M2, as depicted in Fig. 16, provides an additional possibility to adjust g_m . In some applications, however, it is simply omitted to permit more voltage headroom in the circuit.

4.1.4 VCO design for Low Phase Noise

Phase noise is a very central concept in VCO design. It has been extensively researched for many decades [43–46], and new research into the phenomenon still results in new insights [47,48].



Figure 17: Phase noise.

Ideally, the output of the VCO is a single, stable frequency. If this was the case, the signal that is up- or downconverted with the VCO signal will be a perfect, frequency-translated version of itself, and even closely spaced channels can coexist without interference. However, in reality there will be random phase fluctuations. Intuitively, the signal more often stays close to its intended frequency, and more rarely occurs further away from it. Hence, the output of the VCO has a skirt-like spectrum centered around the oscillation frequency, see Fig. 17. In a transmitter, it will lead to spectral regrowth, and it can make symbols in complex modulation schemes uninterpretable. Since the LO is often also used for the receiver chain, it is worth noting that a receiver is even more vulnerable to LO phase noise.

A complete analytical model of phase noise in oscillators in every silicon technology is very difficult to develop, since the noise is a small perturbation of a large signal present at the transistors, and hence small-signal approximations can not be used. Even if the noise cannot be expressed in an analytical model, measurements and modern simulation tools can still be used to predict and understand phase noise. Already in 1966, Leeson formulated a simple model that describes phase noise in LC feedback oscillators [43]:

$$\mathscr{L}(\Delta f) = 10 \log \left[\frac{FkTB}{2P_{carrier}} \left(1 + \left(\frac{f_{osc}}{2Q\Delta f} \right)^2 \right) \left(1 + \frac{f_{corner}}{\Delta f} \right) \right]$$
(7)

where $\mathscr{L}(\Delta f)$ is the single-sided phase noise at distance Δf Hz from the carrier, F is the amplifier noise factor, k is the Boltzmann constant, T is the temperature in Kelvin, B is the bandwidth of integration for the noise (typically set to 1 Hz), f_{osc} is the frequency of oscillation in Hz, Q is the quality



Figure 18: Phase noise regions.

factor of the tank, and f_{corner} is the flicker noise corner. To summarize, there are three distinguishable regions of the phase noise behaviour, see Fig. 18. Far from the carrier, the phase noise will be effectively white. Closer to the carrier, inside the bandwidth set by the Q of the tank, the phase noise will increase as $1/f^2$, i.e. 20 dB/decade. In the region closest to the carrier, the flicker noise will be dominant, and the phase noise will increase as $1/f^3$. Most often, the regions closest to the carrier are the ones of interest, and for VCOs, the noise at 1 or 10 MHz offset from the carrier is often quoted. The frequency that separates the $1/f^2$ and $1/f^3$ regions is called the flicker noise corner.

From Eq. 7, it can be concluded that the most influential contributor to phase noise is the Q value of the tank, and the lower the value, the higher the phase noise. Hence, trying to create a more wideband VCO using a tank with a lower Q-value results in high phase noise. Also, the VCO voltage swing should be maximized to reduce phase noise, but still stay within safe operation limits of the transistors. The noise from each individual transistor should also be kept as low as possible, as should the noise originating from outside sources. Among these are noise from buffers, on the supply voltage or on the control voltage input. The amplitude limitation mechanism of the VCO also plays a part, as the noise performance will be degraded when the transistors stay in triode for too long during each cycle. The phase noise performance will be optimized when the resonance tank is on the verge of being current limited. As identified in [49], the NMOS transistor in the tail current source will spend most of its time in triode, meaning that the current will not stay constant, and neither will the drain current of the cross-coupled transistors. Effectively, the VCO will sample the noise at twice the oscillation frequency when the differential voltage is zero, and actually half the noise from the current source lie around frequencies



Figure 19: Basic VCO variations. (a) A low phase noise VCO with current source filtering. (b) A push-pull VCO for low power consumption.

close to the second harmonic. To address this, a very useful variation on the cross-coupled topology was presented in [50], see Fig. 19(a). It inserts a filter around the current source, tuned to twice the oscillation frequency. It also introduces a capacitor in parallel with the current source, to provide the noise from the current source transistor with a short path to ground. In practice, it trades area for phase noise, but this also makes it practical at mm-wave frequencies, because of the small geometries at shorter wavelengths.

4.1.5 VCO design for Low Power Consumption

As previously discussed, there is a trade-off between phase noise and power consumption. However, instead of sacrificing the phase noise performance, some other options exist. Several ultra-low power mm-wave VCOs based on the cross-coupled topology have recently been demonstrated [51–53]. One example that achieves low power consumption, by increasing the Q-value of the tank by increasing the size of the inductance and reducing the varactor size, is presented in [51]. As expected, the drawback is a very small tuning range. Many demonstrated ultra low-power VCO designs use lowered supply voltages [52,53], which can lead to excellent power consumption and low phase noise. However, the output voltage swing will be degraded, and a buffer may be required to follow the VCO, ultimately adding to the power consumption. A further drawback if this VCO design is that when put in a transmitter, a separate voltage domain has to be created. Another way to go is to extend the cross-coupled topology with another cross-coupled pair of PMOS transistors. An example of this topology, with PMOS transistor M3 and M4, is shown in Fig. 19(b). As demonstrated in Paper IV in this thesis, it can achieve ultra-low power consumption. One drawback is that the maximum output voltage swing will only be between 0 and VDD, but this can be preferable to ensure long-term stable operation of the VCO.

4.1.6 VCO design for Wide Tuning Range

If no extra measures are taken to expand the tuning range, a CMOS crosscoupled oscillator at 60 GHz with a reasonably sized varactor as the frequencycontrol mechanism typically has around 10 % tuning range. One reason for this is that varactors usually have low Q values at mm-wave frequencies, and that increasing the tuning range by increasing the varactor size would decrease the parallel resistance in the tank. To compensate for the lower signal swing, the cross-coupled pair transistors have to be larger, which in turn increases their parasitic capacitance, limiting their tuning range and lowering the maximum oscillation frequency.

Even if 10 % would be enough to precisely cover the ISM band in parts of the world, a wider tuning range is needed to ensure that the band is covered, also when variations in process, supply voltage and temperature cause the frequency range to shift. Hence, it is obvious that special measures must be taken to increase the tuning range, and during the past years this has attracted a lot of research interest [54–57].

Some main approaches can be distinguished. One is to circumvent the problem by designing a VCO that is working at lower frequency, and feed its signal to a multiplier, which creates the desired frequency [54]. It facilitates the VCO design, but the noise increases when the signal is upconverted. However, extracting and boosting a harmonic signal directly from a VCO as in [55], has demonstrated good phase noise performance as well. If the parasitic capacitances can be kept small and the Q-value high enough, it is possible to use a VCO directly at the intended mm-wave frequency and use a capacitance bank for the coarse tuning, as is commonly used in VCOs at lower frequencies. An example of this is [56], that employs a capacitor bank and a back-gate controlled varactor. Another solution is to create some coarse tuning by switching in different inductors in the tank, as demonstrated in for example [57]. The concept is presented in Fig. 20(a). The main challenges of these designs are the layout around the extra inductors, and the inevitable extra loss added by



Figure 20: Basic VCO variations for increased tuning range. (a) A wide tuning range VCO with switched inductances. (b) A wide-band VCO with a transformer.

the switches. In [58] and [59] the effective impedance is tuned by changing the magnetic coupling between different coils, and yet another approach is to make the tank inductance part of a transformer, to be able to change the effective impedances in the tank by switching in different loads through the transformer [60]. An example of this concept is shown in Fig. 20(b).

Most of these methods for extending the tuning range would be compatible with the current-source filtering technique for improving the phase noise performance that is investigated at mm-wave frequencies in this thesis. This could lead to VCOs that are both wide-band and low-noise.

4.2 mm-Wave Phase-Locked Loops

Instead of simply using a VCO as the local oscillator in a tranceiver, it is common practice to instead employ a PLL for the frequency generation. A PLL is a circuit that contains a VCO to synthesize a frequency, but at the same time decreases the VCO phase noise at frequencies close to the carrier. The PLL is



Figure 21: Basic PLL.

able to do this because the control voltage to the VCO will be controlled by a feedback loop that works in the phase domain, and forces the phase noise of the VCO to follow the phase noise of an input reference frequency. Since the reference frequency is generated at a lower frequency, it can be very clean from noise, especially if it is generated with a crystal oscillator. The basic concept of a PLL is shown in Fig. 21. The phase detector will detect any difference in phase between the input reference frequency, f_{ref} , and the divided version of the VCO frequency, f_{fb} . The output of the phase detector, V_{ctrl} will be the control voltage of the VCO.

4.2.1 PLL metrics

The following metrics are the most important in PLL design.

As for the VCOs, the **tuning range**, in GHz or percent, and the **center frequency** in GHz, are important parameters. Naturally, so is the **power consumption**, P_{dc} , in W. Also similar to the VCO, the phase noise, in dBc/Hz, is an important measure. In a PLL, the phase noise will often be presented as integrated over a limited frequency range and related to the output frequency. This is the **integrated average jitter**, called root-mean-square (RMS) jitter, which is measured in seconds. Depending on what PLL topology is used, different levels of unwanted spurs can appear in the output spectrum. Hence, the **output spur levels** in dBc must also be taken into account. Finally, the **lock time**, i.e. the time it takes for the PLL to lock to the desired output frequency in case of a frequency change, is an important parameter. For radars, another way to measure the speed of the PLL is often used, called the **slope**, in Hz/s.

4.2.2 PLL Architectures

The most popular type of PLL, for any application, was for many decades the analog PLL. Recently, subsampling PLLs [61,62], and also all-digital PLLs [63, 64] have made their way also into mm-wave designs and demonstrated excellent performance. However, this thesis will focus on analog PLLs, specifically the



Figure 22: Type-II PLL. (a) Architecture. (b) PFD implementation with D-flipflops. (c) Waveforms in the phase-frequency detector and charge pump.

well-known charge-pump or type-II PLL, and ways to improve its architecture and parts.

Figure 22(a) shows a very common way to implement a PLL. An implementation of a phase-frequency detector (PFD) using resettable, edge-triggered D-flipflops is shown in Fig. 22(b). The PFD reacts to the phase difference on its inputs, se Fig. 22 (c). If the feedback signal arrives later than the reference signal, it will produce an output voltage on the UP terminal, lasting from the rising edge of the reference signal and until the rising edge of the f_{fb} signal. Ideally, it will not produce a DN signal, but in practical implementations there will be a short glitch. The UP signal will turn the upper switch on, and because the DN switch is off, current will flow into the filter capacitor, increasing the voltage in the node V_{ctrl} . This in turn changes the frequency of the VCO, and if the loop is connected properly, makes the VCO output frequency closer to the correct one. Similarly, if the feedback signal leads the reference signal, the PFD will produce output voltage on the DN signal, and the lower switch will be on, and unload current from the filter capacitor and lower the value of V_{ctrl} .

The closed loop transfer function in the Laplace domain, for small signals, can be expressed as [65]

$$H(s) = \frac{\frac{I_{cp}K_{VCO}}{2\pi C_1} \left(R_1 C_1 s + 1\right)}{s^2 + \frac{I_{cp}K_{VCO}}{2\pi N} R_1 s + \frac{I_{cp}K_{VCO}}{2\pi C_1 N}}$$
(8)

where I_{cp} is the charge pump current, K_{VCO} is the gain of the VCO, R_1 and C_1 are the values of the filter components, and N is the division number in the feedback division chain. The charge-pump PLL has two poles in its transfer function, hence the name type-II PLL. Rewriting the denominator on the form $s^2 + 2\zeta\omega_n s + \omega_n^2$, where ζ is called the damping factor, and ω_n is called the natural frequency, gives

$$\zeta = \frac{R_1}{2} \sqrt{\frac{I_{cp} C_1 K_{VCO}}{2\pi N}} \tag{9}$$

$$\omega_n = \sqrt{\frac{I_{cp}K_{VCO}}{2\pi C_1 N}} \tag{10}$$

and that the closed-loop poles are given by $\omega_{p1,2} = [-\zeta \pm \sqrt{\zeta^2 - 1}]\omega_n$. Having established that a PLL is a negative feedback system, and using these equations as a base, it is possible to look into the stability, noise performance, and tradeoffs in the design of a charge-pump PLL.

The PLL will reduce the noise by tying it to the input reference phase noise, but only close to the output frequency, within the bandwidth of the PLL. Here, the phase noise of the free-running VCO will be shaped by the transfer function. However, inside the bandwidth, the reference phase noise in a charge-pump PLL will be scaled up as N^2 , and the noise from other circuits in the PLL, such as the charge pump, will be added. Hence, the bandwidth is often chosen to be around the intersection of the free-running VCO phase noise and the reference phase noise, see Fig. 23(a). However, other considerations also apply when deciding the bandwidth of the PLL. For instance, an oftenused rule-of-thumb [66] is that the bandwidth should never be more than 1/10of the reference frequency f_{ref} to ensure that the PLL acts as a continuoustime system. The bandwidth is also linked to the settling time of the PLL, i.e. how fast it will lock to a new frequency if a step is introduced in the loop. For small ratios between f_{ref} and the loop bandwidth, the settling time decreases with increasing bandwidth. However, for high ratios a phenomenon called cycle slips will make the setting time increase again. Figure 23(b) shows the settling behaviour of a PLL with visible cycle-slips. For f_{ref} -to-loop bandwidth ratios of



Figure 23: PLL loop dynamics. (a) Noise shaping in a charge-pump PLL. (b) Settling behaviour with and without cycle slips.

a hundred or more, the settling time will be excessively long [67]. Cycle slips are difficult to describe analytically, because they happen when the PFD is driven out of its range, and small-signal approximations are no longer valid. However, simulation tools are useful to evaluate the transient behaviour of a PLL design from a cycle-slip perspective. The bandwidth is also linked to the phase margin of the PLL, which needs to be optimized for stability and speed. The stability, i.e. the phase margin, must be designed with some extra headroom for process variations, but also because it is often degraded in practical charge-pump PLLs by an extra capacitor that is added to the filter. The reason for this addition is to remove ripple on V_{ctrl} , created by imperfections in the PFD, that result in very short and often mismatched pulses on V_{ctrl} , even when the loop has stabilized. The maximum reasonable size of the extra capacitance is often set to be 20 % of C_1 , which maximizes the phase margin of the PLL, if ζ is chosen to be around 0.7. Additionally, introducing an extra capacitance also sets a maximum value for R_1 in the loop filter [65].

From the brief explanation above, it is clear that there are many degrees of freedom when designing a charge-pump PLL. As in any design, the target application has to decide the system requirements. The environment will set what output frequencies the PLL must be able to synthesize, what input reference sources are available, if fractional-N operation is necessary, if the VCO needs to have a direct I/Q output, if the PLL must provide an extra modulation possibility, or phase shifting capabilities, and much more. The considerations here also tie into considerations for the entire tranceiver. Additionally, even if the architecture is optimized, each individual subcircuit of the PLL must also be carefully designed.

Some notable published CMOS implementations of charge-pump PLLs for the mm-wave frequency range from recent years can be found in [55, 68–72]. In [68], a fractional-N PLL for 5G applications demonstrates state-of-the-art noise performance, and in [69], a quadrature 60-GHz PLL with an inductorless divider chain and an in-phase injection-coupled VCO is presented. A successful strategy used in PLLs for the 60-GHz band presented in [55,70–72] is to design the PLL for operation on a lower frequency. To reach the 60-GHz band [71] uses a frequency tripler after the VCO, [70] presents a sub-harmonic quadrature injection-locked VCO, and in [55] the third harmonic of the VCO in a 20-GHz PLL is extracted and boosted. The PLL in [72] operates at RF frequencies, at a few GHz, and uses a set of different multipliers to cover many mm-wave bands.

4.2.3 Charge Pumps

The charge-pump concept is presented in Fig. 22(a). The main concern in charge-pump design is how to minimize pulse mismatch, i.e. when the current waveforms of the UP and DN pulses are not perfectly matched. It leads to ripple, noise, and spurs in the PLL output frequency spectrum. The problem sometimes originates from unmatched signal delays from the PFD, but there are many possible inherent causes within the charge-pump that may exacerbate the mismatch. Fig. 24 shows a straightforward implementation of a charge-pump. Many variations exist, and propose different ways to deal with imperfections that cause mismatch.

In the charge-pump in Fig. 24, the PMOS switch needs to have an inverted signal on its input to switch on, and as a first precaution against time delay, the inverting buffer in the UP path is matched by a transmission gate in the DN path. Other inherent causes of mismatch are affected by the transistor geometries. The individual transistors of the charge-pump are often made very wide, so that they can deliver current even when V_{ctrl} is close to ground or VDD. Large devices will have more parasitic capacitance and more space for storing charge in their inversion layers. This charge will be absorbed by the current sources when they are on, and dissipated when they turn off, causing unwanted nonlinear effects. Another nonlinear effect in charge-pumps is channel-length modulation, which is more dominant in short-channel devices. The channellength modulation leads to mismatch because as V_{ctrl} changes, so does the drain source voltage of the current sources. This effect will always make one current source more effective at pushing current, while the other is less effective. A popular technique for reducing current mismatch due to channel-length modulation is presented in [73] Finally, random mismatches and transistor noise in the charge-pump also play a part in the PLL.

4.2.4 Dividers

The division in the PLL is done in steps, and often one or more of the steps are programmable to enable the PLL to synthesize different frequencies without



Figure 24: Charge pump transistor implementation.

having to change the input reference frequency. The most common choice for controlling these dividers is a sigma-delta modulator, which changes between division steps very fast and creates the desired frequency when averaged out. Closest to the PFD, the signal will have a lower frequency and more conventional digital divider designs can be used. At this position in the chain, the dividers can be programmable pulse-swallow dividers or dual-modulus dividers. Closer to the VCO, the frequencies that the dividers have to handle are higher. Faster logic is needed, such as current-mode logic (CML). A latch-based CML divide-by-two circuit is shown in Fig. 25 [74]. It is used in the divider chain of the PLL in paper V in this thesis, where it demonstrated good performance up to over 20 GHz.

When the divider speed needs to be even higher, as is the case of the first divider after a VCO that is operating at 60 GHz and above, a common choice is to instead turn to a tuned topology called injection-locked dividers (ILFDs). In their core they are similar to VCOs, and consequently often implemented as a cross-coupled VCO. To act as dividers, they are tuned to the divided output frequency, and the input signal to be divided is injected into the VCO in a way that makes it lock to it. Injection locking is of interest in many systems, and there are well-researched limits for when an injection lock can occur. The locking range for an ILFD is expressed as [75, 76]

$$\omega_{range} \le \frac{\omega_0}{2Q} \cdot \frac{2}{\pi} \cdot \frac{I_{inj}}{I_{osc}} \tag{11}$$

where Q is the quality factor of the divider's LC tank, I_{inj} the magnitude of the injected current, and I_{osc} the magnitude of the free running oscillator current at f_0 . It is clear that the more efficient the injection is, the wider the locking range becomes. A varactor can also be used in an ILFD to increase



Figure 25: (a) High-speed latches in a tuned divide-by-two configuration. (b) Individual latch.

the locking range by tuning the circuit at the expense of complexity, loading of the tank, and larger chip area. Figure 26 shows two possible choices for injection of a signal at twice the output frequency. In the variation in Fig. 26(b), a differential signal is injected, which permits the use of the full output voltage swing of the preceding VCO. ILFD with higher division ratios have also been demonstrated. Figure 27(a) shows a divide-by-three ILFD. Here, the injected signal is at three times the self-oscillation frequency, and the transistors used for the injection will act as mixers and mix the input signal at $3f_0$ with the existing self-oscillation frequency f_0 , creating even-order harmonics in the node between the injection transistors, where the second harmonic of the self-oscillation frequency naturally occurs [77]. A small change in this node, suppressing the second harmonic while favouring the fourth harmonic, as is shown in Fig. 27(b), makes the ILFD a divide-by-five circuit instead [78].

The output phase noise of an ILFD will generally be the same as the phase noise of the signal that it is locked on to, but it will increase at the edge of the locking range. Inside the locking range it will generally not introduce spurs, and because of its tuned nature, harmonics of the divided frequency will be suppressed.



Figure 26: Injection-locked divide-by-two circuits. (a) Injection through the current source. (b) Direct differential injection.



Figure 27: Injection-locked circuits. (a) Division-by-three. (b) Division-by-five.

Chapter 5

Paper Summary and Conclusions

This chapter presents summaries and conclusions of the results obtained in the included research papers. The author's contributions to each paper are also stated.

5.1 Summary

Paper I describes a two-stage, class-A PA for the 60 GHz ISM band. Each stage is differential and uses NMOS transistors in common-source configuration. To counteract the significant gate-drain capacitance at mm-wave frequencies, each stage is equipped with capacitive cross-coupling neutralization. It is implemented with NMOS transistors to ensure that it is matched to the transistors whose parasitic capacitances it should counteract. The input stage uses transistors that are dimensioned to cope with the large currents involved. The transistors of the subsequent output stage are 60 % larger. For measurement purposes, the PA has baluns at both the input and output, each carefully tuned to provide good input and output matching. Neither of the baluns are excluded from the calculations of the PAE. The inter-stage matching employs a 2:1 transformer, which is sufficiently wideband. The design was implemented in 65-nm CMOS and the manufactured chip was mounted on a PCB with bond-wire connections for bias and supply voltages. The mm-wave signals were measured using on-chip probing and a network analyzer. The chip photo is shown in Fig. 28. The input and output matching is at their best around 67 GHz, where the gain also peaks. The maximum gain is 16.8 dB and the 3 dB bandwidth is 9 GHz, enough to cover the ISM band at 60 GHz. Operating from a 1.2 V supply voltage, its peak PAE is 18.5 %, well above other two-stage PAs presented before it. It has a maximum output power of 11.8 dBm, and a 1-dB output-related compression point at 8.4 dBm.

Contribution: I did the analysis, simulations, layout, measurement setup, measurements, and the manuscript writing, under supervision of the second and third authors.



Figure 28: Chip microphotograph of the circuit in paper I.

Paper II describes a varactorless VCO, operating in the 60-GHz ISM band. At its core, it is a basic cross-coupled VCO. Instead of using a varactor for the frequency tuning, it changes the effective inductance seen by the cross-coupled NMOS transistors. This is achieved using an extra, in-phase branch, with its inductance placed right outside the coil of the primary inductance. The mutual inductance is dependent on the coupling factor and on the ratio between the currents in the two branches. The design was implemented in 65-nm CMOS and the manufactured chip was mounted on a PCB with bond-wire connections for bias and supply voltages. The mm-wave signals were measured with an onchip probe, a spectrum analyzer, and a phase noise analyzer. The chip photo is shown in Fig. 29. Changing the currents in branch 1 and 2, and consuming 6-30 mA from a 1.2 V supply, the frequency can be tuned from 62.2 to 65.1 GHz. The measured phase noise stays flat across the tuning range, and hence the best FOM of -182.4 dB/Hz is found at the highest frequency, when the current consumption is at its lowest. In all, the circuit demonstrates that a varactorless structure can lead to good phase noise performance. However, the architecture inherently narrows the tuning range.

Contribution: I did the analysis, simulations, layout, measurement setup, measurements, and the manuscript writing, under supervision of the second and third authors.

Paper III presents an empirical study of the efficiency at 60 GHz of a wellknown technique of suppressing phase noise in VCOs. Two cross-coupled VCOs are manufactured: one baseline VCO, and one identical design except for the added noise source filtering at the current source. The filtering is implemented



Figure 29: Chip microphotograph of the circuit in paper II.

with an inductance above the current source, tuned to twice the frequency of oscillation, to prevent the resistance of the cross-coupled pairs from loading the resonator in the switched stage. A capacitance is also inserted in parallel with the current source, providing a low-impedance path to ground for the noise generated in the current source. As this method of lowering the phase noise trades phase noise for area, it is especially interesting at 60 GHz, where inductors are small. However, the second harmonic will be at 120 GHz, closer to f_T and f_{max} of the technology, and it was not previously investigated if this would impact the efficiency of the method. Simulations show that the Q-value of the source inductance has a small impact on the efficiency, but the value of the inductance is more important. The VCOs were implemented in 28-nm FD-SOI CMOS and the manufactured chip was mounted on a PCB with bond-wire connections for bias and supply voltages. The mm-wave signals were measured with an on-chip probe, a spectrum analyzer, and a phase noise analyzer. The chip photo is shown in Fig. 30. At 0.9 V supply, both VCOs consume 3.15 mW, and the filtered VCO achieves a FOM of -187.3 dBc/Hz, which was state-of-the-art performance at the time of publication. Both VCOs have a tuning range of about 11 %, and even with the extra inductance in the filtering VCO, their areas are competitive. The improvement originating from the filtering was measured to between 5 and 12 dB across the tuning range.

Contribution: I did the analysis, simulations, and layout. The second and fifth authors did the measurements. I did the manuscript writing, with assistance from the second and third authors. All was done under the supervision of the last two authors.



Figure 30: Chip microphotograph of the circuits in paper III.

Paper IV presents two ultra low-power VCOs for the 30 and 60 GHz bands. They both use a push-pull variation of the cross-coupled NMOS oscillator topology. The output voltage swing is restricted to the interval between 0 and VDD, which is lower than what can be achieved by the traditional crosscoupled architecture. Other potential drawbacks are that using a varactor for tuning may not be as efficient as in a basic cross-coupled topology, and that using two stacked transistors leaves less voltage headroom for the current source. The VCOs were implemented in 28-nm FD-SOI CMOS and the manufactured chip was mounted on a PCB with bond-wire connections for bias and supply voltages. The mm-wave signals were measured with an on-chip probe, a spectrum analyzer, and a phase noise analyzer. The chip photos are shown in Fig. 31. Both VCOs have tuning ranges around 11 % in their intended bands. The measured power consumption at 1 V for the 30 GHz VCO is 1.06 mW, and 1.35 mW for the 60 GHz VCO. Hence, they both demonstrate ultra-low power consumption, that is on par with, or even lower, than previously published sub-threshold VCOs, but without the need for an additional voltage domain. They both have excellent FOMs, -188.4 dBc/Hz for the 30 GHz VCO, and -186.2 dBc/Hz for the 60 GHz VCO.

Contribution: I did the analysis, simulations, layout, measurement setup, measurements, and the manuscript writing, under supervision of the second, third and fourth authors.

Paper V presents a mm-wave analog phase-locked loop with a VCO operating around 55 GHz, with very low power consumption and an improved lock time. The intended application for the PLL is integration into a 5G transceiver,



Figure 31: Chip microphotograph of the circuits in paper IV.

where another PLL for the RF bands already exists. This RF PLL can be used as the input for the mm-wave PLL and be responsible for channel selection and modulation, alleviating the demands of fractional-N division in the mm-wave PLL. At a first glance, the noise from the mm-wave PLL benefits from having a high input reference frequency of around 2.2 GHz, since the in-band noise adds as the division ratio squared. It also seems to imply that the PLL can use a higher bandwidth, and thus achieve a faster lock time. However, to keep the noise low, the bandwidth cannot be increased as much, and a low bandwidth coupled with a high reference frequency leads to cycle-slips and an excessively long lock-time. To address this problem, the proposed PLL employs a modeshifting scheme, where it can operate in either a fast-locking or a low-noise mode. Both modes keep the same small-signal parameters, to ensure stability in both modes. When using the fast-locking mode to acquire a lock, the current in the charge pump is temporarily increased by a factor of 8, and the input reference frequency, as well as the divider chain in the PLL, are extended with divide-by-eight circuits. The PLL VCO uses the low-noise source-filtering design from Paper III. The VCO is followed by a divide-by-three injection-locked circuit. Traditionally, these are difficult to make wideband enough. However, the paper presents a novel double-injection topology that reuses the second harmonic present between the two injection transistors in parallel with the tank. This second harmonic signal is fed to the current source input, which creates a second injection path and increases the injection efficiency. Altogether, it makes the ILFD wideband enough to cover the whole VCO tuning range, while only consuming 0.5 mW. To verify the efficiency of the doubleinjection ILFD topology, two ILFDs are manufactured stand-alone, one with







(b)

Figure 32: Chip microphotographs of the circuits in paper V. (a) The PLL, overlayed with a semi-transparent layout. (b) The two versions of the ILFD.

and the other without the double-injection path, to facilitate a fair comparison. The PLL also includes a charge-pump with a novel current mismatch mitigation-technique based on negative feedback, and an improved operational amplifier that allows operation over a large common-mode range. Both versions of the ILFD, and the PLL, were implemented in 28-nm FD-SOI CMOS and the manufactured chips were mounted on PCBs with bond-wire connections for bias and supply voltages. The mm-wave signals were measured with on-chip probes, a spectrum analyzer, and a phase noise analyzer. The chip photos are shown in Fig. 32. Measurements of the two stand-alone ILFDs show that the locking range of the double-injection topology is almost doubled compared to the single-injection topology at a fixed varactor voltage and at the same power consumption. Measurements of the PLL show that it has a record low total power consumption of 10 mW. Together with an excellent integrated jitter of 176 fs, it achieves a FOM of -245 dB. Additionally, it has a lock time of 3 μ s, demonstrating that the architecture can overcome problems with excessive lock time associated with a high input reference frequency.

Contribution: Together with the first author, I did the analysis, simulations, layout, measurement setup, measurements, and the manuscript writing. My primary responsibility was the mm-wave parts, but since no part of a PLL is independent of the others, we were both involved in all parts of the system. All was done under the supervision of the third and fourth authors.

Paper VI presents an injection-locked divide-by-two circuit for the 60 GHz band, stand-alone as well as integrated with the low-noise VCO from Paper III. The ILFD consists of a cross-coupled oscillator, which is injection locked by injecting a signal at around twice the frequency of self-oscillation. The efficency of the injection mechanism determines how wide the locking range is. To be able to use the full swing of a differential VCO, and create an efficient mechanism of injection, the signal is fed to the oscillator by direct differential injection. Simulations show that this creates a very wide locking range, which more than covers the tuning range of the VCO, even with process variations. One ILFD is integrated with the VCO to verify this. Both circuits were implemented in 28-nm FD-SOI CMOS and the manufactured chip was mounted on a PCB with bond-wire connections for bias and supply voltages. The mm-wave signals were measured with using on-chip probing, a spectrum analyzer, a phase noise analyzer, and a network analyzer. The chip photo is shown in Fig. 33. Measurements show that the widest locking range of 30%, at -1.5 dBm of input power for the ILFD, is achieved at a power consumption of 4.3 mW from a 0.9 V supply. If the possibility to measure with even higher input power had been available, an even higher locking range could likely have been reported. Put together with the 60 GHz low-noise VCO from Paper III, the tuning range is set by the VCO to 10.2 %. The best measured phase noise from the VCO and ILFD combination is -111 dBc/Hz at 10 MHz offset from

the $28.5~\mathrm{GHz}$ carrier.



Figure 33: Chip microphotograph of the circuits in paper VI.

Contribution: I did the analysis, simulations, layout, measurement setup, measurements, and the manuscript writing, under supervision of the second, third and fourth authors.

Chapter 6

Discussion and Future Work

This dissertation presents several CMOS circuits that have been designed to be efficient parts of mm-wave transmitters. They have all been manufactured and verified by measurements.

A natural next step is to combine the different presented parts into complete, efficient transmitters. The presented PA, which demonstrates good performance when measured stand-alone, has already been integrated with mixers, a quadrature VCO and efficient injection-locked negative-resistance buffers into a sliding-LO transmitter for the 60 GHz band. It has been manufactured and my measurements show that the signal is properly upconverted and that the transmitter has wideband operation. However, the conversion gain is low. This was later confirmed in simulations to be mainly due to a mistuning of a passive structure, which reduces the input signal power to the PA. This structure could easily be re-tuned and manufactured again, as long as the silicon process is not discontinued. An improvement of this transmitter circuit should include a PLL with phase-shifting capability around the quadrature VCO, to make it viable for a 60 GHz beam-steering system.

The main drawback of the presented PLL is the limited tuning range. The most straightforward solution would be to increase it by increasing the size of the varactor, and thus accept the penalty of increased phase noise. However, the current-source filtering method used in the VCO to lower the phase noise could fit into many cross-coupled VCO topologies, also those a wider tuning range. At the same time, the PLL could also be extended to include phase-shifting possibilities by injecting current into the charge pump.

The PLL was designed to ultimately be a part of a whole homodyne transmitter for the 5G bands around 30 GHz. The signal from the 60 GHz VCO would be divided by the divide-by-two circuits presented in this thesis, to create a quadrature signal at 30 GHz. The dividers, which are low-power, have enough output power to directly drive a passive mixer for the upconversion. I have started this work and made initial simulations, but the transmitter still needs a power amplifier in the same technology node to be ready for manufacturing. Finally, further integration with PCB antennas and on-chip high-speed DACs, in collaboration with other research groups, would be an interesting direction for future research.

References

- Ericsson AB, "Mobile subscriptions worldwide Q1 2018," https://www.ericsson.com/en/mobility-report/reports/june-2018/mobile-subscriptions-worldwide-q1-2018, Jun. 2018.
- Systems [2] Cisco Inc., "Cisco Visual Networking Index: Global Mobile Data Traffic Forecast Update, 20162021 White Paper," https://www.cisco.com/c/en/us/solutions/collateral/serviceprovider/visual-networking-index-vni/mobile-white-paper-c11-520862.html, Feb. 2017.
- [3] T. Baykas, C. Sum, Z. Lan, J. Wang, M. A. Rahman, H. Harada, and S. Kato, "IEEE 802.15.3c: the first IEEE wireless standard for data rates over 1 Gb/s," *IEEE Communications Magazine*, vol. 49, no. 7, pp. 114– 121, July 2011.
- [4] ITU-R, "Recommendation ITU-R P.676-11, Attenuation by atmospheric gases," Radiocommunication Sector of International Telecommunication Union, Tech. Rep., Sep. 2016.
- [5] "IEEE Standard for Information technology–Telecommunications and information exchange between systems Local and metropolitan area networks–Specific requirements - Part 11: Wireless LAN Medium Access Control (MAC) and Physical Layer (PHY) Specifications," *IEEE Std* 802.11-2016 (Revision of IEEE Std 802.11-2012), pp. 1–3534, Dec 2016.
- [6] Z. Du, E. Ohlmer, K. Aronkytö, J. Putkonen, J. Kapanen, and D. Swist, "5G E-band backhaul system measurements in urban street-level scenarios," in 2017 47th European Microwave Conference (EuMC), Oct 2017, pp. 372–375.
- [7] T. S. Rappaport, S. Sun, R. Mayzus, H. Zhao, Y. Azar, K. Wang, G. N. Wong, J. K. Schulz, M. Samimi, and F. Gutierrez, "Millimeter wave mobile communications for 5g cellular: It will work!" *IEEE Access*, vol. 1, pp. 335–349, 2013.
- [8] "3GPP TR 38.815 V15.0.0 New frequency range for NR (24.25-29.5 GHz)," 3GPP(Release 15), pp. 1–22, July 2018.
- [9] H. Kim, B. Park, S. Oh, S. Song, J. Kim, S. Kim, T. Moon, S. Kim, J. Chang, S. Kim, W. Kang, S. Jung, G. Tak, J. Du, Y. Suh, and Y. Ho, "A 28GHz CMOS direct conversion transceiver with packaged antenna arrays for 5G cellular system," in 2017 IEEE Radio Frequency Integrated Circuits Symposium (RFIC), June 2017, pp. 69–72.
- [10] K. Okada, K. Kondou, M. Miyahara, M. Shinagawa, H. Asada, R. Minami, T. Yamaguchi, A. Musa, Y. Tsukui, Y. Asakura, S. Tamonoki, H. Yamagishi, Y. Hino, T. Sato, H. Sakaguchi, N. Shimasaki, T. Ito, Y. Takeuchi, N. Li, Q. Bu, R. Murakami, K. Bunsen, K. Matsushita, M. Noda, and A. Matsuzawa, "A full 4-channel 6.3Gb/s 60GHz directconversion transceiver with low-power analog and digital baseband circuitry," in 2012 IEEE International Solid-State Circuits Conference, Feb 2012, pp. 218–220.
- [11] G. Mangraviti, K. Khalaf, Q. Shi, K. Vaesen, D. Guermandi, V. Giannini, S. Brebels, F. Frazzica, A. Bourdoux, C. Soens, W. V. Thillo, and P. Wambacq, "13.5 A 4-antenna-path beamforming transceiver for 60GHz multi-Gb/s communication in 28nm CMOS," in 2016 IEEE International Solid-State Circuits Conference (ISSCC), Jan 2016, pp. 246–247.
- [12] T. Sowlati, S. Sarkar, B. Perumana, W. L. Chan, B. Afshar, M. Boers, D. Shin, T. Mercer, W. Chen, A. P. Toda, A. G. Besoli, S. Yoon, S. Kyriazidou, P. Yang, V. Aggarwal, N. Vakilian, D. Rozenblit, M. Kahrizi, J. Zhang, A. Wang, P. Sen, D. Murphy, M. Mikhemar, A. Sajjadi, A. Mehrabani, B. Ibrahim, B. Pan, K. Juan, S. Xu, C. Guan, G. Geshvindman, K. Low, N. Kocaman, H. Eberhart, K. Kimura, I. Elgorriaga, V. Roussel, H. Xie, L. Shi, and V. Kodavati, "A 60GHz 144-element phased-array transceiver with 51dBm maximum EIRP and ±60° beam steering for backhaul application," in 2018 IEEE International Solid -State Circuits Conference - (ISSCC), Feb 2018, pp. 66–68.
- [13] E. Cohen, M. Ruberto, M. Cohen, O. Degani, S. Ravid, and D. Ritter, "A CMOS Bidirectional 32-Element Phased-Array Transceiver at 60 GHz With LTCC Antenna," *IEEE Transactions on Microwave Theory* and Techniques, vol. 61, no. 3, pp. 1359–1375, March 2013.
- [14] J. D. Dunworth, A. Homayoun, B. Ku, Y. Ou, K. Chakraborty, G. Liu, T. Segoria, J. Lerdworatawee, J. W. Park, H. Park, H. Hedayati, D. Lu, P. Monat, K. Douglas, and V. Aparin, "A 28GHz Bulk-CMOS dualpolarization phased-array transceiver with 24 channels for 5G user and basestation equipment," in 2018 IEEE International Solid - State Circuits Conference - (ISSCC), Feb 2018, pp. 70–72.
- [15] T. Chi, J. S. Park, S. Li, and H. Wang, "A 64GHz full-duplex transceiver front-end with an on-chip multifeed self-interference-canceling antenna and an all-passive canceler supporting 4Gb/s modulation in one antenna footprint," in 2018 IEEE International Solid - State Circuits Conference -(ISSCC), Feb 2018, pp. 76–78.
- [16] V. Vidojkovic, V. Szortyka, K. Khalaf, G. Mangraviti, S. Brebels, W. v. Thillo, K. Vaesen, B. Parvais, V. Issakov, M. Libois, M. Matsuo,

J. Long, C. Soens, and P. Wambacq, "A low-power radio chipset in 40nm LP CMOS with beamforming for 60GHz high-data-rate wireless communication," in 2013 IEEE International Solid-State Circuits Conference Digest of Technical Papers, Feb 2013, pp. 236–237.

- [17] A. Axholt and H. Sjöland, "A 60 GHz receiver front-end with PLL based phase controlled LO generation for phased-arrays," in Asia-Pacific Microwave Conference 2011, Dec 2011, pp. 1534–1537.
- [18] C. A. Balanis, Antenna theory: analysis and design. Wiley & Sons, 1997.
- [19] B. Sadhu, Y. Tousi, J. Hallin, S. Sahl, S. Reynolds, O. Renström, K. Sjögren, O. Haapalahti, N. Mazor, B. Bokinge, G. Weibull, H. Bengtsson, A. Carlinger, E. Westesson, J. Thillberg, L. Rexberg, M. Yeck, X. Gu, D. Friedman, and A. Valdes-Garcia, "7.2 A 28GHz 32-element phasedarray transceiver IC with concurrent dual polarized beams and 1.4 degree beam-steering resolution for 5G communication," in 2017 IEEE International Solid-State Circuits Conference (ISSCC), Feb 2017, pp. 128–129.
- [20] K. Khalaf, K. Vaesen, S. Brebels, G. Mangraviti, M. Libois, C. Soens, W. V. Thillo, and P. Wambacq, "A 60-GHz 8-Way Phased-Array Front-End With T/R Switching and Calibration-Free Beamsteering in 28-nm CMOS," *IEEE Journal of Solid-State Circuits*, vol. 53, no. 7, pp. 2001– 2011, July 2018.
- [21] B. Park, S. Jin, D. Jeong, J. Kim, Y. Cho, K. Moon, and B. Kim, "Highly Linear mm-Wave CMOS Power Amplifier," *IEEE Transactions on Mi*crowave Theory and Techniques, vol. 64, no. 12, pp. 4535–4544, Dec 2016.
- [22] R. Minami, K. Matsushita, H. Asada, K. Okada, and A. Matsuzawa, "A 60 GHz CMOS power amplifier using varactor cross-coupling neutralization with adaptive bias," in *Asia-Pacific Microwave Conference 2011*, Dec 2011, pp. 789–792.
- [23] D. Zhao and P. Reynaert, "A 60-GHz Dual-Mode Class AB Power Amplifier in 40-nm CMOS," *IEEE Journal of Solid-State Circuits*, vol. 48, no. 10, pp. 2323–2337, Oct 2013.
- [24] S. N. Ali, P. Agarwal, L. Renaud, R. Molavi, S. Mirabbasi, P. P. Pande, and D. Heo, "A 40% PAE Frequency-Reconfigurable CMOS Power Amplifier With Tunable Gate-Drain Neutralization for 28-GHz 5G Radios," *IEEE Transactions on Microwave Theory and Techniques*, vol. 66, no. 5, pp. 2231–2245, May 2018.
- [25] S. Kulkarni and P. Reynaert, "A 60-GHz Power Amplifier With AM-PM Distortion Cancellation in 40-nm CMOS," *IEEE Transactions on Mi*crowave Theory and Techniques, vol. 64, no. 7, pp. 2284–2291, July 2016.

- [26] D. Zhao, S. Kulkarni, and P. Reynaert, "A 60-GHz Outphasing Transmitter in 40-nm CMOS," *IEEE Journal of Solid-State Circuits*, vol. 47, no. 12, pp. 3172–3183, Dec 2012.
- [27] A. Ghahremani, A. Annema, and B. Nauta, "Outphasing class-e power amplifiers: From theory to back-off efficiency improvement," *IEEE Journal* of Solid-State Circuits, vol. 53, no. 5, pp. 1374–1386, May 2018.
- [28] M. Mehrjoo and J. Buckwalter, "13.4 A microwave injection-locking outphasing modulator with 30dB dynamic range and 22% system efficiency in 45nm CMOS SOI," in 2016 IEEE International Solid-State Circuits Conference (ISSCC), Jan 2016, pp. 244–245.
- [29] A. Larie, E. Kerherve, B. Martineau, L. Vogt, and D. Belot, "2.10 A 60GHz 28nm UTBB FD-SOI CMOS reconfigurable power amplifier with 21% PAE, 18.2dBm P1dBand 74mW PDC," in 2015 IEEE International Solid-State Circuits Conference - (ISSCC) Digest of Technical Papers, Feb 2015, pp. 1–3.
- [30] P. M. Farahabadi and K. Moez, "A Dual-Mode Wideband +17.7-dBm 60-GHz Power Amplifier in 65-nm CMOS," *IEEE Transactions on Components, Packaging and Manufacturing Technology*, vol. 7, no. 12, pp. 1998– 2007, Dec 2017.
- [31] L. Kuang, B. Chi, H. Jia, W. Jia, and Z. Wang, "A 60-ghz cmos dual-mode power amplifier with efficiency enhancement at low output power," *IEEE Transactions on Circuits and Systems II: Express Briefs*, vol. 62, no. 4, pp. 352–356, April 2015.
- [32] C. R. Chappidi, X. Wu, and K. Sengupta, "A digital mm-Wave PA architecture with Simultaneous Frequency and back-off Reconfigurability," in 2017 IEEE Radio Frequency Integrated Circuits Symposium (RFIC), June 2017, pp. 328–331.
- [33] S. Chen, G. Wang, Z. Cheng, P. Qin, and Q. Xue, "Adaptively Biased 60GHz Doherty Power Amplifier in 65-nm CMOS," *IEEE Microwave and Wireless Components Letters*, vol. 27, no. 3, pp. 296–298, March 2017.
- [34] A. Agah, H. Dabag, B. Hanafi, P. M. Asbeck, J. F. Buckwalter, and L. E. Larson, "Active millimeter-wave phase-shift doherty power amplifier in 45-nm soi cmos," *IEEE Journal of Solid-State Circuits*, vol. 48, no. 10, pp. 2338–2350, Oct 2013.
- [35] E. Kaymaksut, D. Zhao, and P. Reynaert, "E-band transformer-based Doherty power amplifier in 40 nm CMOS," in 2014 IEEE Radio Frequency Integrated Circuits Symposium, June 2014, pp. 167–170.

- [36] S. Hu, F. Wang, and H. Wang, "2.1 A 28GHz/37GHz/39GHz multiband linear Doherty power amplifier for 5G massive MIMO applications," in 2017 IEEE International Solid-State Circuits Conference (ISSCC), Feb 2017, pp. 32–33.
- [37] T. Johansson and J. Fritzin, "A Review of Watt-Level CMOS RF Power Amplifiers," *IEEE Transactions on Microwave Theory and Techniques*, vol. 62, no. 1, pp. 111–124, Jan 2014.
- [38] J. Xia, A. Chung, and S. Boumaiza, "A wideband millimeter-wave differential stacked-FET power amplifier with 17.3 dBm output power and 25% PAE in 45nm SOI CMOS," in 2017 IEEE MTT-S International Microwave Symposium (IMS), June 2017, pp. 1691–1694.
- [39] D. Y. C. Lie, J. Tsay, T. Hall, T. Nukala, and J. Lopez, "High-efficiency silicon RF power amplifier design, current status and future outlook," in 2016 IEEE International Symposium on Radio-Frequency Integration Technology (RFIT), Aug 2016, pp. 1–4.
- [40] D. Zhao and P. Reynaert, "An E-Band Power Amplifier With Broadband Parallel-Series Power Combiner in 40-nm CMOS," *IEEE Transactions on Microwave Theory and Techniques*, vol. 63, no. 2, pp. 683–690, Feb 2015.
- [41] —, "21.3 dBm 18.5 GHz-BW 8-way E-band power amplifier in 28 nm high performance mobile CMOS," *Electronics Letters*, vol. 53, no. 19, pp. 1310–1312, 2017.
- [42] C. Tseng and Y. Wang, "A 60 GHz 19.6 dBm Power Amplifier With 18.3% PAE in 40 nm CMOS," *IEEE Microwave and Wireless Components Letters*, vol. 25, no. 2, pp. 121–123, Feb 2015.
- [43] D. B. Leeson, "A simple model of feedback oscillator noise spectrum," *Proceedings of the IEEE*, vol. 54, no. 2, pp. 329–330, Feb 1966.
- [44] A. Hajimiri and T. H. Lee, "A general theory of phase noise in electrical oscillators," *IEEE Journal of Solid-State Circuits*, vol. 33, no. 2, pp. 179– 194, Feb 1998.
- [45] J.-C. Nallatamby, M. Prigent, M. Camiade, and J. J. Obregon, "Extension of the Leeson formula to phase noise calculation in transistor oscillators with complex tanks," *IEEE Transactions on Microwave Theory and Techniques*, vol. 51, no. 3, pp. 690–696, March 2003.
- [46] J.-C. Nallatamby, M. Prigent, M. Camiade, and J. Obregon, "Phase noise in oscillators - Leeson formula revisited," *IEEE Transactions on Microwave Theory and Techniques*, vol. 51, no. 4, pp. 1386–1394, April 2003.

- [47] F. Pepe and P. Andreani, "A General Theory of Phase Noise in Transconductor-Based Harmonic Oscillators," *IEEE Transactions on Cir*cuits and Systems I: Regular Papers, vol. 64, no. 2, pp. 432–445, Feb 2017.
- [48] —, "Still More on the $1/f^2$ Phase Noise Performance of Harmonic Oscillators," *IEEE Transactions on Circuits and Systems II: Express Briefs*, vol. 63, no. 6, pp. 538–542, June 2016.
- [49] A. Hajimiri and T. H. Lee, "Design issues in CMOS differential LC oscillators," *IEEE Journal of Solid-State Circuits*, vol. 34, no. 5, pp. 717–724, May 1999.
- [50] E. Hegazi, H. Sjöland, and A. A. Abidi, "A filtering technique to lower LC oscillator phase noise," *IEEE Journal of Solid-State Circuits*, vol. 36, no. 12, pp. 1921–1930, Dec 2001.
- [51] M. Tiebout, H. . Wohlmuth, and W. Simburger, "A 1 V 51GHz fully-integrated VCO in 0.12μm CMOS," in 2002 IEEE International Solid-State Circuits Conference. Digest of Technical Papers (Cat. No.02CH37315), vol. 1, Feb 2002, pp. 300–468 vol.1.
- [52] L. Li, P. Reynaert, and M. S. J. Steyaert, "Design and Analysis of a 90 nm mm-Wave Oscillator Using Inductive-Division LC Tank," *IEEE Journal* of Solid-State Circuits, vol. 44, no. 7, pp. 1950–1958, July 2009.
- [53] A. Mariano, O. Mazouffre, B. Leite, Y. Deval, J. B. Begueret, D. Belot, F. Rivet, and T. Taris, "An ultra low power consumption millimeter-wave voltage controlled oscillator in a 65 nm cmos-soi technology," *Analog Integrated Circuits and Signal Processing*, vol. 76, no. 3, pp. 277–286, Sep 2013. [Online]. Available: https://doi.org/10.1007/s10470-013-0040-z
- [54] T. Siriburanon, T. Ueno, K. Kimura, S. Kondo, W. Deng, K. Okada, and A. Matsuzawa, "A 60-GHz sub-sampling frequency synthesizer using subharmonic injection-locked quadrature oscillators," in 2014 IEEE Radio Frequency Integrated Circuits Symposium, June 2014, pp. 105–108.
- [55] Z. Zong, M. Babaie, and R. B. Staszewski, "A 60 GHz Frequency Generator Based on a 20 GHz Oscillator and an Implicit Multiplier," *IEEE Journal of Solid-State Circuits*, vol. 51, no. 5, pp. 1261–1273, May 2016.
- [56] C. Zhang and M. Otto, "A wide range 60 GHz VCO using back-gate controlled varactor in 22 nm FDSOI technology," in 2017 IEEE SOI-3D-Subthreshold Microelectronics Technology Unified Conference (S3S), Oct 2017, pp. 1–3.

- [57] Y. Peng, J. Yin, P. Mak, and R. P. Martins, "Low-Phase-Noise Wideband Mode-Switching Quad-Core-Coupled mm-wave VCO Using a Single-Center-Tapped Switched Inductor," *IEEE Journal of Solid-State Circuits*, pp. 1–11, 2018.
- [58] J. Yin and H. C. Luong, "A 57.5-90.1-GHz Magnetically Tuned Multimode CMOS VCO," *IEEE Journal of Solid-State Circuits*, vol. 48, no. 8, pp. 1851–1861, Aug 2013.
- [59] H. Jia, B. Chi, L. Kuang, and Z. Wang, "A 47.6-71.0-GHz 65-nm CMOS VCO Based on Magnetically Coupled π-Type LC Network," *IEEE Transactions on Microwave Theory and Techniques*, vol. 63, no. 5, pp. 1645– 1657, May 2015.
- [60] P. Agarwal, P. P. Pande, and D. Heo, "25.3 GHz, 4.1 mW VCO with 34.8% tuning range using a switched substrate-shield inductor," in 2015 IEEE MTT-S International Microwave Symposium, May 2015, pp. 1–4.
- [61] V. Szortyka, Q. Shi, K. Raczkowski, B. Parvais, M. Kuijk, and P. Wambacq, "A 42 mW 200 fs-Jitter 60 GHz Sub-Sampling PLL in 40 nm CMOS," *IEEE Journal of Solid-State Circuits*, vol. 50, no. 9, pp. 2025– 2036, Sept 2015.
- [62] T. Siriburanon, S. Kondo, M. Katsuragi, H. Liu, K. Kimura, W. Deng, K. Okada, and A. Matsuzawa, "A Low-Power Low-Noise mm-Wave Subsampling PLL Using Dual-Step-Mixing ILFD and Tail-Coupling Quadrature Injection-Locked Oscillator for IEEE 802.11ad," *IEEE Journal of Solid-State Circuits*, vol. 51, no. 5, pp. 1246–1260, May 2016.
- [63] A. I. Hussein, S. Vasadi, and J. Paramesh, "A 50-66-GHz Phase-Domain Digital Frequency Synthesizer With Low Phase Noise and Low Fractional Spurs," *IEEE Journal of Solid-State Circuits*, vol. 52, no. 12, pp. 3329– 3347, Dec 2017.
- [64] W. Wu, R. B. Staszewski, and J. R. Long, "A 56.4-to-63.4 GHz Multi-Rate All-Digital Fractional-N PLL for FMCW Radar Applications in 65 nm CMOS," *IEEE Journal of Solid-State Circuits*, vol. 49, no. 5, pp. 1081– 1096, May 2014.
- [65] B. Razavi, *RF Microelectronics*. Prentice Hall, 2012.
- [66] W. Egan, Frequency synthesis by phase lock. Wiley, 1981.
- [67] R. He, J. Li, W. Rhee, and Z. Wang, "Transient analysis of nonlinear settling behavior in charge-pump phase-locked loop design," in 2009 IEEE International Symposium on Circuits and Systems, May 2009, pp. 469–472.

- [68] S. Ek, T. Påhlsson, A. Carlsson, A. Axholt, A. K. Stenman, and H. Sjöland, "A 16-20 GHz LO system with 115 fs jitter for 24-30 GHz 5G in 28 nm FD-SOI CMOS," in *ESSCIRC 2017 - 43rd IEEE European Solid State Circuits Conference*, Sept 2017, pp. 251–254.
- [69] X. Yi, C. C. Boon, H. Liu, J. F. Lin, and W. M. Lim, "A 57.9-to-68.3 GHz 24.6 mW Frequency Synthesizer With In-Phase Injection-Coupled QVCO in 65 nm CMOS Technology," *IEEE Journal of Solid-State Circuits*, vol. 49, no. 2, pp. 347–359, Feb 2014.
- [70] C. H. Tsai, G. Mangraviti, Q. Shi, K. Khalaf, A. Bourdoux, and P. Wambacq, "A 54-64.8 GHz subharmonically injection-locked frequency synthesizer with transmitter EVM between -26.5 dB and -28.8 dB in 28 nm CMOS," in *ESSCIRC 2017 - 43rd IEEE European Solid State Circuits Conference*, Sept 2017, pp. 243–246.
- [71] Z. Huang, H. C. Luong, B. Chi, Z. Wang, and H. Jia, "A 70.5-to-85.5GHz 65nm phase-locked loop with passive scaling of loop filter," in 2015 IEEE International Solid-State Circuits Conference - (ISSCC) Digest of Technical Papers, Feb 2015, pp. 1–3.
- [72] H. Yoon, J. Kim, S. Park, Y. Lim, Y. Lee, J. Bang, K. Lim, and J. Choi, "A -31dBc integrated-phase-noise 29GHz fractional-N frequency synthesizer supporting multiple frequency bands for backward-compatible 5G using a frequency doubler and injection-locked frequency multipliers," in 2018 IEEE International Solid - State Circuits Conference - (ISSCC), Feb 2018, pp. 366-368.
- [73] J.-S. Lee, M.-S. Keel, S.-I. Lim, and S. Kim, "Charge pump with perfect current matching characteristics in phase-locked loops," *Electronics Letters*, vol. 36, no. 23, pp. 1907–1908, Nov 2000.
- [74] R. Y. Chen, "High-speed CMOS frequency divider," *Electronics Letters*, vol. 33, no. 22, pp. 1864–1865, Oct 1997.
- [75] H. R. Rategh and T. H. Lee, "Superharmonic injection-locked frequency dividers," *IEEE Journal of Solid-State Circuits*, vol. 34, no. 6, pp. 813–821, Jun 1999.
- [76] B. Razavi, "A study of injection locking and pulling in oscillators," *IEEE Journal of Solid-State Circuits*, vol. 39, no. 9, pp. 1415–1424, Sept 2004.
- [77] Y. Chen, M. Li, H. Kuo, T. Huang, and H. Chuang, "Low-Voltage K-Band Divide-by-3 Injection-Locked Frequency Divider With Floating-Source Differential Injector," *IEEE Transactions on Microwave Theory* and Techniques, vol. 60, no. 1, pp. 60–67, Jan 2012.

[78] M. Li, H. Kuo, T. Huang, and H. Chuang, "60GHz CMOS divide-by-5 injection-locked frequency divider with an open-stub-loaded floatingsource injector," in 2011 IEEE Radio Frequency Integrated Circuits Symposium, June 2011, pp. 1–4.

Paper I

Paper I

A two-stage mm-wave PA with 18.5% PAE in $65~\mathrm{nm}~\mathrm{CMOS}$

O2015 IEEE. Reprinted, with permission, from

Therese Forsberg, Henrik Sjöland and Markus Törmänen, "A two-stage mm-wave PA with 18.5% PAE in 65 nm CMOS," in *Proc. of IEEE Asia-Pacific Microwave Conference (APMC)*, Nanjing, China, Dec 2015, pp. 1-3.

A Two-stage mm-Wave PA with 18.5% PAE in 65 nm CMOS

Therese Forsberg, Henrik Sjöland, Markus Törmänen Department of Electrical and Information Technology Lund University Lund, Sweden

Abstract— A two-stage mm-wave power amplifier (PA) is presented. Designed in a 65 nm CMOS process, the PA employs capacitive neutralization in each stage for increased differential isolation and gain. Baluns are used for single-ended input/output signal to balanced signal conversion, and the interstage matching consists of a 2:1 transformer. With a 1.2 V supply, at 67 GHz, measurements show a gain of 16.8 dB, a 1dB-compression point (P1dB) of 8.4 dBm and a saturated output power (Psat) of 11.8dBm, with a peak power added efficiency (PAE) of 18.5 %. The PA core occupies an area of 100 um x 300 um.

Index Terms— PA, power amplifier, cross-coupling, neutralization, transformer-coupling, mm-wave, 60 GHz, CMOS, 65 nm, millimeter-wave.

I. INTRODUCTION

High data-rate communication in the unlicensed band around 60 GHz has lately attracted a lot of interest. The wide bandwidth of 7 GHz combined with a high atmospheric attenuation makes it suitable for high throughput short-range standards such as IEEE 802.15.3c for wireless personal area networks and IEEE 802.11ad for wireless local area networks.

To introduce products using such standards to the market, it is critical that the component cost is kept low. A key factor in making transceiver designs more economically attainable is the use of large scale complementary metal oxide semiconductor (CMOS) technology for implementation. Continuous scaling of CMOS technologies has improved their capability to provide gain at higher frequencies and made it a viable choice also for the design of power amplifiers. However, supply voltage limitations in deep-submicron CMOS technologies reduce the achievable PA output power. To overcome this problem a promising approach is to use several PAs for the amplification, and employ output power combining of the amplificatignals [1-3]. For each individual PA, a topology that provides sufficient gain, stability, linearity and efficiency is thus needed.

In this work, a two-stage PA implemented in 65 nm CMOS is presented. The objective has been to design a linear power amplifier that has a competitive efficiency and bandwidth. The PA is designed for 50 Ω single-ended input and output signals.

II. THEORY AND DESIGN

The schematic of the design can be seen in Fig. 1. Class A operation is used to ensure high linearity of the PA. The core of each amplifying stage is a differential pair employing

capacitive cross-coupling neutralization, which counteracts the significant effect of the gate-drain capacitance at higher frequencies, thus providing higher differential isolation and enabling higher gain without increasing the power consumption. The neutralization capacitances are implemented using off state transistor devices for improved matching. This alleviates process spread effects that might otherwise negatively impact the performance, and in severe cases even jeopardize stability. The interstage matching is realized with a transformer for a wide bandwidth operation. The transformer, seen in Fig 2, is designed to provide a 2:1 transformation.



Fig. 1. Schematic of the two-stage PA with capacitive neutralization.



Fig. 2. The interstage matching transformer.

Both the transformer and the input and output baluns were simulated in ADS Momentum and their dimensions were tuned to make the entire system operate at the targeted 60 GHz frequency band. All three transformers are realized in the top metal layers. Simulations of the transformer show a coupling coefficient of 0.7 and a loss of 2 dB. The simulated loss in the input and output balun is 1.1 dB and 2.1 dB, respectively.

The bias is applied to the gates through 30 k Ω resistors, rather than through the transformer center tap, in order to avoid potential common mode instability. Simulations show unconditional stability at all frequencies.

The dimensions of the transistors in the differential pair are 60μ m/65nm for the first stage, and 100μ m/65nm for the output stage. The theoretical optimal size for the cross-coupling capacitances is when it equals the gate-drain capacitance of the amplifying devices [4]. Since the capacitance is realized with off state transistors, providing overlap capacitance at both drain and source, the widths of these devices are chosen approximately half the width of the corresponding amplifying device.

III. MEASUREMENT RESULTS

The PA was implemented in the standard ST 65 nm processes with 8 metal layers. The die area of the PA, including the input and output matching, is 100 x 300 μ m². Including pads the total chip size is 0.4 x 0.5 mm². The chip photo is shown in Fig. 3.



Fig. 3. Chip photo.

The chips were mounted on printed circuit boards, to which supply and bias signals were wire bonded. Infinity GSG microprobes from Cascade Microtech were used for the input and output signals. The S-parameters were measured using an Agilent E8361A network analyzer. To be able to measure the input-output power characteristics above the maximum output power of the network analyzer, an extra amplifier HXI HLNAV-383 was inserted before the device under test (DUT). The output power was recorded using a V8486A power sensor. The same sensor was also used to calibrate the input power to the DUT.

The S-parameters measurements are presented in Fig. 4 and Fig. 5. As can be clearly seen, both the input and output matching is good. The gain peaks at 16.8 dB around 67 GHz, and the 3 dB bandwidth is estimated to 9 GHz.



Fig. 5. Measured gain and reverse isolation.



Fig. 7. Power added efficiency versus output power at 67 GHz.

TABLE I. PERFORMANCE SUMMARY AND COMPARISON OF RECENTLY PUBLISHED TWO- AND THREE-STAGE PAS.

Ref.	This work	[5]	[6]	[7]	[8]	
CMOS Technology	65 nm	45 nm	28 nm	65 nm	65 nm	
Supply Voltage	1.2	2	1	1.2	1	
Frequency (GHz)	67	60	53	55 ^a	60.5	
Gain (dB)	16.8	20	13	12.1	10	
P _{1dB} (dBm)	8.4	11.2	12	9.5	7.1	
Psat (dBm)	11.8	14.5	13	12.2	11.5	
Bandwidth, 3dB (GHz)	9	10 ^b	27	а	8.5	
PAE _{max} (%)	18.5	14.4	16	12.5	15.3	
PAE _{1dB} (%)	9	а	а	7.7	а	
Area (mm ²)	0.03	0.09	0.056	0.576	0.053	
a. Not clearly expressed in the source article. b.1dB bandwidth.						

a. Not clearly expressed in the source article. 0.103 bandwide

In Fig. 6, the output power versus input power at 67 GHz is shown. The maximum output power is 11.8 dBm and the 1dB compression point is 8.4 dBm. Using a 1.2 V supply, the power amplifier has a peak power added efficiency (PAE) of 18.5%. The PAE as a function of output power is shown in Fig. 7. From this graph, it can also be seen that the PAE at the 1dB compression point is 9%.

The measured performance of the two-stage PA is compared with other similar published two- and three-stage CMOS PAs in Table I.

IV. CONCLUSION

A two-stage class A CMOS PA in 65 nm CMOS is presented. It has a competitive efficiency, also at the compression point. The bandwidth is enough to cover the unlicensed band at 60 GHz and the circuit is using a very small active chip area.

ACKNOWLEDGMENT

The authors would like to thank STMicroelectronics for donating the IC fabrication, and the Swedish Research Council for funding this project.

REFERENCES

- J. Oh, B. Ku, S. Hong, "A 77-GHz CMOS Power Amplifier With a Parallel Power Combiner Based on Transmission-Line Transformer," *IEEE Transactions on Microwave Theory and Technique*, vol.61, no.7, pp.2662,2669, July 2013.
- [2] D. Zhao and P. Reynaert, "14.1 A 0.9V 20.9dBm 22.3%-PAE E-band power amplifier with broadband parallel-series power combiner in 40nm CMOS," 2014 IEEE International Solid-State Circuits Conference Digest of Technical Papers (ISSCC), pp.248,249, 9-13 Feb. 2014.
- [3] R. Bhat, A. Chakrabarti, H. Krishnaswamy, "Large-Scale Power Combining and Mixed-Signal Linearizing Architectures for Watt-Class mmWave CMOS Power Amplifiers," *IEEE Transactions on Microwave Theory and Technique*, vol.63, no.2, pp.703,718, Feb. 2015.
- [4] H. Asada, K. Matsushita, K. Bunsen, K. Okada, and A. Matsuzawa, "A 60GHz CMOS power amplifier using capacitive cross-coupling neutralization with 16 % PAE," 2011 European Microwave Integrated Circuits Conference (EuMIC), pp.554,557, 10-11 Oct. 2011.
- [5] M. Abbasi, T. Kjellberg, A. de Graauw, E. van der Heijden, R. Roovers, and H. Zirath, "A broadband differential cascode power amplifier in 45 nm CMOS for high-speed 60 GHz system-onchip," 2010 IEEE Radio Frequency Integrated Circuits Symposium (RFIC), pp.533,536, 23-25 May 2010.
- [6] J. Zhao, M. Bassi, A. Bevilacqua, A. Ghilioni, A. Mazzanti, F. Svelto, "A 40–67GHz power amplifier with 13dBm PSAT and 16% PAE in 28 nm CMOS LP," ESSCIRC 2014 - 40th European Solid State Circuits Conference (ESSCIRC), pp.179,182, 22-26 Sept. 2014.
- [7] R. Minami, K. Matsushita, H. Asada, K. Okada, A. Matsuzawa, "A 60 GHz CMOS power amplifier using varactor cross-coupling neutralization with adaptive bias," 2011 Asia-Pacific Microwave Conference Proceedings (APMC), pp.789,792, 5-8 Dec. 2011.
- [8] W. L. Chan and J. R. Long, "A 58–65 GHz Neutralized CMOS Power Amplifier With PAE Above 10% at 1-V Supply," *IEEE Journal of Solid-State Circuits*, vol.45, no.3, pp.554,564, March 2010.

Paper II

Paper II

A 65 nm CMOS varactorless mm-wave VCO

©2014 IEEE. Reprinted, with permission, from

Therese Forsberg, Henrik Sjöland and Markus Törmänen, "A 65 nm CMOS varactorless mm-wave VCO," in *Proc. of 2014 International Symposium on Integrated Circuits (ISIC)*, Singapore, Singapore, Dec 2014, pp. 54–57.

A 65 nm CMOS varactorless mm-wave VCO

Therese Forsberg, Henrik Sjöland, Markus Törmänen Department of Electrical and Information Technology Lund University Lund, Sweden

Abstract— We present a cross-coupled 60 GHz differential LC voltage controlled oscillator (VCO) implemented in 65 nm CMOS technology. To eliminate the need for varactors, the frequency tuning is implemented using transformers in the resonator, where the effective inductance can be controlled by injecting current into the secondary windings. Measurements show an average current consumption of 15 mA from a 1.2 V supply, resulting in a minimum FOM of -182.4 dBc/Hz, and the frequency versus tuning current in the secondary is close to linear. The die area of the oscillator core, including the transformers, is 70 um x 85 um.

Index Terms— VCO, mm-wave, 60 GHz, CMOS, 65 nm, millimeter-wave.

I. INTRODUCTION

As the frequency spectrum at lower frequencies becomes more and more crowded, wireless communication will have to use higher frequencies for expansion. The unlicensed band that resides between 57 and 66 GHz has 7 GHz of bandwidth, slightly differently placed depending on country [1]. The standards IEEE 802.15.3c for wireless personal area networks (WPAN) and IEEE 802.11ad for wireless local area networks use these frequencies [2],[3], and complete millimeter wave transceivers for these standards and others have already been presented [4]-[6]. It is not unreasonable to believe that just like the unlicensed 2.4 GHz band, the millimeter wave band will also soon be filled with several competing or complementing standards that create a need for cheap and energy efficient transceivers. To implement such transceivers in cheap and energy-effective large scale complementary metal oxide semiconductor (CMOS) technology is an attractive choice [1].

A key component in any transceiver is the voltage controlled oscillator (VCO). When designing VCOs for the 60 GHz band the common choice involves a cross-coupled pair with an LC tank. The frequency is then tuned by changing the capacitance in the tank using varactors or switched capacitor banks in different structures [7], [8]. A typical example of this kind of LC VCO is shown in Fig. 1. For this circuit the oscillation frequency is given by Eq. 1 below.

$$f_{osc} = \frac{1}{2\pi\sqrt{LC_{tot}}} \tag{1}$$

where C_{tot} includes both the varactor capacitance, the load capacitance, and the parasitic capacitance.



Fig. 1. Schematic of a typical cross-coupled VCO.

A high quality factor (Q) of the LC resonator is needed to achieve low phase noise and power consumption, but varactors have a low Q at millimeter wave frequencies [9], and besides that they are often associated with too much parasitic capacitance [7], [8]. Inductive tuning, implemented as a loaded transformer structure, is a way to get around this. Common solutions include switching on or off parts of the secondary coil. Low loss switches, however, are difficult to implement at millimeter wave frequencies. Usually the tank also includes a varactor [7], [10]-[13].

This paper proposes a method to tune the frequency without any varactor in the tank, or the need to switch inductances.

II. THEORY AND DESIGN

The core of the varactorless mm-wave VCO is a basic cross-coupled design, as can be seen in Fig. 2. The capacitive part of the tank of is constituted by the inherent capacitance in the transistors together with the other parasitic capacitances throughout the circuit. The inductive part of the tank is the primary, inner coil of a transformer. The frequency of the oscillator is tuned solely by changing the effective inductance of the tank, which is changed by controlling the current in the second and third branch. These branches inject current in phase and antiphase, respectively, in the secondary, outer coil of the transformer. The tuning mechanism can be understood as follows. For two magnetically coupled inductors, the voltage v_l of inductor 1 can be expressed as

$$v_1 = L_1 \frac{di_1}{dt} + L_m \frac{di_2}{dt} \tag{2}$$

where L_m is the mutual inductance. If i_2 is proportional to i_1 , i.e. $i_2 = xi_1$, then the effective inductance of the first inductor can be expressed as $(L + xL_m)$, since

$$v_{1} = L_{1} \frac{di_{1}}{dt} + L_{m} x \frac{di_{1}}{dt} = (L + xL_{m}) \frac{di_{1}}{dt}$$
(3)

Assuming that $L_1 = L_2$, then the effective inductance can be re-written as L(1+kx), since

$$L_m = k\sqrt{L_1L_2} = k\sqrt{L^2} = kL \to L + xL_m = L + xkL \quad (4)$$

where k is the coupling factor between 0 and 1.

The effective inductance can thus be controlled by changing *x*, which is accomplished by tuning v_{b1} , v_{b2} and v_{b3} . For instance, tuning *x* between 0 and 1 gives an effective inductance range of *L* to 1.4*L*, assuming k = 0.4, resulting in a frequency tuning range of 18%, as

$$\frac{f_{max}}{f_{min}} = \sqrt{1.4} = 1.18 \to 18\%$$
(5)

For this to work, however, the currents of the two inductors must be exactly in phase. Being 90 degrees out of phase would tune the resistance instead of the inductance. It is thus necessary to have the same resonance frequency in both tanks. Again, there is a contradiction between high Q and tunability. The frequency for which 90 degree shift occurs is, for a reasonable Q value of 20, separated from the center frequency by

$$\Delta f_{res} = \frac{f_0}{Q} = \frac{60GHz}{20} = 3GHz \tag{6}$$

That is, to obtain a wider tuning range it might be necessary to intentionally create a transformer with a lower Q value.

The purpose of the third branch in Fig. 2 is to decrease the effective inductance in branch 1, i.e. to generate negative values of x - contrary to the function of branch 2. When using this branch, the second branch should be turned off. This functionality, however, has not yet been verified.

The transformer was simulated in ADS Momentum and the dimensions were tuned to create a frequency span of 60 to 63 GHz together with the 22 μ m wide transistors in each g_m stage. The coupled inductors are placed in the top metal layer, M7. Simulations show that the unloaded inductance of the primary coil is 72 pH and that it is 95 pH for the secondary coil. Also, simulations show that the quality factor is 11 for both coils and that the coupling coefficient is 0.42.

The signals from the VCO core branch were routed to two open drain buffers, which provide the measurement instruments with up to 5 dBm of signal power.



Fig. 2. Schematic of the varactorless VCO.



Fig. 3. Chip photo.

2014 International Symposium on Integrated Circuits (ISIC)

III. MEASUREMENT RESULTS

The VCO was implemented in the standard ST 65 nm processes with 7 metal layers. The die area of the oscillator core, including the transformer, is 70 x 85 μ m². Including pads the total chip size is 0.4 x 0.7 mm². The chip photo is shown in Fig. 3.

The chips were mounted on printed circuit boards, to which supply and bias signals were wire bonded. The high frequency signals were measured using Cascade Microtech GSG microprobes. A Rohde&Schwarz FSU Spectrum Analyzer with a range extender was used to measure the single-ended signal from one of the buffers. The phase noise was measured using an Aeroflex PN9000 phase noise measurement system, where the measured signal was an amplified version of the IF signal from the spectrum analyzer. The frequency tuning of the VCO was verified by keeping $V_{b,3}$ at 0 V, V_{b1} at 0.8 V, and changing V_{b2} . The result, in which the voltage-to-current characteristics of the bias transistor can be clearly seen, is shown in Fig. 4.

A more complete picture of frequency as a function of i_l and i_2 is shown in Fig. 5. During the measurements the third branch was turned off. The data, consisting of 74 measurement points, show that the highest frequency of 65.083 GHz was attained for $i_l = 6.6$ mA and $i_2 = 0.001$ mA. The lowest frequency of 62.198 GHz was attained when $i_l = 12.9$ mA and $i_2 = 15.9$ mA, yielding a tuning range of 2.88 GHz or 4.5%. It was also found that the entire tuning range can be shifted by more than 1 GHz by changing the biasing of the buffers, and thus effectively change the capacitive load of the VCO the same way that a varactor would do.

Phase noise measurements and the corresponding FOM at 10 MHz offset for each frequency are presented in Fig. 6. In the phase noise measurement, both supply voltages, as well as V_{b1} , are set to 1.2 V, and the frequencies are swept by changing V_{c2} from 0.2 to 1.2 V in three steps. No corrections have been made in the phase noise data to account for added noise from the down conversion and 30 dB amplification of the signal. The best FOM of -182.4 dBc/Hz was found at the highest frequency, where the measured phase noise is low while the dissipated power is also at its minimum.

The measured performance of the VCO is compared with other recently published CMOS VCOs in Table I. As can be seen, this work has excellent phase noise and a small area. Although the tuning range may be on the low side, it may still be enough for the purpose, as many standards will eventually coexist in the millimeter wave band.

IV. CONCLUSION

A new varactorless VCO architecture is proposed in this paper. In contrast to previously published VCOs using inductive tuning methods, the tuning is carried out by continuously changing the effective inductance seen by the cross coupled pair. The measurement results show that the lack of a varactor leads to small size and a good phase noise performance.



Fig. 4. Simulated and measured frequency and current consumption vs. vb2.







Fig. 6. Phase noise at 1, 10 and 40 MHz offset and FOM vs. frequency.

Ref.	This work	[7]	[10]	[11]	[12]	[13]
Technology (CMOS)	65 nm	90 nm	65 nm	65 nm	90 nm	65 nm
Supply Voltage	1.2	0.7	1.2	1	0.43	1
Frequency (GHz)	63.7	56.75	73.8	61	61.7	59
Tuning Range (%)	4.5	16.07	41.1	14.2	4.81	5.4
Power consumption (mW)	6 to 30	8.7	8.4 to 10.8	6	1.2	16.5 ^e
PN@10MHz (dBc/Hz)	-114.9 to -118.6	-118.8	-104.6 to -112.2	-110.8	-110 ^b	-112 ^d
FOM ^a (dBc/Hz)	-177.4 to -182.4	-184.3	-172 to -180	-179.3	-185°	-173.6
Area (mm ²)	0.006	0.1	0.03	0.031	0.007	0.21 ^f

TABLE I. PERFORMANCE SUMMARY AND COMPARISON OF RECENTLY PUBLISHED TRANSFORMER BASED VCOS

a. The figure of merit, FOM=(PN)-20 log(fosc/Δf)+10log(P diss/1mW)

b. Calculated as PN at 1 MHz -20 dB, c. Calculated from 1 MHz phase noise measurements. d. Minimum measured figure reported. e. Including buffers. f. Including pads

ACKNOWLEDGMENT

The authors would like to thank STMicroelectronics for donating the IC fabrication and to Swedish Research Council for funding this project.

REFERENCES

- [1] T. Baykas, C. Sum, Z. Lan, J. Wang, M. Rahman, H. Harada, and S. Kato, "lece 802.15. 3c: the first iece wireless standard for data rates over 1 gb/s," IEEE Communications Magazine, vol. 49, no. 7, pp. 114–121, 2011.
- [2] Wireless Medium Access Control (MAC) and Physical Layer (PHY) Specifications for High Rate Wireless Personal Area Networks (WPANs), IEEE Standard 802.15.3c, 2009.
- [3] Wireless LAN Medium Access Control (MAC) and Physical Layer (PHY) Specifications, IEEE Standard P802.11ad/D2.0, 2011.

- [4] K. Okada, K. Matsushita, K. Bunsen, R. Murakami, A. Musa, T. Sato, H. Asada, N. Takayama, N. Li, S. Ito, W. Chaivipas, R. Minami, and A. Matsuzawa, "A 60-GHz 6QAM/8PSK/QPSK/BPSK direct-conversion transceiver for IEEER02.15.3c," in Proc. IEEE Int. Solid-State Circuit Conf., Feb. 2011, pp. 160–161.
- [5] A. Siligaris, O. Richard, B. Martineau, C. Mounet, F. Chaix, R. Ferragut, C. Dehos, J. Lanteri, L. Dussopt, S. D. Yamamoto, R. Pilard, P. Busson, A. Cathelin, D. Belot and P. Vincent, "A 65nm CMOS Fully Integrated Transceiver Module for 60GHz Wireless HD Applications," in IEEE Int. Solid-State Circuits Conf. Dig. Tech. Papers, pp. 162-163, Feb. 2011.
- [6] J. Sato, T. Shima, M. Iwamoto, T. Akizuki, K. Mizuno, "A 60 GHz fully integrated CMOS transceiver with amplitude/phase imbalance cancellation technique," IEEE Int. Symp. on RFIT Dig., pp. 101-104, Nov. 2011.
- [7] T.-Y. Lu, C.-Y. Yu, W.-Z. Chen, and C.-Y. Wu, "Wide tunning range 60 GHz VCO and 40 GHz DCO using single variable inductor," IEEE Trans. Circuits Syst. I: Regular Papers, vol. 60, pp. 257–267, Feb. 2013.
- [8] M. Demirkan, S. P. Bruss, and R. R. Spencer, "Design of wide tuning range CMOS VCOs using switched coupled-inductors," IEEE J. Solid- State Circuits, vol. 43, pp. 1156–1163, May 2008.
- [9] Y. Takigawa, H. Ohta, L.Qing, S. Kurachi, N. Itoh, and T. Yoshimasu, "A 92.6% tuning range VCO utilizing simultaneously controlling of transformers and MOS varactors in 0.13 m CMOS technology," in IEEE RFIC Symp., 2009, pp. 83–86
- [10] J. Yin and H. C. Luong, "A 57.5-to-90.1 GHz magneticallytuned multi-mode CMOS VCO," in Proc. CICC, 2012, pp. 1–4.
- [11] W. Fei, H. Yu, H. Fu, J. Ren and K. Seng Yeo, "Design and Analysis of Wide Frequency-Tuning-Range CMOS 60 GHz VCO by Switching Inductor Loaded Transformer," IEEE Trans. Circuits Syst. I: Regular Papers, vol. 61, no. 3, pp. 699–711, Mar. 2014.
- [12] L. Li, P. Reynaert, and M. Steyaert, "Design and analysis of a 90 nm mm-wave oscillator using inductive-division LC tank," IEEE J. Solid-State Circuits, vol. 44, pp. 1950–1958, July 2009.
- [13] M. Kraemer, D. Dragomirescu, R. Plana, "A high efficiency differential 60 GHz VCO in CMOS technology for WSN applications", IEEE Microw. Wireless Compon. Lett., vol. 21, no. 6, pp. 314–316, Jun. 2011.

Paper III

Paper III

Two mm-Wave VCOs in 28-nm UTBB FD-SOI CMOS

O2017 IEEE. Reprinted, with permission, from

Therese Forsberg, Johan Wernehag, Anders Nejdel, Henrik Sjöland and Markus Törmänen, "Two mm-Wave VCOs in 28-nm UTBB FD-SOI CMOS," *IEEE Microwave and Wireless Components Letters*, vol. 27, no. 5, pp. 509–511, May 2017

Two mm-Wave VCOs in 28-nm UTBB FD-SOI CMOS

Therese Forsberg, Student Member, IEEE, Johan Wernehag, Member, IEEE, Anders Nejdel, Member, IEEE, Henrik Sjöland, Senior Member, IEEE, and Markus Törmänen, Senior Member, IEEE

Abstract—Two 60-GHz band voltage controlled oscillators (VCOs) designed in a 28-nm ultrathin body and buried oxide fully depleted silicon on insulator (UTBB FD-SOI) CMOS process are demonstrated and compared. Both VCOs have identical cross-coupled nMOS cores and dissipate 3.15 mW from a 0.9-V supply. The first design uses a standard FET current source and achieves a figure of merit (FOM) of -181 dBc/Hz, whereas the second employs a filtered current source and achieves 4 demonstrates the efficiency of the filtering technique at millimeter wave frequencies and the feasibility of efficient low-phase noise designs in 28-nm UTBB FD-SOI CMOS. The active area of the filtered VCO is 90 μ m × 180 μ m and the standard VCO has an area of 80 μ m × 110 μ m.

Index Terms—Analog integrated circuits, CMOS integrated circuits, low-phase noise, millimeter wave (mm-wave) integrated circuit, ultrathin body and buried oxide fully depleted silicon on insulator (UTBB FD-SOI) CMOS, voltage controlled oscillator (VCO).

I. INTRODUCTION

TNCREASING data rates demand increased bandwidth, such as the 7 GHz available in the unlicensed 60-GHz band, but also employment of more complex modulation schemes. These put stringent requirements on the phase-noise performance of the frequency generation circuitry.

Digital designs are moving to the smaller process nodes, and the analog designs must follow. Current state-of-theart millimeter wave (mm-wave) CMOS VCOs [1]–[4] will thus be replaced by counterparts in process nodes, such as the 28-nm ultrathin body and buried oxide fully depleted silicon on insulator (UTBB FD-SOI). Indeed, this process also provides analog benefits, such as higher f_t and f_{max} , higher intrinsic gain, and less parasitics. However, the high flicker noise of modern downscaled technologies requires efficient mitigation methods, such as employing a filter to resonate the differential pair source node at $2f_0$, and also to prevent tail current noise at $2f_0$ from creating phase noise [5].

Manuscript received November 8, 2016; accepted January 31, 2017. Date of publication April 19, 2017; date of current version May 8, 2017. This work was supported in part by the Swedish Research Council and in part by STMicroelectronics (for donating chip fabrication). (*Corresponding Author: Therese Forsberg.*)

T. Forsberg, J. Wernehag, H. Sjöland, and M. Törmänen are with the Department of Electrical and Information Technology, Lund University, 221 00 Lund, Sweden (e-mail: therese.forsberg@eit.lth.se; johan.wernehag@eit.lth.se; henrik.sjoland@eit.lth.se; markus.tormanen@eit.lth.se).

A. Nejdel was with the Department of Electrical and Information Technology, Lund University, 221 00 Lund, Sweden. He is now with Mellanox Technologies, 4000 Roskilde, Denmark (e-mail: andersn@mellanox.com).

Color versions of one or more of the figures in this paper are available online at http://ieeexplore.ieee.org. Digital Object Identifier 10.1109/LMWC.2017.2690845 Although this technique has already been applied at many different frequencies, including the mm-wave bands [6]–[10], it has not clearly been investigated what improvement in phase-noise performance that can be expected from it in CMOS VCOs at mm-wave frequencies.

In this letter, two 60-GHz band VCOs have been designed, fabricated, and characterized through measurements. Both VCOs have identical active cores, and one of them also uses the filtering technique of interest for improved phase-noise performance. The aim is to investigate the efficiency of the filtering technique at mm-wave frequencies and to create a high-performance VCO for the 60-GHz band.

II. THEORY AND DESIGN

The differential cross-coupled *LC* oscillator topology is chosen for both designs as it is well known for its good phase-noise performance and ease of implementation [11].

The schematics of the implemented VCOs are shown in Fig. 1. Both VCO cores are designed to be as similar to each other as possible to facilitate a fair comparison. In the standard cross-coupled nMOS VCO, an FET current source is used. Phase-noise performance is optimized when the resonance tank is on the limit of being current limited. In this configuration, the design is susceptible to the noise generated in the current source transistor. To improve the flicker noise performance, its width is made three times the minimum channel length of the process.

Ideally, the current source should also have high impedance at $2 f_0$ to prevent the triode resistance of the cross-coupled pairs from loading the resonator in the switched state. The second implemented VCO design, shown to the right in Fig. 1, thus employs a filter at the FET current source transistor [5]. A source inductor is used to resonate the parasitics of the source node at $2 f_0$. Simulations show that while the tank waveform frequency contents of both VCO designs are almost identical, the second harmonic in the source node of the crosscoupled pair is three times stronger in the filtered design. A capacitor in parallel with the FET current source shunts high frequency noise from the current source to ground. However, low-frequency tail current noise can still cause phase noise due to the varactor nonlinearities [12]. In all, simulations show that for these designs, the maximum expected improvement in phase-noise performance at 10-MHz offset is 6.6 dB.

The varactors in both the designed VCOs consist of two transistors, dimensioned to give a desired tuning range of about 10%. The minimum channel length is used in order to maximize the varactor Q. The cross-coupled pairs are sized to provide enough loop gain to fulfill the oscillation criteria.

1531-1309 © 2017 IEEE. Personal use is permitted, but republication/redistribution requires IEEE permission. See http://www.ieee.org/publications_standards/publications/rights/index.html for more information.



Fig. 1. Middle: chip microphotograph. Schematics of the cross-coupled nMOS VCOs. Left: standard design. Right: design with a filtered current source.



Fig. 2. Simulated phase noise at 10-MHz offset versus source filter inductor L and Q, at a varactor voltage of 0 V.

All inductors were designed in the top metal layers and simulated in ADS Momentum. The tank inductance dimensions were tuned to make the two oscillators operate at the targeted 60-GHz frequency band.

Since the layout of the two VCO designs needs to be different in order to accommodate for the source node filtering in the second design, the parasitic capacitances and inductive coupling associated with the VCO tanks are slightly different. Thus, this also results in the inductance values having to differ slightly to ensure operation in the targeted frequency range. Simulations of the tank inductors show that the inductance values are 112 and 125 pH, each with an associated Q of 22.

The filter inductance value is 95 pH, with a Q of 18. Simulations show that the resonance frequency of the source node is important. As can be seen in Fig. 2, there is an optimal value of the inductance. Because of the steeper degradation on the lower side of the optimum, a slightly higher inductance value was chosen for the design to make it more robust to variations. The difference between the phase noise performance of the optimal inductance and the actual inductance is approximately 1 dB.

As for the impact of the Q of the filter inductance, simulations show that it is of minor importance to the filter noise reduction efficiency, as can be seen in Fig. 2.

For measurement purposes, the signals from the VCOs are routed to open drain buffers with approximately 5 dBm of output power.



Fig. 3. Measured frequency tuning for both VCO designs.

III. MEASUREMENT RESULTS

The two VCOs to be benchmarked were implemented in a 28-nm UTBB FD-SOI CMOS process with ten metal layers for interconnect from STMicroelectronics. The chip microphotograph is shown in Fig. 1. The chips were mounted on printed circuit boards, to which supply and bias signals were wire bonded. An Infinity GSG microprobe from Cascade Microtech was used for on-chip probing of the VCO buffer output signal. A Rohde & Schwarz FSU50 spectrum analyzer with a *V*-band frequency range extender, i.e., a subharmonic mixer, was used to measure the single-ended VCO buffer output signal. The phase noise was measured using an Aeroflex EuropTest PN9000 phase noise measurement system, and the measured signal was an amplified version of the downconverted VCO buffer output signal, at an intermediate frequency of 404.4 MHz, available from the spectrum analyzer.

Measurement results were obtained for both VCOs operating from a 0.9-V supply, each with a 3.5-mA core current, i.e., dissipating 3.15-mW dcpower.

The frequency tuning characteristics of the VCOs is presented in Fig. 3. The phase noise measurements along with the figure of merit (FOM) at 10-MHz offset are presented in Fig. 4, where the phase-noise improvement given by the filtering is clearly visible. At 1-MHz offset, the improvement in phase noise is measured to be between 5 and 12 dB. The best performance for both VCOs is at a varactor voltage of 0.2 V,

Ref.	Technology	Supply	Frequency	Tuning	Power	PN@10MHz	PN@1MHz	FOM ^a	Area
	(CMOS)	Voltage (V)	(GHz)	Range (%)	consumption (mW)	(dBc/Hz)	(dBc/Hz)	@10 MHz or 1MHz (dBc/Hz)	(mm2)
This work ^b	28nm FD-SOI	0.9	67.3	10.9	3.15	-109.9	-84.5	-181	0.088
This work ^c	28nm FD-SOI	0.9	64	11.1	3.15	-116.5	-92.5	-187.3	0.162
[1]	90nm bulk	1.2	61.3	8.44	7.2	-	-90.9	-178.1	0.34 ^d
[2]	65nm bulk	1	61	14.2	6	-108.3	-	-176.2	0.031
[3]	130 nm	1.2	57.7	16.8	7.6	-124.8	-100.6	-186.3	0.201 ^d
[6]	40 nm bulk	0.9	58	16	14	-	-91 to -94.5	-175 to -177	-
[7]	65 nm bulk	1	59	5.4	16.5 (incl. buffers)	-	-90.3	-167.3	-
[8]	90 nm bulk	1.2	83	0.8	13	-109.86	-83.21	-177	0.776^{d}
[9]	65 nm bulk	1.2	70	6.5	7.2	-113.2	-	-181.8	0.048
[10]	65 nm bulk	1.2	56	17	15	-	-99.4	-182.2	-

TABLE I Performance Summary and Comparison of Recently Published 60-GHz CMOS VCOs

a. The figure of merit, FOM=(PN)-20 log(fosc/\Deltaf)+10log(P_diss/1mW). b. Standard VCO design. c. Filtered current source VCO design. d. Pads included.



Fig. 4. Measured phase noise at 1- and 10-MHz carrier offset frequencies for both VCOs (top) and VCO FOM versus varactor control voltage (bottom).

where the phase noise at 10-MHz offset is improved by 6.6 dB and the filtered design achieves an FOM of -187.3 dBc/Hz.

The measured performance of the VCOs is compared with each other and to other recent published state-of-the-art 60-GHz band CMOS VCOs in Table I. The table also includes other mm-wave VCOs, which use the same filtering technique. Among the compared designs, both designs demonstrated here have the lowest reported power consumption. The filtered current source VCO design achieves an excellent phase-noise performance, and to the best of our knowledge, the highest reported FOM for a 60-GHz band CMOS VCO.

IV. CONCLUSION

This letter presents two measured cross-coupled nMOS VCO designs for the 60-GHz band in 28-nm UTBB FD-SOI CMOS process technology. While otherwise as identical as possible, one of design employs filtering of the current source. A comparison between the measured results empirically show that the improvement gained from the current source filtering technique is between 5 and 12 dB. The filtered VCO design achieves an excellent phase-noise performance and a state-of-the-art FOM of -187 dBc/Hz. It also has a good tuning range of 11 % and low-power consumption.

REFERENCES

- Y.-C. Chiang and Y.-H. Chang, "A 60 GHz CMOS VCO using a fourthorder resonator," *IEEE Microw. Wireless Compon. Lett.*, vol. 25, no. 9, pp. 609–611, Sep. 2015.
- [2] W. Fei, H. Yu, H. Fu, J. Ren, and K. S. Yeo, "Design and analysis of wide frequency-tuning-range CMOS 60 GHz VCO by switching inductor loaded transformer," *IEEE Trans. Circuits Syst. I, Reg. Papers*, vol. 61, no. 3, pp. 699–711, Mar. 2014.
- [3] A. H. M. Shirazi, A. Nikpaik, R. Molavi, S. Mirabbasi, and S. Shekhar, "A Class-C self-mixing-VCO architecture with high tuning-range and low phase-noise for mm-wave applications," in *Proc. IEEE Int. Symp. Radio-Freq. Integr. Technol.*, May 2015, pp. 107–110.
- [4] Y. Chen, Y. Pei, and D. M. W. Leenaerts, "A dual-band LO generation system using a 40 GHz VCO with a phase noise of -106.8 dBc/Hz at 1-MHz," in *Proc. IEEE Radio Freq. Integr. Circuits Symp.*, Jun. 2013, pp. 203–206.
- [5] E. Hegazi, H. Sjöland, and A. A. Abidi, "A filtering technique to lower LC oscillator phase noise," *IEEE J. Solid-State Circuits*, vol. 36, no. 12, pp. 1921–1930, Dec. 2001.
- [6] V. Szortyka, Q. Shi, K. Raczkowski, B. Parvais, M. Kuijk, and P. Wambacq, "A 42 mW 200 fs-jitter 60 GHz sub-sampling PLL in 40 nm CMOS," *IEEE J. Solid-State Circuits*, vol. 50, no. 9, pp. 2025–2036, Sep. 2015.
- [7] M. Kraemer, D. Dragomirescu, and R. Plana, "A high efficiency differential 60 GHz VCO in a 65 nm CMOS technology for WSN applications," *IEEE Microw. Wireless Compon. Lett.*, vol. 21, no. 6, pp. 314–316, Jun. 2011.
- [8] W. Chaivipas, K. Okada, and A. Matsuzawa, "A 80 GHz voltage controlled oscillator utilizing a negative varactor in 90 nm CMOS technology," in *Proc. IEEE Asian Solid-State Circuits Conf.*, Nov. 2008, pp. 133–136.
- [9] C. Bryant, J. Lindstrand, H. Sjöland, and M. Törmänen, "A 70 and 210 GHz LO generator in 65 nm CMOS," in *Proc. IEEE Int. Symp. Radio-Freq. Integr. Technol.*, Nov. 2012, pp. 195–197.
- [10] J. L. Gonzalez, F. Badets, B. Martineau, and D. Belot, "A 56-GHz LC-tank VCO with 17% tuning range in 65-nm bulk CMOS for wireless HDMI," *IEEE Trans. Microw. Theory Techn.*, vol. 58, no. 5, pp. 1359–1366, May 2010.
- [11] A. Hajimiri and T. H. Lee, "Design issues in CMOS differential LC oscillators," *IEEE J. Solid-State Circuits*, vol. 34, no. 5, pp. 717–724, May 1999.
- [12] C. Samori, A. L. Lacaita, A. Zanchi, S. Levantino, and F. Torrisi, "Impact of indirect stability on phase noise performance of fully-integrated LC tuned VCOs," in *Proc. IEEE Eur. Solid-State Circuits Conf.*, Sep. 1999, pp. 202–205.

Paper IV

Paper IV

Two Ultra-Low Power mm-Wave Push-Pull VCOs in FD-SOI CMOS

O2018 IEEE. Reprinted, with permission, from

Therese Forsberg, Johan Wernehag, Henrik Sjöland and Markus Törmänen, "Two Ultra-Low Power mm-Wave Push-Pull VCOs in FD-SOI CMOS," in *Proc. of 2018 IEEE Asia-Pacific Microwave Conference (APMC)*, Kyoto, Japan, Nov 2018, 3 pages. Accepted.
Two Ultra-Low Power mm-Wave Push-Pull VCOs in FD-SOI CMOS

Therese Forsberg #1, Johan Wernehag #2, Henrik Sjöland #3, Markus Törmänen #4

*Department of Electrical and Information Technology, Lund University Box 117, 221 00 Lund, Sweden ¹therese.forsberg@eit.lth.se ²johan.wernehag@eit.lth.se ³henrik.sjoland@eit.lth.se ⁴markus.tormanen@eit.lth.se

Abstract — Two low-power mm-wave voltage controlled oscillators (VCOs) designed in a 28-nm fully-depleted silicon-oninsulator (FD-SOI) CMOS process are demonstrated, using a push-pull differential architecture. Measurement show that the first VCO has a 10.8% tuning range at 58.7 GHz. The 30 GHz VCO consumes 1.06 mW from a 1 V supply, and has a -119 dBc/Hz phase noise at 10 MHz offset, achieving a figure of merit of -188.4 dB. The 60 GHz VCO consumes 1.35 mW and has a -111.9 dBc/Hz phase noise at 10 MHz offset, achieving a figure of merit of -186.2 dB. Their active areas are 0.03 mm² and 0.096 mm², respectively.

Index Terms — CMOS analog integrated circuits, mm-wave circuits, millimeter wave oscillators, voltage controlled oscillators, phase noise

I. INTRODUCTION

The ever-increasing demand for higher data rates in wireless communication requires wider bandwidths. As the frequency spectrum at lower frequencies becomes increasingly crowded, the available required bandwidths can instead be found at mm-wave frequencies. In the 7 GHz wide unlicensed band at 60 GHz, several standards, such as the IEEE 802.11ad for wireless local area networks, are already in place [1]. Moreover, several parts of the spectrum around 30 GHz will be used by the 5G standard. Since such communication standards target battery-operated consumer electronics, it means that it is imperative that the transceiver designs are low-power and low-cost.

It is desirable to use CMOS processes also for the analog parts of the transceiver to achieve low cost and high integration level. However, that may lead to higher power consumption in some parts of the transceiver. This is especially true for the frequency generation circuits in modern downscaled CMOS technologies due to higher flicker noise. The power consumption of the voltage-controlled oscillator (VCO) may thus need to be increased in order to obtain the necessary noise performance required to employ the complex modulation schemes that are a prerequisite for high data rates.

In this work, it has been investigated if ultra-low power mm-wave VCOs can be successfully implemented in 28-nm FD-SOI CMOS technology, especially in the unlicensed 60 GHz band and around 30 GHz.

II. THEORY AND DESIGN

An attractive VCO topology for low-power operation, that is often used in different variations at RF frequencies [2]-[4], is shown in Fig. 1. To increase the attainable transconductance without increasing the current consumption, the basic differential cross-coupled NMOS LC oscillator topology is extended with a PMOS cross-coupled pair, in a push-pull configuration. This means that the current is effectively reused. Compared to the NMOS-only cross-coupled topology the output voltage swing will be lower, and limited to between ground and VDD. However, this also ensures safe operation of the transistors over time.

This current-reuse topology was chosen for the two VCO designs, one operating at around 30 GHz and the other around 60 GHz. To fit in a high performance mm-wave transmitter, a supply voltage of 1 V was used in the VCOs, thus avoiding the need of a separate low supply voltage domain. The drawn transistor dimensions that are equal in the two designs are indicated in Fig. 1.



Fig. 1. Schematic of the push-pull VCOs.

Although the FET current source in the bottom both adds to the noise and decreases the voltage headroom available for the cross-coupled pairs, it is important to counteract amplitude variations due to process variations. Also, it facilitates the possibility to turn off the VCO without turning off the lowdropout regulator supplying the VCO. Its length was made three times the minimum channel length of the process to improve the flicker noise performance.

In the VCO tank, the varactors in both of the designed VCOs consist of two transistors, dimensioned to give a desired tuning range of about 10 %. To maximize the varactor Q, the minimum channel length was used. To tune the VCOs to the targeted frequency bands, the varactor of the 30 GHz VCO was designed to be twice the size of the varactor in the 60 GHz VCO. This also preserves the tuning range. Since the oscillation frequency is inversely proportional to the square root of inductance times capacitance, the inductance was also scaled up in the 30 GHz VCO. However, the ratio was slightly larger than 2 because the 30 GHz VCO parasitic capacitances are not scaled up. The inductors were designed in the top metal layer and simulated in ADS Momentum. The simulated inductance value for the 30 GHz VCO was 207 pH and 73 pH for the 60 GHz VCO.

The signals from the VCOs are routed to open drain buffers with approximately 5 dBm of output power for measurement purposes.

III. MEASUREMENT RESULTS

VCOs implemented The two were in the STMicroelectronics 28-nm ultra-thin body and buried oxide fully-depleted silicon-on-insulator (UTBB FD-SOI) CMOS technology with 10 metal layers for interconnect. The chip microphotograph is shown in Fig. 2. The chips were mounted on printed circuit boards, from which the bias and supply voltages were provided through bond wires. The mm-wave single-ended output signals from the VCO output buffers were probed on-chip using an Infinity GSG microprobe from Cascade Microtech, and measured using a Rohde&Schwarz FSU50 spectrum analyzer equipped with an external V-band mixer. A down-converted version of the signal, available from the spectrum analyzer around the internal intermediate frequency of 404.4 MHz, was then amplified and sent to an Aeroflex EuropTest PN9000 phase noise measurement system.

Measurement results were obtained for both VCOs operating from a 1 V supply. The power consumption was 1.06 mW for the 30 GHz VCO, and 1.35 mW for the 60 GHz. The frequency tuning characteristics of the VCOs are shown in Fig. 3. The phase noise measurements for both VCOs, along with the figure of merit (FOM) at 10 MHz offset, are presented in Fig. 4 and Fig. 5, respectively. The best performance for both VCOs was at a varactor voltage of 0.1 V, where the 30 GHz VCO had a phase noise of -119 dBc/Hz and the 60 GHz VCO -111.9 dBc/Hz, both at 10 MHz offset from the carrier. The measured phase noise versus frequency offset

for both VCOs at this varactor voltage is shown in Fig. 6, where the high flicker noise corner is visible.

The measured performance is compared to other recently published state-of-the-art low-power mm-wave CMOS VCOs in Table I. The designs demonstrated in this work achieve state-of-the-art performance. The 30 GHz VCO has the lowest power consumption of all compared designs. Both have about 10 % tuning range and work from a 1 V supply voltage.



Fig. 2. Chip microphotograph.



Fig. 3. Measured frequency tuning characteristics of the two VCOs.



Fig. 4. Measured phase noise at 10 MHz offset from the carrier, for both VCOs.

PERFORMANCE SUMMARY AND COMPARISON OF RECENTLY PUBLISHED LOW-POWER MM-WAVE CMOS VCOS									
Ref.	Tech.	Supply	Frequency	Tuning Range	Power	PN @10MHz	FOM ¹	Area	
	(CMOS)	Voltage	(GHz)	(%)	Consumption	(dBc/Hz)	@10 MHz	(mm²)	
		(V)			(mW)		(dBc/Hz)		
This work	28 nm	1	30.3	10.8	1.06	-119	-188.4	0.096	
This work	28 nm	1	58.7	8.9	1.35	-111.9	-186.2	0.03	
[6]	90 nm	0.7	58.4	9.32	8.1	-91 (@1MHz)	-177.2 (@1MHz)	0.008	
[၁]	90 nm	0.43	61.7	4.81	1.2	-90 (@1MHz)	-185 (@1MHz)	0.008	
[6]	32 nm	0.6	30.4	7.6	4.56	-128	-190.9	0.3	
[7]	65 nm	1	63.7	9.7	2	-110.86	-184.07	0.047	
[/]	65 nm	0.8	62.9	8.6	1.1	-108	-183.56	0.047	
[8]	40 nm	0.7	55.5	25.4	10.5 ²	-122.3 ²	-183.7 ²	0.0672 ²	

TABLE I Performance Summary and Comparison of Recently Published Low-Power MM-Wave CMOS VCOs

The figure of merit, FOM=PN-20 log(f_{osc}/Δf)+10log(P_{diss}/1mW).
 Including only the first buffer/amplifier stage.



Fig. 5. Measured FOM at 10 MHz offset from the carrier, for both VCOs.



Fig. 6. Measured phase noise of the VCOs, at $V_{control} = 0.1$ V.

IV. CONCLUSION

This paper presents two measured mm-wave push-pull VCOs, one operating at 30 GHz and the other one at 60 GHz, implemented in 28-nm UTBB FD-SOI CMOS technology. Both have a good frequency tuning range of about 10%, and FOMs comparable to other recently published low-power mm-wave CMOS VCOs. They both have a very low power consumption, with just over 1 mW each. This is achieved with a 1 V supply, avoiding the need for a dedicated low supply voltage domain for the VCO. To conclude, the results demonstrate the feasibility of low-power VCOs at mm-wave frequencies in down-scaled CMOS technologies.

ACKNOWLEDGEMENT

The authors would like to thank STMicroelectronics for donating the IC fabrication, and the Swedish Research Council for funding this project.

REFERENCES

- [1] "IEEE Standard for Information technology– Telecommunications and information exchange between systems Local and metropolitan area networks–Specific requirements - Part 11: Wireless LAN Medium Access Control (MAC) and Physical Layer (PHY) Specifications," *IEEE Std* 802.11-2016 (*Revision of IEEE Std 802.11-2012*), pp. 1–3534, Dec 2016.
- [2] M. Tiebout, "Low-power low-phase-noise differentially tuned quadrature VCO design in standard CMOS," *IEEE Journal of Solid-State Circuits*, vol. 36, no. 7, pp. 1018-1024, July 2001.
- [3] L. Fanori and P. Andreani, "A high-swing complementary class-C VCO," 2013 Proceedings of the ESSCIRC (ESSCIRC), Bucharest, 2013, pp. 407-410.
- [4] I. Ghorbel, F. Haddad and W. Rahajandraibe, "Ultra low power RF cross-coupled VCO design in the subthreshold regime with high immunity to PVT variations in 130nm CMOS technology," 2015 IEEE 13th International New Circuits and Systems Conference (NEWCAS), Grenoble, 2015, pp. 1-4.
- [5] L. Li, P. Reynaert and M. S. J. Steyaert, "Design and Analysis of a 90 nm mm-Wave Oscillator Using Inductive-Division LC Tank," *IEEE Journal of Solid-State Circuits*, vol. 44, no. 7, pp. 1950-1958, July 2009.
- [6] D. Shin, S. Raman and K. J. Koh, "A 0.6-V, 30-GHz six-phase VCO with superharmonic coupling in 32-nm SOI CMOS technology," 2015 IEEE Custom Integrated Circuits Conference (CICC), San Jose, CA, 2015, pp. 1-4.
- [7] A. Mariano, O. Mazouffre, B. Leite Y. Deval, J. B. Begueret, D. Belot, F. Rivet, T. Taris, "An ultra low power consumption millimeter-wave voltage controlled oscillator in a 65 nm CMOS-SOI technology," *Springer Analog Integrated Circuits and Signal Processing*, vol. 76, no. 7, pp. 277-286, February 2013.
- [8] Z. Zong, M. Babaie and R. B. Staszewski, "A 60 GHz Frequency Generator Based on a 20 GHz Oscillator and an Implicit Multiplier," *IEEE Journal of Solid-State Circuits*, vol. 51, no. 5, pp. 1261-1273, May 2016.

Paper V

Paper V

A 10 mW mm-Wave Phase-Locked Loop with Improved Lock Time in 28-nm FD-SOI CMOS

Based on

Mohammed Abdulaziz, Therese Forsberg, Markus Törmänen and Henrik Sjöland, "A 10 mW mm-Wave Phase-Locked Loop with Improved Lock Time in 28-nm FD-SOI CMOS," submitted to *Transactions on Microwave Theory and Techniques*.

A 10 mW mm-Wave Phase-Locked Loop with Improved Lock Time in 28-nm FD-SOI CMOS

Mohammed Abdulaziz, Member, IEEE, Therese Forsberg, Student Member, IEEE, Markus Törmänen, Senior Member, IEEE, Henrik Sjöland, Senior Member, IEEE.

Abstract—This paper presents a phase-locked loop (PLL) architecture that considerably improves the lock time. The architecture is demonstrated by a millimeter-wave PLL, that also features a novel double injection-locked divide-by-3 circuit and a charge-pump mismatch compensation scheme, resulting in stateof-the-art power consumption and jitter performance. The PLL is implemented in a low power 28-nm FD-SOI CMOS process and its output frequency is centred at 54.5 GHz. The in-band phase noise is between -93 and -96 dBc/Hz across the tuning range, and the integrated jitter is 176 fs to 212 fs. The total power consumption of the PLL is only 10.1 mW, resulting in a best case PLL figure-of-merit of -245 dB, and the active area is just 0.19 mm².

Index Terms—60 GHz, 5G, charge pump, CMOS, divide-bythree, fast lock time, frequency synthesizer, ILFD, injectionlocked divider, local oscillator, low phase noise, low-power, mmwave, phase-locked loop, PLL.

I. INTRODUCTION

PHASE-LOCKED LOOP (PLL) frequency synthesizers are key parts of today's wireless transceivers, and designing PLLs for millimeter-wave (mm-wave) communication standards is challenging from many aspects. Their phase noise limits the highest achievable modulation order [1], and they also need to have fast settling and be able to operate at low power in battery operated devices [2]–[4]. In particular, fast settling time is required for communication standards that support very high data rates, to avoid losing large amounts of data during frequency locking. In frequency-modulated continuous-wave (FMCW) radar applications a fast settling PLL is required to be able to receive baseband signals at higher frequencies, and hence reduce the impact of the high flicker noise corner in short-channel technologies [3], [4].

A frequency synthesizer operating at about 60 GHz can be used for direct conversion tranceivers in the unlicensed 60 GHz band, where for instance the WiGig/IEEE 802.11ad standard [2], future 5G standards [5], and high precision radars [3] will

Manuscript received Month Date, 2018; revised Month Date, 2018; accepted Month Date, 2018. This paper was approved by Editor Name.

The first two authors contributed equally to this work.

This work was supported by the Swedish Research Council and the European Commission in the framework of the H2020-ICT-2014-2 project Flex5Gware (Grant agreement no. 671563).

M. Abdulaziz was with the Department of Electrical and Information Technology, Lund University, SE 221 00 Lund, Sweden and with the Department of Microtechnology and Nanoscience, Chalmers University of Technology, SE 412 96 Göteborg, Sweden. He is now with Ericsson Research, Ericsson AB, 223 62 Lund, Sweden (e-mail: mohammed.abdulaziz@ericsson.com).

T. Forsberg, M. Törmänen and H. Sjöland are with the Department of Electrical and Information Technology, Lund University, SE 221 00 Lund, Sweden. (e-mail: therese.forsberg@eit.lth.se, markus.tormanen@eit.lth.se, henrik.sjoland@eit.lth.se).



Fig. 1. Example of a mm-wave frequency generation architecture for 5G applications that utilizes the presence of an RF PLL.

reside. Such a frequency synthesizer, followed by a divide-by-2 circuit, can also be used to generate quadrature signals for 28 GHz front-ends intended for emerging 5G applications. It can also be used as the local oscillator (LO) in dual-frequency conversion transceivers, by using the 60 GHz PLL output signal together with the divided 30 GHz or 20 GHz signals present in the PLL, depending on whether the first division step after the voltage controlled oscillator (VCO) divides by 2 or 3. Higher frequencies in the E-band can then be targeted. Clearly, high performance and low cost 60 GHz PLLs have many potential applications.

The anticipated 5G wireless communication systems will support mm-wave links together with lower frequency cellular ones. This means that a mm-wave PLL operating with a high reference frequency (f_{REF}) may have another PLL as its input, instead of a crystal oscillator (XO). This has been the case in recent works and prior art, such as [6]-[8]. Similarly, a radio frequency (RF) f_{REF} can be provided by direct digital synthesis, which has been demonstrated in several state-of-theart sub-millimeter frequency imaging radar systems [9]-[11]. Increasing f_{REF} is beneficial for the PLL noise performance, because when the f_{REF} noise is added to the PLL in-band noise it is first multiplied by the PLL division ratio squared. If the loop bandwidth is kept low, a PLL with high f_{REF} may also decrease the current consumption in the charge pump (CP). However, if not addressed properly, this approach will inevitably lead to problems with prolonged settling time. An additional advantage of using the PLL for cellular bands as an input to the mm-wave PLL in 5G systems, is that the RF PLL can provide the fine-grain resolution required for channel selection and/or modulation and thus simplify the design of the mm-wave PLL. The architecture of such a system is shown in Fig. 1.

In this paper a new PLL architecture is presented. Tuned to 56 GHz, the PLL is aimed at 5G applications, but it also has the more general goal of demonstrating an architecture that balances low noise, low power consumption and fast settling time, when a restriction is that the f_{REF} is several hundreds times the PLL loop bandwidth. The demonstrated PLL also

features a novel double injection-locked divide-by-3 circuit, that achieves a wide lock range at low current consumption. Furthermore, the CP includes a novel current mismatch mitigation technique based on negative feedback. To support rail-torail output signals and hence wider VCO frequency range, the CP also features an improved operational amplifier that allows operation over a large common-mode range. The presented PLL shows competitive performance at a power consumption of just 10.1 mW, a value which to our knowledge is the lowest presented at such high output frequency. The PLL architecture is first introduced in Section II. Detailed description of circuit design is then presented in Section III. The measurement results together with a comparison to the state-of-the-art are then presented in Section IV. Finally, the paper is concluded in Section V.

II. PLL ARCHITECTURE

A conventional type-II PLL is shown in Fig. 3(a). Increasing f_{BEF} is an effective technique to achieve lower phase noise [6]–[8]. Traditionally, using a higher f_{REF} means that the loop bandwidth can be higher, which in turn leads to an improved lock time. However, when the reference frequency becomes very high, the loop bandwidth is no longer limited by the reference frequency. Instead, if minimum jitter is targeted, the loop bandwidth should be chosen as the frequency where the VCO phase noise and extrapolated in-band phase noise at the PLL output intersect. Generally, for state-of-the-art mm-wave PLLs a typical bandwidth for optimum noise performance is in the order of a few MHz. If f_{REF} is then chosen to be about 2 GHz, the ratio of f_{REF} to loop bandwidth becomes very high. To still get the desired bandwidth when using such a high f_{REF} , either the loop filter (LF) capacitances must be increased, or the charge pump current (I_{CP}) must be reduced. Increasing the size of the LF leads to a large area for the capacitors, while reducing I_{CP} is an attractive way of reducing the overall power consumption. However, this will also limit the available output current from the CP that charges the filter capacitances, and the PLL lock time will be severely degraded due to so-called cycle slips, originating from nonlinear effects in the transient when the PLL is out of lock-in range [12]. An example of this is shown in Fig. 2, where the simulated VCO control voltage settling behavior for two PLLs with the same bandwidth is presented. One uses an f_{REF} that is 8 times higher, and to keep the loop bandwidth unchanged the I_{CP} is then also reduced 8 times. In the case of a high f_{REF} to-bandwidth ratio, it can be seen that the cycle-slips prevents the PLL from approaching the correct frequency exponentially, and that the settling time is significantly prolonged.

It is thus clear that aggressively increasing the f_{REF} to obtain better noise performance affects the settling time, and that low phase noise and fast settling time are contradictory in this case. To solve this conflict of requirements we propose a PLL architecture that can achieve fast settling by disregarding the noise performance during PLL settling, and then seamlessly shift to a low noise mode at steady state, thereby achieving both fast settling and low noise. The proposed PLL architecture, that features two such optimized modes of operation, is shown in Fig.3(b). Fast settling mode is



Fig. 2. Example of the impact of cycle-slips. Both cases have the same loop bandwidth, but f_{REF} is 8 times larger and I_{CP} is 8 times smaller in case 2

enabled when SW_{fast} is set to logic 1. Both multiplexers then forward the signal divided by an extra factor N_2 . The effective reference frequency becomes f_{REF}/N_2 , and the divided VCO frequency to be compared to it by the phase frequency detector (PFD) becomes $\frac{f_{VCO}}{N_1 N_2}$. At the same time, the amplitude of the current pulses fed to the loop filter from the CP become $N_2 I_{CP}$. This mode has an increased maximum current that can charge the capacitors in the loop filter, which yields a faster settling. Low noise mode is enabled when setting SW_{fast} to logic 0. Both multiplexers then forward f_{REF} and f_{VCO}/N_1 without the extra division. The amplitude of the CP current pulses is at the same time reduced to I_{CP} . The reduced CP current increases the settling time, but the total power consumption is reduced and the decrease in total division ratio improves the in-band phase noise of the PLL. In this work, f_{REF}/N_2 is chosen to be 8, large enough to demonstrate the efficiency of the mode-switching architecture.

It is important to note that for either mode setting, the small signal PLL characteristics remain the same. One CP is switched on or off, but a seamless transition between modes is possible because the steady state value of V_{ctrl} is the same in both modes and no loop filter reconfiguration is needed. The signal SW_{fast} can be easily generated when the current operating frequency is to be changed. As will be demonstrated by measurements in Section IV, the mode transition will indeed be seamless and not generate any sudden transients. Therefore, the mode change back to low-noise mode can be performed simply after a predefined delay, without any need of calibration.

III. PLL IMPLEMENTATION AND CIRCUIT DETAILS

A. The VCO and divider chain

The PLL was implemented in a low power 28-nm CMOS SOI process. The VCO and divider chain is shown in Fig. 4. The phase noise performance of the cross-coupled pair VCO is improved by using the tail current filtering technique [13], which is effective also at 60 GHz [14]. The VCO output is fed differentially to an injection-locked frequency divide-by-3 circuit (ILFD), as well as to two single-ended buffers, for measurement purposes. Simulations show that the full differential VCO voltage swing of 2.7 V peak-to-peak can be preserved, even if the VCO output signal is also required to drive additional, mainly capacitive loads, such as divide-by-two ILFD circuits or a mixer. The differential output of the



Fig. 3. Schematics of (a) a type II PLL and (b) the proposed PLL architecture, where $SW_{fast} = 1$ activates fast-locking mode, and $SW_{fast} = 0$ activates the low-noise mode.

ILFD is fed directly to a differential latch-based divide-by-2 circuit, followed by true single phase clock (TSPC) dividers and a multiplexer [15]. The first stage of the TSPC divider divides the frequency of the input by 4 or 6, depending on if the *Sel* control signal is set to logic 0 or 1, respectively. This signal is used to perform step response measurements. Based on whether the PLL is to operate in low-noise mode or fast-settling mode, the control signal SW_{fast} is set to either logic 1 or 0. The sinusoidal reference signal is buffered and converted to a square wave on chip, and fed to the PFD though a multiplexer in a similar configuration as in the divider chain path.

For mm-wave PLLs it is preferable to use a division ratio higher than two in the first feedback divider stage after the VCO, if it can be achieved without increasing the power budget. At such frequencies, however, as explored in [16], [17], special architectures suited for mm-wave operation have to be employed. One such architecture that has lately gained increased attention due to its attractive properties, such as low power consumption and small area, is the mm-wave dynamic current mode logic (DCML) divider. The operation at mmwave frequencies is attributed to that the memory elements in the latches are the parasitic capacitances of the active devices. Unfortunately, this kind of divider is sensitive to process, voltage and temperature (PVT) variations. Injection locking is a more attractive technique in terms of robustness, since LC tanks are then used to tune the divided frequency output, and ILFDs with higher division ratios such as 3 have been demonstrated [18]-[20]. However, current consumption and locking range remain major concerns when increasing the division ratio, due to less effective current injection. The ILFD presented in this work addresses these issues.

Differential ILFD divide-by-two operation is typically based on injecting a current signal at a frequency close to the second harmonic of the ILFD LC resonance (f_0), at the source of the current commutation MOS pair, as shown in Fig. 5(a) [21]. The current signal is then injected by the tail transistor M₁ which acts as a transconductance. Note that M₂ and M₃ act as a single balanced mixer with a maximum conversion gain of $G = \frac{2}{\pi}$, down-converting the injected signal to a frequency close to f_0 at the LC tank. A more general observation is that injection locking division is achieved by injecting a harmonic signal to the circuit that by some mechanism results in a current signal at the drains of M_2 and M_3 , with a frequency close to f_0 and a magnitude sufficiently large to pull the oscillator to that frequency. An upper bound of the frequency locking range for the ILFD shown in Fig. 5(a) is expressed by [21], [22]

$$\omega_{range} \le \frac{\omega_0}{2Q} \cdot \frac{2}{\pi} \cdot \frac{I_{inj}}{I_{osc}} \tag{1}$$

where Q is the quality factor of the divider's LC tank, I_{inj} the magnitude of the injected current, and I_{osc} the magnitude of the free running oscillator current at f_0 . In (1) $G = \frac{2}{\pi}$ is used assuming M₂ and M₃ switch on/off abruptly [22]. Injecting a voltage V_{inj} at the gate of M₁ makes the frequency locking range ω_{range} become

$$\omega_{range} \le \frac{\omega_0}{2Q} \cdot \frac{2}{\pi} \cdot \frac{G_m \cdot V_{inj}}{I_{osc}} \tag{2}$$

It is important to note that ω_{range} is referred to the output frequency of the divider. Clearly, a way to improve the lock range without compromizing performance, such as reducing Q, is by maximizing I_{ini} (i.e. increasing the injection efficiency).

To make the ILFD to divide by 3, the current can be injected directly to the output using differential back-to-back MOS devices, as shown in Fig. 5(b) [20], [23]. In this work a novel double injection-locked divide-by-3 circuit is proposed, see Fig. 5(c). The proposed divider uses double injection to achieve increased injection efficiency, enabling a wide tuning range at a reduced current consumption. The triode-multiplier constituted by devices M₄ and M₅ multiplies the injected voltage (V_{inj}) at frequency $f \approx 3f_0$ with that of the divider output, tuned to f_0 . It was shown in [24] that the output voltage of the triode-multiplier is comprised of even order harmonics. In this case, the multiplier is excited with rail-to-rail signals from the VCO and therefore it acts as an efficient voltagemode single-balanced mixer with a differential input and a single-ended output. The output of the multiplier is simply the even order inter-modulation terms, which in locked state are at frequencies close to $2f_o$, $4f_o$ and higher even order harmonics. These tones are in turn fed to the tail current device, which acts as a transconductance that injects current mainly at $\approx 2f_o$



Fig. 5. Injection-locked frequency dividers. (a) Divide-by-2. (b) Divide-by-3. (c) Proposed novel double injection divide-by-3.

to the source of the active pair (M₂ and M₃). This results in an additional injection mechanism similar to that of the divideby-2 ILFD shown in Fig. 5(a). Higher order harmonics are not significant, as they are suppressed by the circuit. The triodemultiplier also injects a current signal at the first harmonic, $I_{inj,direct}$ directly to the output. Hence, the lock range of the divider is proportional to the sum of lock ranges of the two dividers in Fig. 5(a) and 5(b). Including the transfer of the passive mixer $V_{inj} = \frac{2}{\pi}V_{out}$ in (2) and adding the direct injection gives

$$\omega_{range} \le \frac{\omega_0}{2Q} \cdot \left(\frac{2}{\pi^2} \cdot \frac{g_m V_{out}}{I_{osc}} + \frac{I_{inj,direct}}{I_{osc}}\right) \tag{3}$$

Equation (3) presents an upper bound to the lock range of the double injection-locked divider, indicating that the lock range can be increased considerably compared to the directinjection-only circuit, without any current penalty. The current budget can also be reduced significantly, since with the second injection path established, the first injection devices (M_4 and M_5 in Fig. 5(c)) can be small, increasing the *LC* resonator impedance. Further efficiency enhancement is also achieved due to the push-push regime that the tail current source M_1 is operated in.

The designed divider covers the frequency range of the VCO with a margin, to account for PVT variations. Simulations of the divider sensitivity with single injection and with the proposed double injection are shown in Fig. 6. At the same power consumption, with an input power of 5 dBm and at a fixed varactor voltage of VDD/2, the double injection increases the locking range from 2.5% to 8.5%, i.e. by a factor of 3.4. All possible spurs created by the ILFD are harmonics of the divided signal and, since the output is differential, the odd order harmonics dominate. Directly after the ILFD are inverting buffers that aim to make the signal even more squarewave shaped and thus increase the odd harmonics even further. The spurs of the ILFD are thus of little concern in this case.



Fig. 6. The simulated sensitivity of the ILFD at a fixed varactor voltage of VDD/2, with and without the double injection, at the same power consumption.



Fig. 7. Differential high speed CMOS latch.

The simulated phase noise of the ILFD at 10 MHz offset from the carrier is below -141 dBc/Hz, with a noise of -146 dBc/Hz close to its self-resonance frequency.

The ILFD is followed by a divide-by-2 circuit as shown in Fig. 5 (a). The latch proposed in [25], shown in Fig. 7, can be used directly after the ILFD thanks to the high speed 28-nm technology, while the last divider stages are implemented in TSPC logic.

B. The PFD-CP

The three-state PFD used in this work is TSPC-based and produces UP/DOWN pulses, which are converted to differential signals by inverters. Transmission gates are used to match the inverter delay, thereby mitigating imbalance between the differential signals.

The CP schematic is shown in Fig. 8, using a differential architecture $(M_1 - M_6)$ for high speed and reduced charge sharing. The UP/DOWN currents are matched to a first order through a 1:1 current mirror $(M_1, M_6, M_{11} - M_{13})$. Due to the reduced feature size, however, the channel length modulation causes mismatch between the UP and DOWN currents at high and low values of the output voltage V_{ctrl} . This results in different UP and DOWN pulse widths at steady state, where the pulse with less current must become wider to compensate for the current difference and produce zero net charge to

the LF. This issue results in increased spur levels and noise contribution from the CP, as described below.

In [26], [27] it was shown that high frequency noise folding due to CP gain mismatch can result in large PLL in-band phase noise increase, especially in fractional-N PLLs with strong $\Sigma\Delta$ -modulation noise. This effect is even more pronounced by charge injection, further increasing gain mismatch in the cross-over region, resulting in more noise folding. To reduce charge injection, an operational amplifier OP1 is therefore used to make the voltage of the dummy node in the differential CP track V_{ctrl} [28]. Furthermore, with increased f_{REF} , the contribution of the CP to the PLL phase noise becomes more significant. This in-band contribution is [29], [30]:

$$\mathcal{L}_{CP} = \frac{2k_B T \gamma g_m n_T}{(I_{CP}/2\pi)^2} \cdot (T_p/T_{REF}) \cdot (1 + \frac{(T_p/T_{REF})f_c}{f_{offset}}) \cdot N^2$$
(4)

where $T_{REF} = 1/f_{REF}$, k_B is the Boltzmann constant, T the absolute temperature, γ the MOS gamma factor, g_m the transconductance of the devices in the 1:1 current mirror, n_T the number of MOS devices used to copy current to the CP, f_c the 1/f noise corner of the MOS devices, and T_p the current pulse width in steady state (in this work $T_p \approx 20$ ps). This relation indicates that the noise contribution of the CP remains constant if I_{CP} is scaled down by the same factor as the PLL division ratio N. However, T_p is dependent on the speed of the latches in the PFD and hence do not scale with f_{REF} . As T_{REF} is reduced the CP phase noise contribution will therefore increase, both thermal noise and even more so 1/fnoise. A mismatch in UP/DOWN currents will result in wider pulses, as the minimum pulse width is set by the PFD reset delay. The pulses will then contain more charge and contribute more phase noise. There are thus two mechanisms that cause increased phase noise due to CP mismatch, high frequency noise folding and loop filter noise injection, motivating a technique to counteract mismatch.

To reduce the mismatch between UP and DOWN currents, compensation of the channel length modulation effect is required [31]. The schematic of the CP is shown in Fig. 8, where an additional dc current branch $(M_7 - M_{10})$ has been introduced, with 1:1 replicas of (M_1, M_3, M_5, M_6) . A negative feedback loop, using amplifier OP2 with a push-pull output stage (M_{14}, M_{15}) , controls the UP current so that the dc current branch outputs V_{ctrl} . Since the NMOS and PMOS currents are equal in the dc branch, and it has the same output voltage as the CP, and replicated devices, the UP and DOWN currents must also be equal.

The operational amplifiers used in the CP needs to have high gain as well as capability to handle rail-to-rail commonmode signal levels. This is crucial as the CP output voltage range limits the PLL frequency range. The schematic of a conventional amplifier proposed in [32] is shown in Fig. 9(a). All the devices are biased in weak inversion for low current consumption, keeping in mind that only low frequency signals are processed. The input differential pair limits the minimum common-mode input voltage, and an improved version is shown in Fig. 9(b), where the input stage is a differential pair in parallel with a differential cross-coupled PMOS source follower. As the common mode input voltage drops below



Fig. 8. Schematic of the charge pump with replica-based feedback loop to correct for UP/DOWN current mismatch.

a threshold voltage, the common source gain is drastically reduced. However, inserting the PMOS source follower in parallell, which turns on under these conditions, preserves some of the gain. Since the differential pair common source amplifiers are loaded with diode connected devices, the voltage gain from the gate to drain is not high, making the gain of the common source an the source follower more similar. This technique has also been used in [24] to improve the linearity of operational transconductance amplifiers. Simulation of the voltage gain with and without the proposed technique is shown in Fig. 10. As can be seen in the figure, using a PMOS source follower helps providing gain even at 0 V commonmode input. Traditionally the amplifier shown in Fig. 9(a) has been used without phase compensation, however, in this case the high gain of the dc loop required compensation, realized by the resistor-capacitor link at the output. The 20 pF capacitor creates a dominant pole, and the resistor a high frequency zero, advancing the phase.

The CP was simulated when driven by two in-phase signals with a frequency of 2 GHz, to resemble the PLL locked state with DIV and REF matched in phase and frequency. The CP output was forced to a fixed voltage and the output current was observed. The net output current then represents the mismatch between UP and DOWN currents. The simulated mismatch versus CP output voltage with/without the proposed CP mismatch correction loop is shown in Fig. 11. As can be seen, the current mismatch is reduced to less than 0.1% in the range from 0.2 V to 0.8 V. The excessive mismatch near supply (1 V) and ground voltage proximity is due to the V_{dsat} drop required over the tail current devices M₁ and M₆ shown in Fig. 8.

IV. MEASUREMENT RESULTS

Before designing the complete PLL, a stand-alone ILFD was designed and fabricated. For comparison it shares a chip with an identical ILFD - but without the double injection path. Both the ILFDs and the subsequent PLL were fabricated in the STMicroelectronics 28-nm ultra-thin body and buried oxide

(UTBB) fully-depleted (FD) silicon on insulator (SOI) CMOS process with 10 metal layers for interconnect, and with the metal-insulator-metal (MIM) capacitance option. The active area of each divider is 0.05 mm² and the total area of the PLL chip is 0.9 x 0.9 mm², of which the active area is just 0.17 mm². The ILFD die microphotograph is shown in Fig. 12, and that of the PLL chip overlaid with a layout image in Fig. 13. Both chips were mounted on FR-4 printed circuit boards, to which all needed supply, bias and signal pads were wirebonded. Only the mm-wave signals were probed, using Infinity microprobes from Cascade Microtech. The PLL measurement setup is depicted in Fig. 14. The input reference frequency was generated by an Agilent E4438C signal generator with low-noise option UNJ. For the phase noise measurements, an FSWP phase noise from Rohde & Schwarz with harmonic mixers for the 50 to 75 GHz band was used. The FSWP also provided the setup with low-noise supply voltages. An FSU50 spectrum analyzer with a harmonic mixer, also from Rohde & Schwarz, was used for the output spectrum measurements. The loop control voltage measurements used a 4 GHz 20 GS/s Rohde & Schwarz RTO 1044 digital oscilloscope. Pulses for the mode switch control were generated by a WW2572A 250 MS/s waveform generator from Tabor Electronics. To not introduce noise on the sensitive loop control voltage node, the oscilloscope was disconnected during phase noise measurements.

Measurement results from the the ILFD are shown in Fig. 15, and the two versions are compared in Table I. When utilizing the full range of the varactor, the measured locking range is increased from 13% for the single injection circuit, to 17.5% for the double injection circuit. Even if these ILFD measurements use less than 0 dBm input signal due to measurement setup limitations, which is less than the expected VCO output voltage in the PLL, also the single-injection ILFD demonstrates a wide locking range covering the VCO frequencies. However, using the single injection ILFD in the PLL would allow for almost no drift in tuning due to process and temperature variations between the VCO and



Fig. 9. (a) The amplifier proposed in [32] and (b) improved amplifier with parallel source follower stage for rail-to-rail common-mode operation and RC narrow-banding for loop stability.



Fig. 10. Simulated low-frequency gain of the amplifier in Fig. 9, with and without the parallel source follower.



Fig. 11. Simulated CP current mismatch, with and without feedback loop, at 1 V supply.

the ILFD. Since they will be tuned together by the same control voltage, the key concern is how sensitive the mm-wave parts of the PLL are to unforeseen tuning mismatch, and the robustness is dependent on the locking range of the ILFD for a fixed varactor control voltage. At a fixed varactor voltage of VDD/2, measurements show that the double injection technique increases the lock range by more than a factor of two, from 2.4% to 5.2%.

When measuring the PLL, deviations from simulated performance were found. One such difference was that the outof-band noise was worse than anticipated from measurements of earlier stand-alone VCOs in the same CMOS process [14], possibly due to the decision to use a single ended output buffer for the VCO signal, or due to noise coupling to the loop control voltage node. This was addressed by increasing the PLL



Fig. 12. ILFD chip photo. The active area is inside the white rectangles.

TABLE I Comparison Between Measured Results for the Single and Double Injection Circuits

At approximately 0 dBm	Single injection	Double injection		
input power				
Locking range at 2.5 mW	53 - 60.5 GHz	52.8 - 62.9 GHz		
	(13%)	(17.5%)		
Locking range at 1.5 mW	54 - 60.9 GHz	53.5 - 62 GHz		
	(12%)	(15%)		
Locking range at 2.5 mW	54.1 - 55.4 GHz	54.1 - 57 GHz		
when $V_{varactor} = VDD/2$	(2.4%)	(5.2%)		

bandwidth. The in-band noise was also higher than expected from simulations, which was alleviated by the decision to increase the CP current, and to complement the internal loop filter with additional capacitance, connected externally. The choice of external components was a capacitor twice the size of the internal one, in series to ground with a resistor half the size of the internal one. If implemented on chip, the extra loop filter would add approximately 0.02 mm² of active area, making the new total active area 0.19 mm².

All PLL performance measurements were taken using the same settings applied to a single chip in room temperature. Subsets of the full measurement set were also taken on three additional chips, to investigate variations between samples. The supply voltages for the VCO, ILFD, 20 GHz latch, and



Fig. 13. The chip micrograph with the layout overlayed. The active area of 0.17 mm^2 is inside the white rectangles.

the output buffer were set to 0.8 V, and the supply voltages for the low frequency dividers, PFD, CP and CP amplifiers were set to 1.06 V. The regular and large CP currents were set to 300 μ A and 7 x 300 = 2400 μ A, respectively. The total power consumption of the PLL was then measured to 15.2 mW in fast-locking mode, and 10.1 mW in low-noise mode. The measured power consumption of each contributor is visualized in Fig. 16. It is apparent that the CP current results in a significant power consumption reduction when the PLL operates in low-noise mode. However, the full effect cannot be clearly seen in the figure since the CP shares supply with the PFD, which in low-noise mode will operate at a frequency 8 times higher, hence consuming more power. Since the PLL will be in low-noise mode nearly all the time, the power in low-noise mode will be presented as the PLL power consumption.

The PLL frequency range, i.e. the range where the PLL is able to acquire lock, was measured to 52.87 to 56.81 GHz. At the edges of this range, however, the reduced CP and VCO gain reduce the PLL bandwidth, which along with increased CP current mismatch result in degraded phase noise performance. Hence, the useful frequency range of the PLL is therefore where the phase noise performance is relatively uniform. This was measured to be between 53.2 and 56.1 GHz, a 5.3% range, in which the jitter deviated by less than 1.6 dB from the minimum value of 176 fs, see Fig. 17. While the measured tuning range, which is 1.45 GHz when divided by two, is wide enough to cover the upcoming 5G band in the USA, that is 0.85 GHz located between 27.5 and 28.35 GHz [5], it still requires some slight tuning. To ensure that the intended bands are covered even in the presence of process variations, the PLL frequency range could be increased. An increased PLL output frequency range will also have the added benefit of making the PLL useful in more applications. The output frequency range of the presented PLL is mainly defined by VCO tuning range. The most straight-forward way

to increase this is to increase the VCO varactor size or to include switched capacitor arrays in the VCO tank, resulting in a trade-off between phase-noise and tuning range.

The output spectrum at a PLL output frequency of 54 GHz was measured, see Fig. 18. The PLL was in low-noise mode and the total division ratio of the PLL was set to 24. To access the differential VCO signal outside the chip it was fed to two single-ended on-chip buffers, one of which was accessible by on-chip probing. The buffers were sized down to produce a signal of lower power, to reduce coupling to sensitive parts of the circuit, as well as supply ripple due to high frequency current through bond wires. Because of this, the measured offchip output signal power was about -27 dBm. Note that this is just for measurement purposes, and that the PLL is intended to deliver its output signal to a transceiver on the same chip. Since the measurement setup lacked external amplifiers and the used external mixer has high signal loss, the measured spectrum has a high noise floor. As can be seen in the PLL output spectrum in Fig. 18, the external harmonic mixer also makes the noise floor itself undulate, and creates spurs. The reference spurs, at a distance of 2.25 GHz from the carrier, are measured to be below -61 dBc.

The PLL phase noise measurements in Fig. 17 show the phase noise and RMS jitter across the PLL frequency range, in low-noise and fast-locking mode. The total division ratio was set to 24, by setting the *Sel* signal in Fig. 4 to a logic 0. The RMS jitter was calculated from the measurements by integrating phase noise from 1 kHz to 30 MHz offset frequency. The best measured phase noise performance was at an output frequency of 54 GHz, where the RMS jitter was 176 fs and the phase noise at 1 MHz and 10 MHz offset were -95.7 dBc and -103.5 dBc, respectively. The measured in-band phase noise at 1 MHz offset stayed below -93 dBc in all low-noise mode when used as a clock source, the measured phase noise of this mode will be presented as the overall PLL performance.

The measured noise includes a negligible noise contribution from the frequency reference signal, see Fig. 19. If a CMOS state-of-the-art, low-jitter 2.2 GHz PLL, such as [33], [34], is used as the input reference frequency generator, the integrated jitter is estimated to rise from 176 fs to between approximately 200 and 230 fs.

Separate phase noise measurements were conducted with the current source matching in the CP disabled. They indicate that the technique does help reduce the phase noise, especially at lower offset frequencies, as expected. A measurement showing the impact on the phase noise in low-noise mode is shown in Fig. 20.

To estimate the settling time, and to verify concept of the mode-switching PLL, the loop control voltage was measured during settling with a digital oscilloscope. To introduce a step in the PLL, a pulsed input signal was applied to the *Sel* signal that controls the division ratio of the divide-by-6-or-4 divider in the feedback path, see Fig. 4, while the reference frequency was kept constant. That means that the PLL goes from an unlocked state, when the division ratio times the reference frequency does not fall inside the VCO tuning range,



Fig. 14. The PLL measurement setup.



Fig. 15. Measured locking range at a power consumption of 2.5 mW for the double injection version of the ILFD.



Fig. 16. Measured power consumption of each contributor in low-noise and fast-locking mode.

to a locked state. When Sel = 0, the total division ratio is equal to 24, and if the loop is then in an unlocked state, the targeted frequency is below the VCO tuning range and the control voltage is at its minimum. When Sel = 1, on the other hand, the total division ratio is equal to 36, and if then in the unlocked state the targeted frequency is above the VCO tuning range, and the control voltage is instead at its maximum. Another pulsed signal was used to control the PLL mode switch. In Fig. 21 the settling behaviour with the mode signals switch at the same time, the settling behaviour will be that of the low-noise mode. When the mode



Fig. 17. Measured phase noise of the PLL in (a) fast-locking mode and (b) low-noise mode.

switch signal is delayed, the settling instead starts in fastlocking mode, followed by a switch to low-noise mode. The estimated settling time in low-noise mode is about three times longer than the estimated settling time using fast-locking mode during the first part of the settling. The estimated maximum settling time using fast-locking of 3 μ s was determined as the PLL settling time. The figure also shows that the switching between modes has minimal impact on the output frequency, and that the best time to switch from fast-locking to low-noise mode is when the frequency is close to stable. However, the measurements also show that since the switch can safely be made at any time, and that any time operating in fast-locking mode improves the settling time significantly, there is no need for complicated algorithms and feedback to control the mode switch mechanism.

The performance of the PLL is summarized in Table II.



Fig. 18. Measured PLL output spectrum in low-noise mode, with a division ratio of 24. The mixing products and undulating noise floor are due to the harmonic mixer of the spectrum analyzer.



Fig. 19. Measured phase noise of $f_{REF},$ scaled at output with 20log(N), and the best-case measured PLL phase noise.

The PLL figure-of-merit (FOM_{PLL}) is commonly used for wireless communication PLLs, and it is based on the theory in [35], where the phase noise figure-of-merit (FOM) for VCOs is extended to an entire PLL. Compared to state-of-the-art published mm-wave PLLs, the mode-switching PLL reported in this paper achieves comparable settling time, area and phase noise performance, but at much lower power consumption, which results in a state-of-the-art FOM_{PLL} of -245 dB for this frequency range. Even if the output frequency range is enough to cover the main intended use of the PLL and also enough to demonstrate the architecture, an increased range, attainable with small changes in the VCO, would make it useful for a wider range of applications.

V. CONCLUSION

A novel PLL for mm-wave frequency wireless transceivers is presented, that mitigates the problem of cycle slips during settling by switching between two modes of operation with the same small-signal, but different large-signal properties. Two key building blocks of the PLL include novel circuit techniques. The first is a double injection divide-by-3 circuit which increases the frequency lock range, allowing the power consumption of the mm-wave divider to be robustly scaled



Fig. 20. Measured phase noise of the PLL at 55 GHz in low-noise mode, with the CP current matching ON and OFF.



Fig. 21. Measured loop control voltage during settling. Three different cases of timing switching from fast-locking to low-noise mode are shown. (a) Settling to 56.34 GHz. (b) Settling to 53.232 GHz.

down to less than 0.8 mW. The second is the charge pump, which has a replica-based feedback loop to diminish the current mismatch due to channel length modulation, therefore reducing low-frequency PLL phase noise. Measurements show a PLL lock time of about 3 μ s using the fast-settling mode during the first part of settling, while then operating in low-noise mode achieves a record low power consumption of 10 mW and a state-of-the-art FOM_{PLL} of -245 dB for PLLs in the 60 GHz range.

TABLE II PLL PERFORMANCE SUMMARY AND COMPARISON TO STATE-OF-THE-ART MM-WAVE CMOS PLLS

Ref.	CMOS Tech. (nm)	Туре	Ref. Freq. (MHz)	Power (mW)	Phase Noise @1MHz (dBc/Hz)	RMS Jitter (fs)	Lock Time (µs)	Center Freq. (GHz)	Freq. Range (%)	Core Area (mm2)	Ref. Spur (dBc)	FOM _{PLL} ¹ (dB)
[36]	65	Fund. LO	135	24.6	-89.9 to - 91.5	-	> 5 ²	63.1	16.5	0.192	-54.5	-
[37]	40	Subsampling PLL	40	42	-88 to -92	200 to 350	-	58.5	16.2	0.16	-40	-207.7 to - 202.9
[38] ³	65	Time-to-digital converter all digital PLL	100	48	-90	590.2	3	60	11.7	0.48	-74	-228
[39]	28	Fund. LO, frac. N 24 GHz PLL + 60 GHz QILO	27	107	-93.8 to - 96.5	900 ²	-	59.4	18.2	0.294	-	-221
[40]	65	All digital PLL	100	46	-88 to -94.5	223 to 302.5	< 1 ⁴	58	27	0.45	-48.3 /-52.2 ⁵	-236 to -233
[41]	65	Subsampling 20 GHz PLL + 60 GHz QILO	36 or 40	32	-922	290	-	60.4	16	1.086	-73	-236
This work	28	Fund. LO, inte- ger N	2290	10	-93 to -96	176 to 212	3	54.5	5.3	0.197	-61	-245 to -243

FOM = 10log((⁰/_{1s})² P_{PL1}/_{1mW}), defined in [35] 2) Estimated from graph. 3) Frequency generation part only. 4) Simulated. 5) Fractional spur.
 6) Including pads. 7) Including estimated area for implementation if the externally added loop filter is placed on-chip instead.

ACKNOWLEDGMENT

The authors thank STMicroelectronics for the chip fabrication. The authors also gratefully acknowledge Dr. D. Kuylenstierna at Chalmers University of Technology and Rohde & Schwarz for their support with obtaining measurement instruments, and the Chalmers Microwave Electronics Laboratory for providing an environment for the measurements.

REFERENCES

- [1] T. Pollet, M. V. Bladel, and M. Moeneclaey, "BER sensitivity of OFDM systems to carrier frequency offset and Wiener phase noise," *IEEE Transactions on Communications*, vol. 43, no. 2/3/4, pp. 191–193, Feb 1995.
- [2] "IEEE Standard for Information technology-Telecommunications and information exchange between systems Local and metropolitan area networks-Specific requirements - Part 11: Wireless LAN Medium Access Control (MAC) and Physical Layer (PHY) Specifications," *IEEE* Std 802.11-2016 (Revision of IEEE Std 802.11-2012), pp. 1–3534, Dec 2016.
- [3] W. Wu, R. B. Staszewski, and J. R. Long, "A 56.4-to-63.4 GHz Multi-Rate All-Digital Fractional-N PLL for FMCW Radar Applications in 65 nm CMOS," *IEEE Journal of Solid-State Circuits*, vol. 49, no. 5, pp. 1081–1096, May 2014.
- [4] Y. A. Li, M. H. Hung, S. J. Huang, and J. Lee, "A fully integrated 77GHz FMCW radar system in 65nm CMOS," in 2010 IEEE International Solid-State Circuits Conference - (ISSCC), Feb 2010, pp. 216–217.
- [5] "3GPP TR 38.815 V15.0.0 New frequency range for NR (24.25-29.5 GHz)," 3GPP(Release 15), pp. 1–22, July 2018.
- [6] A. Axholt and H. Sjöland, "A 60 GHz receiver front-end with PLL based phase controlled LO generation for phased-arrays," *Analog Integrated Circuits and Signal Processing*, vol. 80, no. 1, pp. 23–32, 2014.
- [7] W. El-Halwagy, A. Nag, P. Hisayasu, F. Aryanfar, P. Mousavi, and M. Hossain, "A 28-GHz Quadrature Fractional-N Frequency Synthesizer for 5G Transceivers With Less Than 100-fs Jitter Based on Cascaded PLL Architecture," *IEEE Transactions on Microwave Theory and Techniques*, vol. 65, no. 2, pp. 396–413, Feb 2017.
- [8] S. Ikeda, H. Ito et al., "A -244-dB FOM High-Frequency Piezoelectric Resonator-Based Cascaded Fractional-N PLL With Sub-ppb-Order Channel-Adjusting Technique," *IEEE Journal of Solid-State Circuits*, vol. 52, no. 4, pp. 1123–1133, April 2017.

- [9] G. Rubio-Cidre, A. Badolato, L. Úbeda Medina, J. Grajal, B. Mencia-Oliva, and B. Dorta-Naranjo, "DDS-Based Signal-Generation Architecture Comparison for an Imaging Radar at 300 GHz," *IEEE Transactions* on Instrumentation and Measurement, vol. 64, no. 11, pp. 3085–3098, Nov 2015.
- [10] K. B. Cooper, R. J. Dengler *et al.*, "Penetrating 3-D Imaging at 4- and 25-m Range Using a Submillimeter-Wave Radar," *IEEE Transactions* on *Microwave Theory and Techniques*, vol. 56, no. 12, pp. 2771–2778, Dec 2008.
- [11] D. A. Robertson, P. N. Marsh et al., "340-GHz 3D radar imaging test bed with 10-Hz frame rate," vol. 8362, 2012, pp. 8362 – 8362 – 11. [Online]. Available: https://doi.org/10.1117/12.918581
- [12] R. He, J. Li, W. Rhee, and Z. Wang, "Transient analysis of nonlinear settling behavior in charge-pump phase-locked loop design," in 2009 IEEE International Symposium on Circuits and Systems, May 2009, pp. 469–472.
- [13] E. Hegazi, H. Sjöland, and A. A. Abidi, "A filtering technique to lower LC oscillator phase noise," *IEEE Journal of Solid-State Circuits*, vol. 36, no. 12, pp. 1921–1930, Dec 2001.
- [14] T. Forsberg, J. Wernehag, A. Nejdel, H. Sjöland, and M. Törmänen, "Two mm-Wave VCOs in 28-nm UTBB FD-SOI CMOS," *IEEE Microwave and Wireless Components Letters*, vol. 27, no. 5, pp. 509–511, May 2017.
- [15] J. Yuan and C. Svensson, "High-speed CMOS circuit technique," *IEEE Journal of Solid-State Circuits*, vol. 24, no. 1, pp. 62–70, Feb 1989.
- [16] U. Decanis, A. Ghilioni, E. Monaco, A. Mazzanti, and F. Svelto, "A Low-Noise Quadrature VCO Based on Magnetically Coupled Resonators and a Wideband Frequency Divider at Millimeter Waves," *IEEE Journal of Solid-State Circuits*, vol. 46, no. 12, pp. 2943–2955, Dec 2011.
- [17] A. I. Hussein and J. Paramesh, "Design and Self-Calibration Techniques for Inductor-Less Millimeter-Wave Frequency Dividers," *IEEE Journal* of Solid-State Circuits, vol. 52, no. 6, pp. 1521–1541, June 2017.
- [18] H. Wu and L. Zhang, "A 16-to-18GHz 0.18-m Epi-CMOS Divide-by-3 Injection-Locked Frequency Divider," in 2006 IEEE International Solid State Circuits Conference - Digest of Technical Papers, Feb 2006, pp. 2482–2491.
- [19] T. N. Luo, S. Y. Bai, and Y. J. E. Chen, "A 60-GHz 0.13-muhboxm CMOS Divide-by-Three Frequency Divider," *IEEE Transactions on Microwave Theory and Techniques*, vol. 56, no. 11, pp. 2409–2415, Nov 2008.
- [20] Y. T. Chen, M. W. Li, H. C. Kuo, T. H. Huang, and H. R. Chuang, "Low-Voltage K - Band Divide-by-3 Injection-Locked Frequency Divider With Floating-Source Differential Injector," *IEEE Transactions on Microwave Theory and Techniques*, vol. 60, no. 1, pp. 60–67, Jan 2012.

- [21] H. R. Rategh and T. H. Lee, "Superharmonic injection-locked frequency dividers," *IEEE Journal of Solid-State Circuits*, vol. 34, no. 6, pp. 813– 821, Jun 1999.
- [22] B. Razavi, "A study of injection locking and pulling in oscillators," *IEEE Journal of Solid-State Circuits*, vol. 39, no. 9, pp. 1415–1424, Sept 2004.
- [23] H.-H. Hsieh, F.-L. Hsueh et al., "A V-band divide-by-three differential direct injection-locked frequency divider in 65-nm CMOS," in *IEEE Custom Integrated Circuits Conference 2010*, Sept 2010, pp. 1–4.
- [24] M. Abdulaziz, W. Ahmad, M. Törmänen, and H. Sjöland, "A Linearization Technique for Differential OTAs," *IEEE Transactions on Circuits* and Systems II: Express Briefs, vol. 64, no. 9, pp. 1002–1006, Sept 2017.
- [25] R. Y. Chen, "High-speed CMOS frequency divider," *Electronics Letters*, vol. 33, no. 22, pp. 1864–1865, Oct 1997.
- [26] B. D. Muer and M. S. J. Steyaert, "On the analysis of ΔΣ fractional-N frequency synthesizers for high-spectral purity," *IEEE Transactions on Circuits and Systems II: Analog and Digital Signal Processing*, vol. 50, no. 11, pp. 784–793, Nov 2003.
- [27] —, "A CMOS monolithic ΔΣ-controlled fractional-N frequency synthesizer for DCS-1800," *IEEE Journal of Solid-State Circuits*, vol. 37, no. 7, pp. 835–844, Jul 2002.
- [28] M. G. Johnson and E. L. Hudson, "A variable delay line PLL for CPUcoprocessor synchronization," *IEEE Journal of Solid-State Circuits*, vol. 23, no. 5, pp. 1218–1223, Oct 1988.
- [29] F. Herzel, S. A. Osmany, and J. C. Scheytt, "Analytical Phase-Noise Modeling and Charge Pump Optimization for Fractional-N PLLs," *IEEE Transactions on Circuits and Systems I: Regular Papers*, vol. 57, no. 8, pp. 1914–1924, Aug 2010.
- [30] S. Levantino, G. Marzin, C. Samori, and A. L. Lacaita, "A wideband fractional-n pll with suppressed charge-pump noise and automatic loop filter calibration," *IEEE Journal of Solid-State Circuits*, vol. 48, no. 10, pp. 2419–2429, Oct 2013.
- [31] J.-S. Lee, M.-S. Keel, S.-I. Lim, and S. Kim, "Charge pump with perfect current matching characteristics in phase-locked loops," *Electronics Letters*, vol. 36, no. 23, pp. 1907–1908, Nov 2000.
- [32] D. J. Allstot, "A precision variable-supply CMOS comparator," *IEEE Journal of Solid-State Circuits*, vol. 17, no. 6, pp. 1080–1087, Dec 1982.

- [33] J. Sharma and H. Krishnaswamy, "A dividerless reference-sampling RF PLL with -253.5dB jitter FOM and <-67dBc Reference Spurs," in 2018 IEEE International Solid - State Circuits Conference - (ISSCC), Feb 2018, pp. 258–260.
- [34] X. Gao, E. A. M. Klumperink, M. Bohsali, and B. Nauta, "A Low Noise Sub-Sampling PLL in Which Divider Noise is Eliminated and PD/CP Noise is Not Multiplied by N²," *IEEE Journal of Solid-State Circuits*, vol. 44, no. 12, pp. 3253–3263, Dec 2009.
- [35] X. Gao, E. A. M. Klumperink, P. F. J. Geraedts, and B. Nauta, "Jitter Analysis and a Benchmarking Figure-of-Merit for Phase-Locked Loops," *IEEE Transactions on Circuits and Systems II: Express Briefs*, vol. 56, no. 2, pp. 117–121, Feb 2009.
- [36] X. Yi, C. C. Boon, H. Liu, J. F. Lin, and W. M. Lim, "A 57.9-to-68.3 GHz 24.6 mW Frequency Synthesizer With In-Phase Injection-Coupled QVCO in 65 nm CMOS Technology," *IEEE Journal of Solid-State Circuits*, vol. 49, no. 2, pp. 347–359, Feb 2014.
- [37] V. Szortyka, Q. Shi, K. Raczkowski, B. Parvais, M. Kuijk, and P. Wambacq, "A 42 mW 200 fs-Jitter 60 GHz Sub-Sampling PLL in 40 nm CMOS," *IEEE Journal of Solid-State Circuits*, vol. 50, no. 9, pp. 2025–2036, Sept 2015.
- [38] W. Wu, R. B. Štaszewski, and J. R. Long, "A 56.4-to-63.4 GHz Multi-Rate All-Digital Fractional-N PLL for FMCW Radar Applications in 65 nm CMOS," *IEEE Journal of Solid-State Circuits*, vol. 49, no. 5, pp. 1081–1096, May 2014.
- [39] C. H. Tsai, G. Mangraviti, Q. Shi, K. Khalaf, A. Bourdoux, and P. Wambacq, "A 54-64 8 GHz subharmonically injection-locked frequency synthesizer with transmitter EVM between -26.5 dB and -28.8 dB in 28 nm CMOS," in *ESSCIRC 2017 - 43rd IEEE European Solid State Circuits Conference*, Sept 2017, pp. 243–246.
- [40] A. I. Hussein, S. Vasadi, and J. Paramesh, "A 50-66-GHz Phase-Domain Digital Frequency Synthesizer With Low Phase Noise and Low Fractional Spurs," *IEEE Journal of Solid-State Circuits*, vol. 52, no. 12, pp. 3329–3347, Dec 2017.
- [41] T. Siriburanon, S. Kondo et al., "A Low-Power Low-Noise mm-Wave Subsampling PLL Using Dual-Step-Mixing ILFD and Tail-Coupling Quadrature Injection-Locked Oscillator for IEEE 802.11ad," *IEEE Journal of Solid-State Circuits*, vol. 51, no. 5, pp. 1246–1260, May 2016.

Paper VI

Paper VI

A 4.3-mW mm-Wave Divide-by-Two Circuit with 30% Locking Range in 28-nm FD-SOI CM

O2018 IEEE. Reprinted, with permission, from

Therese Forsberg, Henrik Sjöland, and Markus Törmänen, "A 4.3-mW mm-Wave Divide-by-Two Circuit with 30% Locking Range in 28-nm FD-SOI CM," in 2018 *IEEE Nordic Circuits and Systems Conference (NORCAS)*, Tallinn, Estonia, Oct 2018, 4 pages.

A 4.3-mW mm-Wave Divide-by-Two Circuit with 30% Locking Range in 28-nm FD-SOI CMOS

Therese Forsberg #1, Johan Wernehag #2, Henrik Sjöland #3, Markus Törmänen #4

[#]Department of Electrical and Information Technology, Lund University Box 118, 221 00 Lund, Sweden

¹therese.forsberg@eit.lth.se ²johan.wernehag@eit.lth.se ³henrik.sjoland@eit.lth.se ⁴markus.tormanen@eit.lth.se

Abstract — A mm-wave divide-by-two circuit with high injection efficiency, implemented in a 28-nm fully-depleted silicon-on-insulator (FD-SOI) CMOS process is demonstrated stand-alone, as well as using an on-chip voltage controlled oscillator (VCO) as the input signal source. Measurements show that the divider has a 30.1 % tuning range centered at an output frequency of 24 GHz, at an input signal power of -1.5 dBm, and a power consumption of 4.3 mW from a 0.9 V supply. The VCO and divider combination has a tuning range of 10.2 %, centered at an output frequency of 30.2 GHz, at a total power consumption of 6.3 mW, and an output phase noise of -111 dBc/Hz at 10 MHz offset. The active area of the divider is 0.032 mm² and of the divider and VCO combination 0.043 mm².

Index Terms — CMOS analog integrated circuits, mm-wave circuits, frequency dividers, injection-locked frequency dividers.

I. INTRODUCTION

To meet an ever-increasing demand for higher data rates in wireless communication, wider bandwidths must be employed. As the spectrum below 6 GHz gets increasingly crowded, the required bandwidths will instead be found at mm-wave frequencies. The unlicensed band at 60 GHz that offers 7 GHz of bandwidth has thus received a lot of interest. Several standards, such as the IEEE 802.11ad for wireless local area networks, are already in place in this band [1]. Another interesting mm-wave band is around 30 GHz where some of the upcoming 5G NR bands will be situated.

Transceivers that target these standards, which are intended for battery-operated consumer electronics, will have to be both low-power and low-cost. To achieve a high integration level and a reduced cost, it is desirable to implement such transceivers using down-scaled CMOS processes.

A key part of any wireless transceiver is the local oscillator signal generation, which usually includes a phase-locked loop (PLL). A common PLL architecture is shown in Fig. 1 [2], where the phase frequency detector (PFD), charge pump (CP), low-pass filter, and most of the divider chain (N) operate at relatively low frequencies, whereas the voltage controlled oscillator (VCO) and the first divider stage in the chain operate at high frequencies. For mm-wave PLLs, the design of the VCO and first frequency divider stage thus requires careful consideration. Divider circuit techniques used at lower frequencies do not suffice, and the most popular solution is to

978-1-5386-7656-1/18/\$31.00 ©2018 IEEE



Fig. 1. PLL architecture, where the scope of this work is the mm-wave parts highlighted in gray.

use an injection-locked frequency divider (ILFD). A drawback is that the ILFD free-running frequency must be tuned for lock to occur, and without lock a clean frequency divided output signal will not be provided. Hence, there are strong incentives to try to increase the lock range of the divider, to ensure that the whole VCO frequency tuning range is covered, also in the presence of process variations. Successful efforts to implement divide-by-two ILFDs with wider locking ranges have lately been reported [3]-[8], as well as ILFDs with higher division ratios [9],[10]. However, using a division by two in the first frequency divider stage of a 60 GHz PLL means that signals are also generated at 30 GHz. Local oscillator signals for two important mm-wave frequency bands can thus be generated.

In this work, a CMOS divide-by-two circuit with efficient injection and a wide tuning range is presented, along with a version of the divider that receives its input from an on-chip VCO. The objective has been to create efficient mm-wave circuits for use in a 60 GHz PLL.

II. THEORY AND DESIGN

An ILFD is a tuned self-oscillating circuit, which is forced by injection locking to oscillate at the desired frequency. A tuned circuit that is well-known for its ease of implementation is a differential oscillator, which consists of cross-coupled transistor pair, with the DC current controlled by a current source, and a self-oscillation frequency determined by the drain inductance and capacitance. The schematic of the VCO together with the injection-locked divide-by-two circuit, which are both based on this architecture, is shown in Fig. 2. The efficiency of the injection mechanism directly affects the tuning range. An upper bound of the output-related frequency locking range for an ILFD is expressed by [11],[12] as

$$\omega_{range} \le \frac{\omega_0}{2Q} \cdot \frac{2}{\pi} \cdot \frac{I_{inj}}{I_{osc}} \tag{1}$$

where ω_0 is the self-resonance frequency, Q the quality factor of the divider LC tank, Iosc is the amplitude of the current going to the tank while free-running, and Iini is the amplitude of the current injected to the tank. A conventional solution is to inject a signal at the second harmonic frequency at the source terminal of the current source transistor, $V_{b,div}$. However, a more efficient method can be used, called direct injection [13],[14], where the signal is injected at the gate of an injection transistor connected between the differential outputs of the ILFD. This method is used, see Fig. 2, and the signal is also differentially fed to the ILFD, hence utilizing the full signal swing of the differential VCO. The injection transistors can be modeled as mixers, and to maximize their conversion gain the DC gate voltages are adjusted to about a threshold voltage above VDD for the NMOS, and about a threshold below VDD for the PMOS transistor. When dimensioning these transistors, the trade-off between locking range and power consumption in direct-injection ILFDs must be considered. There is an optimal size of the devices where they are large enough to provide a substantial injection current, but small enough not to significantly increase the load of the tank and thereby the power consumption [8]. For the stand-alone ILFD, an external differential local oscillator signal is fed to the circuit at point A, defined in Fig. 2.

The ILFD tank consists of a differential inductor, and a varactor, to maximize the frequency locking range, implemented with MOS devices. The inductor was implemented in the top metal layer of the process and simulated in ADS Momentum. The simulated inductance value for the stand-alone ILFD inductor was 450 pH. Due limited available input power to the ILFD at high frequencies in the measurement setup, the frequency range of the stand-alone ILFD was centered at 48 GHz. Also for measurement purposes, the output signals of the ILFD were routed to open drain buffers with approximately 5 dBm of output power.

To further verify the ILFD design, and to get one step closer to a PLL, the ILFD was as previously mentioned also combined with a VCO. The VCO provides the differential input to the ILFD, see Fig. 2, and it has been previously designed and verified [15]. It utilizes current source node filtering at twice the self-resonance frequency for improved phase noise performance, and it has tuning range of about 11 % centered at 64 GHz. Simulations show that when using the differential output of the VCO, the input signal to the ILFD is strong enough to make its lock range cover the VCO tuning range without changing its varactor setting. When combined with the VCO, the ILFD self-resonance is re-tuned to fit with the VCO frequency by reducing its tank inductance from 450 to 300 pH.



Fig. 2. Schematic of the VCO and divide-by-two circuit. For the stand-alone divider, the external differential signal to be divided is injected at points A+ and A-.

III. MEASUREMENT RESULTS

The stand-alone ILFD, as well as the VCO and ILFD combination, were implemented in the STMicroelectronics 28-nm ultra-thin body and buried oxide fully-depleted siliconon-insulator (UTBB FD-SOI) CMOS technology with 10 metal layers. The chip microphotograph is shown in Fig. 3. The chips were mounted on printed circuit boards, and the bias and supply voltages were provided through bond wires. The mm-wave input signals to the stand-alone divider were provided by a Rohde&Schwarz ZVA 67 network analyzer, through an Infinity i67 SGS microprobe from Cascade Microtech. The single-ended output signals from the divider buffers were also probed on-chip using an Infinity i40 GSG microprobe from Cascade Microtech, and measured using a Rohde&Schwarz FSU50 spectrum analyzer. For the phase noise measurements, the signal was down-converted using an external mixer driven by a Keysight E8257D PSG analog signal generator. It was then amplified and sent to an Aeroflex EuropTest PN9000 phase-noise measurement system.

Both circuits were measured with a 0.9 V supply. The bias was set for a 4.3 mW power consumption of the divider, which yields a wide frequency locking range, see Fig. 4. The divider locking range is 30.1 %, with an input signal power of -1.5 dBm. Even at a reduced power consumption of 1.8 mW, the divider locking range at that input power level is still 13.2 GHz, corresponding to 26.7 %. Fig. 5 shows the frequency locking range of the stand-alone divider at different varactor settings. As can be seen the width of the locking range is maintained when tuning its center frequency. The measured sensitivity of the divider is plotted in Fig. 6. As the maximum input signal power to the divider is limited by the measurement instruments, a higher input power than 0 dBm could only be generated at the lower frequencies of the locking range.



Fig. 3. Chip microphotograph.



Fig. 4. Measured locking range of the stand-alone ILFD vs. varactor setting, with -1.5 dBm input signal power.



Fig. 5. Measured locking range of the stand-alone ILFD vs. varactor voltage.



Fig. 6. The measured sensitivity of the divider, using the full divider varactor voltage range, and when it is fixed to VDD/2 (0.45V).



Fig. 7. Measured frequency tuning characteristic of VCO and divider combination.

Assuming that the sensitivity is symmetrical, which is supported by simulations, the estimated locking range at 0 dBm input power can be calculated to 35 %. The locking range for fixed varactor voltage is also shown in Fig. 6, yielding a 15% tuning range at -1.5dBm.

When measuring the VCO and divider combination, the power consumption of the divider was kept at 4.3 mW, and the total power consumption for the combination was 6.3 mW. The frequency tuning characteristics of the VCO and divider combination is shown in Fig. 7. Complying with simulations and measurements of the stand-alone divider, the entire tuning range of the VCO was covered by a single divider varactor setting. The best measured phase noise from the combined

TABLE I Performance Summary and Comparison to Recently Published MM-Wave Divide-by-Two ILFDs

Ref.	This	[3]	[4]	[5]	[6]	[7]	[8]
	work						
Tech.	28	65	65	65	90	90	65
(nm CMOS)							
Supply	0.9	0.8	0.9	0.75	1.2	0.8	0.8
Voltage							
(V)							
Pin (dBm)	-1.5	0	0	0	0	5	0
Frequency	48.8	62.8	42.25	63.9	58.8	55	66.4
(GHz)							
Tuning Range	30.1	29	53.2	9.7	20.5	34.9	39.2
(%)							
Power	4.3	1.9	9	6.3	8.6	0.8	2.9
Consumption							
(mW)							
Area	0.032	0.023	0.2	0.058	0.083	0.385	0.126
(mm²)							

circuit was at the lowest varactor setting for both the VCO and the divider, where the phase noise was -111 dBc/Hz at an offset of 10 MHz from the 28.5 GHz carrier.

The measured performance of the divider is compared to other recently published state-of-the-art mm-wave divide-bytwo CMOS circuits in Table I. As can be seen the design demonstrated in this work achieves a competitive combination of wide frequency range, low power consumption and input signal power, and small chip area.

IV. CONCLUSION

This paper presents two measured mm-wave designs aimed for PLL integration. The first is a differential direct-injection ILFD circuit dividing by 2, and the other is a re-tuned version of the same ILFD, combined with a 60 GHz VCO. Both designs are implemented on the same die in 28-nm UTBB FD-SOI CMOS technology. At an input signal power of -1.5 dBm the ILFD achieves a wide tuning range of 30.1 %, at a competitive power consumption 4.3 mW, while occupying an active chip area of just 0.032 mm². Even using a single varactor setting in the ILFD, it has wide enough locking range to safely cover the full 10.2 % tuning range of the VCO.

ACKNOWLEDGEMENT

The authors would like to thank STMicroelectronics for donating the IC fabrication, and the Swedish Research Council for funding this project. The authors also thank Peter Herrder at Ericsson in Lund for supplying essential lab equipment for the measurements.

REFERENCES

- [1] "IEEE Standard for Information technology– Telecommunications and information exchange between systems Local and metropolitan area networks–Specific requirements - Part 11: Wireless LAN Medium Access Control (MAC) and Physical Layer (PHY) Specifications," *IEEE Std* 802.11-2016 (*Revision of IEEE Std 802.11-2012*), pp. 1–3534, Dec 2016.
- [2] Razavi, B. *RF Microelectronics*, 2nd ed. Upper Saddle River: Prentice Hall, 2011, pp. 611-651.
- [3] J. Yin and H. C. Luong, "A 0.8V 1.9mW 53.7-to-72.0GHz selffrequency-tracking injection-locked frequency divider," 2012 IEEE Radio Frequency Integrated Circuits Symposium, Montreal, QC, 2012, pp. 305-308.
- [4] W. Chen et al., "A 53.6 GHz direct injection-locked frequency divider with a 72% locking range in 65 nm CMOS technology," 2013 IEEE MTT-S International Microwave Symposium Digest (MTT), Seattle, WA, 2013, pp. 1-3.
- [5] H. Fu, W. Fei, H. Yu and J. Ren, "A 60.8–67GHz and 6.3mW injection-locked frequency divider with switching-inductor loaded transformer in 65nm CMOS," 2014 IEEE MTT-S International Microwave Symposium (IMS2014), Tampa, FL, 2014, pp. 1-4.
- [6] H. Wang et al., "A 60GHz wideband injection-locked frequency divider with adaptive-phase-enhancing technique," 2011 IEEE Radio Frequency Integrated Circuits Symposium, Baltimore, MD, 2011, pp. 1-4.
- [7] T. Luo and Y. E. Chen, "A 0.8-mW 55-GHz Dual-Injection-Locked CMOS Frequency Divider," in *IEEE Transactions on Microwave Theory and Techniques*, vol. 56, no. 3, pp. 620-625, March 2008.
- [8] Y. Chao and H. C. Luong, "A 2.9mW 53.4–79.4GHz frequencytracking injection-locked frequency divider with 39.2% locking range in 65nm CMOS," 2012 IEEE Radio Frequency Integrated Circuits Symposium, Montreal, QC, 2012, pp. 337-340
- [9] Y. T. Chen, M. W. Li, H. C. Kuo, T. H. Huang, and H. R. Chuang, "Low-Voltage K-Band Divide-by-3 Injection-Locked Frequency Divider With Floating-Source Differential Injector," *IEEE Transactions on Microwave Theory and Techniques*, vol. 60, no. 1, pp. 60–67, Jan 2012.
- [10] M. Li, H. Kuo, T. Huang and H. Chuang, "60GHz CMOS divide-by-5 injection-locked frequency divider with an openstub-loaded floating-source injector," 2011 IEEE Radio Frequency Integrated Circuits Symposium, Baltimore, MD, 2011, pp. 1-4.
- [11] H. R. Rategh and T. H. Lee, "Superharmonic injection-locked frequency dividers," *IEEE Journal of Solid-State Circuits*, vol. 34, no. 6, pp. 813–821, Jun 1999.
- [12] B. Razavi, "A study of injection locking and pulling in oscillators," *IEEE Journal of Solid-State Circuits*, vol. 39, no. 9, pp. 1415–1424, Sept 2004.
- [13] M. Tiebout, "A CMOS direct injection-locked oscillator topology as high-frequency low-power frequency divider," in *IEEE Journal of Solid-State Circuits*, vol. 39, no. 7, pp. 1170-1174, July 2004.
- [14] C. Wu and C. Yu, "Design and Analysis of a Millimeter-Wave Direct Injection-Locked Frequency Divider With Large Frequency Locking Range," in *IEEE Transactions on Microwave Theory and Techniques*, vol. 55, no. 8, pp. 1649-1658, Aug. 2007.
- [15] T. Forsberg, J. Wernehag, A. Nejdel, H. Sjöland and M. Törmänen, "Two mm-Wave VCOs in 28-nm UTBB FD-SOI CMOS," in *IEEE Microwave and Wireless Components Letters*, vol. 27, no. 5, pp. 509-511, May 2017.